

Digital Commons @ George Fox University

Doctor of Education (EdD)

Theses and Dissertations

2018

Examination of the Psychometric Properties of the SCAT3 and SCAT5 Baseline Testing Data on NCAA Division III College Athletes

Jill A. Sikkema jsikkema@georgefox.edu

Follow this and additional works at: https://digitalcommons.georgefox.edu/edd

Part of the Education Commons

Recommended Citation

Sikkema, Jill A., "Examination of the Psychometric Properties of the SCAT3 and SCAT5 Baseline Testing Data on NCAA Division III College Athletes" (2018). *Doctor of Education (EdD)*. 117. https://digitalcommons.georgefox.edu/edd/117

This Dissertation is brought to you for free and open access by the Theses and Dissertations at Digital Commons @ George Fox University. It has been accepted for inclusion in Doctor of Education (EdD) by an authorized administrator of Digital Commons @ George Fox University. For more information, please contact arolfe@georgefox.edu.

EXAMINATION OF THE PSYCHOMETRIC PROPERTIES OF THE SCAT3 AND SCAT5 BASELINE TESTING DATA ON NCAA DIVISION III COLLEGE ATHLETES

by

JILL A. SIKKEMA

FACULTY RESEARCH COMMITTEE:

Chair: Susanna Thornhill, Ph. D.

Members: Dane Joseph, Ph. D.

Patrick Allen, Ph. D.

Presented to the Doctoral Department

and College of Education, George Fox University

In partial fulfillment of the requirements for the degree of

Doctor of Education

April 24, 2018



"EXAMINATION OF THE PSYCHOMETRIC PROPERTIES OF THE SCAT3 AND SCAT5 BASELINE TESTING DATA ON NCAA DIVISION III COLLEGE ATHLETES," a Doctoral research project prepared by JILL SIKKEMA in partial fulfillment of the requirements for the Doctor of Education degree in Educational Leadership.

This dissertation has been approved and accepted by:

4.24.18 Alsanconot

Committee Chair

Date

Susanna Thornhill, PhD

Associate Professor of Education

4/24/2018

Allo, a

Date

Patrick Allen, PhD

Date

Dane Joseph, PhD

Professor of Education

Assistant Professor of Education

Abstract

Medical professionals continue to revise on-field sport-related concussion (SRC) assessment tools to increase their validity and reliability. Multiple versions of the Sport Concussion Assessment Tool (SCAT) have been created, with the newest revision (SCAT5) published in 2017. This version contains changes from a required five-item word list to an optional ten-item word list in the *immediate memory* and *delayed recall* sections. The purpose of this study was to determine the underlying and latent structure of the SCAT3 and SCAT5 on pretest data, in addition to evaluating the internal consistency and ceiling effects of the instrument. An exploratory factor analysis was conducted on the Standardized Assessment of Concussion (SAC) portion of the SCAT, with a comparison of proportions for floor and ceiling effects. Results for this study showed the factor structure for both SCAT versions did not adequately align with the four sections of the assessment. Overall internal reliability of the SCAT5 was higher than previously reported for other SCAT versions ($\alpha = 0.764$) and statistical differences were present for ceiling effects between the SCAT3 and SCAT5 for immediate memory $(\chi^2(1, 427) = 218.290, p < .0000)$ and *delayed recall* $(\chi^2(1, 427) = 90.43, p < .0000)$. Findings reveal that the assessment tool structure may be different than what is intended. Despite these concerns, healthcare practitioners should evaluate their SRC decision-making processes to determine if this assessment should be utilized in their testing battery and consider its priority in the return-to-play process.

Keywords: Sports-related concussion, exploratory factor analysis, concussion assessment

Acknowledgements

I am so thankful for the experience to work toward a doctoral degree in education and for those who have supported me throughout the process. The road has been a challenge and it would have been impossible for me to complete this dissertation without the encouragement, support, and love of many people.

Above all I could not have finished without God, Jesus, and the Holy Spirit who have brought strength, encouragement, and peace throughout this process. I am so thankful that despite my weaknesses He has chosen to give me this opportunity and walk with me through it. May the lessons I have learned and the knowledge that I have gained be used to glorify Him to everyone I encounter.

Seth, you are the one and only human I would choose over and over again to be stranded with on a desert island. I am thankful that I have had the privilege of knowing you for more than half my life and don't ever want to do life without you. Each day you challenge me to grow in confronting my fears and learning to be flexible; may that never stop. I have had the privilege of watching you grow in your domestication over the last three years and am so thankful for all the delicious meals that you have made. Who would have thought you would find joy in the kitchen and eat exotic foods? Thank you for being a sounding board, editor, chef, childcare worker, voice of reason, comforter, chauffer, best friend, and lover.

Zeke, Ellie, and Simon you are the most crazy, loving, and adventurous kids I know. Thank you for sacrificing time with me to let me pursue my dreams. Thank you for your hugs of encouragement and patience when I was stressed. May you remember to pursue your dreams especially when it takes lots of hard work. I love you more than the stars in the sky.

ii

Paul, thank you for your listening ear and expertise in helping mitigate my fears during this process. I am very thankful for your generosity and support.

Becky, thank you for all the tea/coffee dates, prayers, and texts. You are a great friend, encourager, and all-around rock star. I have loved watching you push yourself past your fears as you strengthened me to push past mine.

Dr. Dane Joseph, thank you for helping me find a novel topic and see past the difficulties it presented. I appreciate your tireless effort of walking through the methodology and statistical findings where I presented half thoughts and sentences. Thank you for helping me understand the reasons behind the process and for your encouragement along the way.

Dr. Patrick Allen, it has been a joy to know you throughout this process. I appreciate your vulnerability in sharing both the positive and negative experiences you have encountered throughout your academic career. You have blessed many and glorified God in the process. Enjoy the rest before you.

Dr. Susanna Thornhill, it has been a long road since I met you at my doctoral interview and you have been a steady positive light in my life. I am beyond thankful that the Lord put us together for this dissertation process. Your kindness, thoughtfulness, and truthfulness have both encouraged and spurred me on to complete the task. I am thankful that you were willing to learn about concussion assessments with me and to ask me to write for those who did not understand my dissertation topic. May you be blessed beyond measure for all your service.

iii

Table of Contents
Acknowledgementsii
List of Tables viii
List of Figures ix
List of Abbreviations x
Chapter 1 1
Introduction1
Related Studies
Baseline testing
Standardized Assessment of Concussion4
Psychometric properties
Ceiling effect
Problem 6
Purpose Statement
Research Questions
Rationale
Significance of the Study10
Limitations and Delimitations
Definitions of Terms 12
Organization of Study

Chapter 2	
Review of the Literature	
SRC Epidemiology	
Testing Paradigms	
Baseline Testing	19
Normative Data	
Sport Concussion Assessment Test	
Conclusion	
Chapter 3	
Methodology	
Research Design	
Sampling and Participation	
Variables	
Timeline	
Data Analytics	
Ethical Issues	
Chapter 4	
Results	
Participants	
Exploratory Factor Analysis for SCAT3	

Reliabilities for SCAT3	50
Factor naming and structure	51
Idealized Reliabilities	53
Exploratory Factor Analysis for SCAT5	56
Reliabilities for SCAT5	61
Factor naming and structure	62
Idealized Reliabilities	65
Floor and Ceiling Proportions	71
Conclusion	72
Chapter 5	73
Conclusion	73
Discussion of Findings	74
Test administration	74
Participants	75
Structure of SCAT3 and SCAT5.	76
Limitations	83
Implications for Practice	84
Suggestions for Future Research	85
Conclusion	86
References	88

PPENDICES
opendix A
ppendix B 100
ppendix C 105
ppendix D 106
ppendix E 107
ppendix F111
ppendix G117
ppendix I 121
ppendix J 122

List of Tables

Table 1. Number of Questions and Points on SCAT3 and SCAT5 11
Table 2. Normative Data for Non-Concussed High School and Collegiate Football Players 25
Table 3. Normative Data for Non-Concussed Collegiate Mixed-Gender Athletes 27
Table 4. Example of Word Lists used on for immediate memory and delayed recall on SCAT3
and SCAT5
Table 5. Number of variables and appropriate sample size for EFA
Table 6. Sport Demographics for SCAT3 and SCAT5 Sample
Table 7. Sport Demographics for SCAT3 and SCAT5 Independent Sample
Table 8. Eigenvalues for SCAT3 EFA using Principle Component Analysis
Table 9. Structure Matrix Factor Loading for SCAT3 50
Table 10. Cronbach's Alpha Classifications
Table 11. Reliability Scale for SCAT 3 Factors 52
Table 12. Reliability Scale for SCAT3 Idealized Factors
Table 13. Reliability Scale for SCAT3 All SAC Items 56
Table 14. Eigenvalues for SCAT5 EFA using Principle Component Analysis
Table 15. Factor Loading using Structure Matrix for SCAT5
Table 16. Reliability Scale for SCAT 5 Factors 63
Table 17. Reliability Scale for SCAT5 Idealized Factors
Table 18. Reliability Scale for SCAT5 All SAC Items 69

List of Figures

Figure 1. SCAT3 scree plot from principle component analysis and oblique rotation	. 49
Figure 2. SCAT5 scree plot from principle component analysis and oblique rotation	. 59

List of Abbreviations

Athletic-exposures (A-E)

Balance Error Scoring System (BESS)

Certified Athletic Trainer (ATC)

Confirmatory factory analysis (CFA)

Exploratory factor analysis (EFA)

Glasgow Coma Scale (GCS)

Immediate Post-Concussion Assessment and Cognitive Testing (ImPACT)

Kaiser-Meyer-Olkin (KMO)

National Collegiate Athletic Association (NCAA)

Post-Concussive Symptom Scale (PCSS)

Sport Concussion Assessment Tool (SCAT)

Sports-related concussion (SRC)

Standard deviation (SD)

Standardized Assessment of Concussion (SAC)

Chapter 1

Introduction

Sports-related concussion (SRC) have received increased scrutiny from the media over the past two decades (Ku, 2017). In response to this pressure, medical professionals who routinely work with athletes are continually critiquing and evaluating the best protocols for SRC detection, treatment, return-to-play, and return-to-learn. At the forefront of these SRC discussions is the sideline assessment. Medical professionals want to know what tools are available and useful to properly and efficiently detect a SRC during a practice or competition. Although numerous sideline assessment tools have been developed and modified over the course of the last decade, research has questioned their effectiveness, validity, and reliability. These SRC assessments test for various deficiencies in cognition, balance, vestibular-oculomotor, and symptoms, all of which have been observed in players following a head injury.

One primary goal of SRC assessments is to test for cognitive impairments following an injury because cognitive deficits have been observed post-injury for athletes across many sports (Collie, Makdissi, Maruff, Bennell, & McCrory, 2006). This research by Collie et al. (2006) delineates that these cognitive dysfunctions can occur within information processing, memory, and attention subgroups. Various assessment tools have been created to assess SRC as researchers increase their focus on enhancing diagnosis and return-to-play criteria.

The Sport Concussion Assessment Tool (SCAT), created in 2004, by the Concussion in Sport Group during the Second International Conference on Concussion in Sport, is one attempt to standardized sideline SRC assessment (McCrory et al., 2005; Yengo-Kahn et al., 2016). When creating the SCAT, McCrory et al. (2005) combined existing tests to create a comprehensive assessment that would explore neurological and neurocognitive functions; elements of the following tests were included: Standardized Assessment of Concussion (SAC), Post-Concussive Symptom Scale (PCSS), modified Maddock's questions, on-field observations, and return-toplay guidelines. The SCAT was revised to create the SCAT2 in 2008 during the Third International Conference on Concussion in Sport (McCrory et al., 2009). Modifications to the SCAT2 included the addition of the Glasgow Coma Scale (GCS), alternate word lists for the *immediate memory* and *delayed recall* sections of the test, and a modified version of the Balance Error Scoring System (Yengo-Kahn et al., 2016). In 2013, the SCAT3 revision (see Appendix A) had very few modifications but included an additional balance testing option and a neck examination section (McCrory et al., 2013a).

Annual meetings of concussion experts continue to occur as knowledge of SRC assessment grows and informs necessary changes for assessment and management. During the 5th International Consensus Conference on Concussion in Sport, the newest revision of the SCAT (SCAT5) was discussed and subsequently published in 2017 (Echemendia et al., 2017b). The SCAT5 (see Appendix B) includes significant adjustments to the *immediate memory* and *delayed recall* portions of the assessment. This is where the optional use of a ten-item word list, instead of the traditional five-item word list (used in all previous versions of the SCAT), could be utilized. It should be noted that practitioners still retain the option of using the five-item word lists though the goal of providing the new ten-item word lists is to limit the reported ceiling effect. A ceiling effect is defined as a maximal score on the section. This means that if there are 15-items the participant gets all 15 correct. However, Echemendia et al. (2017b) note that this new format needs to be tested to determine if it is psychometrically viable.

In this research, I utilized exploratory factor analysis (EFA) to determine the psychometric properties of the SCAT5 in comparison to the SCAT3 with an additional focus on

the ceiling effects present in the *immediate memory* and *delayed recall* sections of both tests. This research contributes to the conversation on whether the variables tested in the SCAT5 have the same structure as the SCAT3 or if changes still need to be made to the assessment for it to be effectively utilized in clinical practice.

Related Studies

The following section explores studies that relate to the research topic, specifically baseline testing, Standardized Assessment of Concussions, psychometric properties, and ceiling effects.

Baseline testing. Some assessment tools require baseline assessments to help determine the presence of a SRC (Benedict et al., 2015; McCrea et al., 1998; McElhiney, Kang, Starkey, & Ragan, 2014; Sufrinko et al., 2017). Baseline testing allows for a post-injury comparison to the athlete's pre-injury abilities, which provides a more individualized assessment with the potential to minimize possible confounding variables (i.e. learning disabilities, previous injury, etc.). While these instruments have shown some relationship in diagnosing SRC, no single test comprehensively assesses all aspects of a concussion. As a result, clinicians are cautioned to use clinical judgment while using these instruments for diagnosis and return-to-play criteria (Echemendia et al., 2017a). Some tools that have been utilized that require baseline testing include Standardized Assessment of Concussions (SAC) (McCrea, 2001b), Sport Concussion Assessment Tool (SCAT) (Chin, Nelson, Barr, McCrory, & McCrea, 2016), King-Devick test (Brommer, Fowler, Hons, Gerwing, & Payne, 2016), ImPACT (Allen & Gfeller, 2011), CogSport (Collie et al., 2006), and Balance Error Scoring System (BESS) (Yengo-Kahn et al., 2016). Of these SRC assessments, the SCAT is the most commonly-used sideline assessment (Echemendia et al., 2017a).

Standardized Assessment of Concussion. The section of the SCAT under primary focus in this research study is historically referred to as the Standardized Assessment of Concussion (SAC). The SAC, generated in 1997, is one of the earliest tools created for cognitive SRC assessment (Yengo-Kahn et al., 2016). The SAC was incorporated into the earliest version of the SCAT and uses *orientation, immediate memory*, and *concentration* questions to help clinicians recognize and diagnose SRCs (McCrory et al., 2005). In 1998, McCrea explored the SAC's psychometric properties and found a significant difference between SAC baseline testing values and scores following a SRC, which suggests that the assessment can detect changes in cognitive function (McCrea et al., 1998).

Psychometric properties. In 2009, more published research contradicted McCrea's research by concluding that most SAC items have unacceptable psychometric properties with 76% of the items established as too easy (Ragan, Herrmann, Kang, & Mack, 2009). Based on the deficiencies in the psychometric properties discovered by Ragan et al. (2009), research was undertaken to find more psychometrically-sound words for the *immediate memory* and *delayed recall* sections of the SAC (McElhiney et al., 2014). McElhiney et al. (2014) focused on changing the difficulty of the words given and not on the repetition of the words. Therefore, they utilized a 10-item list repeated once, instead of the standard three repetitions, with a wide variation in words. This change maintained the total overall score for the SAC. This 10-item list was psychometrically sound (McElhiney et al., 2014). However, none of these words are found within the SCAT5 and, to the author's knowledge, no evaluation of the 10-item lists repeated three times (as is directed on the SCAT5) presently exists. Therefore, it is necessary to conduct research utilizing the format of the SCAT5 while repeating the a 10-item word list three times to determine if any structural changes have occurred with the revision.

The examples above illustrate the conflicting evidence on the psychometric properties of the SAC and SCAT regarding the instruments' validity, practice effects, and ceiling effects (Hecimovich & Marais, 2017; Ragan et al., 2009) and none of the studies reviewed thus far have reported any exploratory or confirmatory factor analysis, hence the need for a study like this one. Test-retest reliability on the SCAT ranges from .31 to .71 with an overall coefficient of .64 (McCrea, Kelly, & Randolph, 2007). McCrea, Kelly, and Randolph (2007) suggest that this low correlation occurs because there is minimal variation in score along with a small ceiling effect. Barr and McCrea (2001) reported the sensitivity and specificity of the SAC as .94 and .76 respectively with test-retest reliability at .55. Additionally, a practice effect was present when the test was administered after 120 days for male high school and college football athletes (Barr & McCrea, 2001). Barr and McCrea (2001) state that a practice effect can occur when a patient repeatedly takes an assessment, thus their score improves because they know how the test works and the items on it. This can be a problem because most institutions' serial testing strategy for baseline testing use the same instruments and word list annually. This phenomenon can possibly lead to a ceiling effect and is part of the reason for this study.

Ceiling effect. The ceiling effect of the *orientation* and *immediate memory* sections of the SAC portion of the original SCAT have been compiled and critically evaluated (Echemendia & Julian, 2001). McCrea et al. (1998) showed that a perfect score or ceiling effect was achieved for 7% of all subjects. They claimed this percentage as insignificant, yet it meant that there is the possibility that seven percent of the patients who took the SCAT were not being accurately assessed (McCrea et al., 1998). Ragan and Kang (2007) noted that the assessment is flawed during baseline testing and can lead to a misdiagnosis of a SRC, meaning that an assessor may conclude that an injury did not occur when one really did. In contrast, McCrea et al. (1998)

found a mean score of 14.51 out of 15 for all subjects taking the 5-item SAC test, which shows that most athletes scored at the top of the range of the memory assessment. According to McElhiney et al. (2014), this ceiling effect needs to be further examined with a mixed gender population to see if it exists using the SCAT3 and whether this ceiling effect is mitigated through use of the SCAT5 10-item list.

Problem

An emphasis on SRCs has grown over the past two decades along with scrutiny by athletes, parents, and media about the ability of sideline assessment tests to recognize and diagnose SRCs (Ku, 2017). These involve ethical and legal implications for physicians and healthcare professionals engaged in the recognition, diagnosis, and treatment of SRCs (Kirschen, Tsou, Bird Nelson, Russell, & Larriviere, 2014; Pachman & Lamba, 2017). Giving the best standard of care is a priority for healthcare professionals, so they continue to examine ways to assess, diagnose, and treat SRCs. These medical professionals aim to better understand SRCs through an extensive body of research that has been published in the last decade. SRC research explores physiological changes in the brain, on-field assessment, diagnostic procedures, and recovery processes including return-to-learn and return-to-play protocols. Yet despite growing SRC research, it is evident that several gaps still exist in the literature, including the need for objective SRC assessments (Elkington & Hughes, 2016) and sound objective measures that do not have a ceiling effect on mixed gender samples (Yengo-Kahn et al., 2016).

Purpose Statement

Medical professionals continue to revise on-field SRC assessment tools to increase their validity and reliability (Echemendia et al., 2017a; Echemendia et al., 2017b). Thus, multiple versions of the SCAT have been created over the last decade with the newest revision (SCAT5)

published in 2017 (Echemendia et al., 2017b). This version contains changes from a required five-item word list to an optional ten-item word list in the *immediate memory* and *delayed recall* sections of the test. These changes aim to address two issues within the SCAT instrument: specifically, ceiling effects and "sandbagging" (Echemendia et al., 2017b).

A ceiling effect occurs when an athlete takes the assessment and receives a perfect score. This means that there is a high likelihood that the athlete could receive a higher score if more variables were present. "Sandbagging" occurs when athletes purposefully study for the test ahead of time and/or pretend to remember less than they are capable of remembering on the baseline test in order to increase their chances of scoring well on a post-injury test in the event that they actually do have a SRC. This practice increases the chances that they could be returned to play sooner, but essentially invalidates the test results. What is particularly difficult about this practice is that there are no measurable data on how often it happens. While sandbagging can be assessed on the SCAT3 and SCAT5 to confirm participants have offered their best effort on a pretest, this question was not administered in the data set for this study. This is a significant concern because clinicians rely on these concussion assessments to return athletes to participation. Having said this, clinicians are also cautioned to utilize a battery of assessments before making a final return-to-play decision. Further research should focus on what influence this has on SCAT5 baseline testing.

The *immediate memory* portion of the SCAT instructs patients to repeat any words they can remember after a clinician reads a list of words. These words are to be spoken at a rate of one word per second. Immediately following the cessation of the words, the patient is to begin repeating as many words as s/he can remember in any order. This same word list is repeated by the assessor for a total of three attempts. The *delayed recall* section utilizes the same word list

but requires that the patient recall all the words on his/her own. This *delayed recall* occurs at the very end of the test and no less than five minutes after the *immediate memory* portion.

Test results for the SAC, which used a five-item word list, indicated that seven percent of patients received a perfect score on the entire test with no scores reported by section (i.e. *orientation, immediate memory, concentration,* and *delayed recall*) (McCrea et al., 1998). This means patients hit the ceiling on the test by being able to reach the maximum score, whether or not they had a SRC. This ceiling effect impedes the differentiation of post-injury assessment because patient ability is likely to be greater than five-items during baseline testing, yet only five words are tested. This effect would create a discrepancy between the patient's ability and observed score (Ragan & Kang, 2007), subsequently impeding the diagnosis of a SRC.

The SCAT5 version contains changes from a required five-item word list to an optional ten-item word list in the *immediate memory* and *delayed recall* sections of the test. Due to this structural change, a psychometric evaluation of the SCAT5 is necessary to determine if the structure of the assessment has changed and if a ceiling effect still exists with the ten-item word list. The literature specifically calls for further research on the structural change in the SCAT5 and to this author's knowledge, no such evaluation has been conducted. Other research provides a solid argument that SRC assessments are not psychometrically sound and should be re-evaluated to assess each item for difficulty level and discrimination capabilities (Ragan & Kang, 2007). Specifically, if these SRC assessments do not provide a valid baseline measure, then results from these assessments post-injury are questionable.

Research Questions

This research aimed to answer two questions:

RQ1. Are there differences in the factor structure and internal reliability between the SCAT3 and SCAT5 subsections of *immediate memory* and *delayed recall* assessments in an analysis of baseline data?

RQ2. Are there differences in the proportions of floor/ceiling effects for the SCAT3 and SCAT5 subsections of *immediate memory* and *delayed recall* assessments in an analysis of baseline data?

Rationale

SCAT is an assessment given to athletes before they are cleared for sports participation and is used as a baseline comparison for SRC identification during pre- and post-injury assessment (Yengo-Kahn et al., 2016). SCAT measures four functional components that can be impacted when an injury occurs: symptom evaluation, cognitive screening, neurological screening, and delayed recall. The cognitive screening is based on the Standardized Assessment of Concussion (SAC) and includes three sections: orientation, immediate memory, and concentration. The delayed recall section is included as a separate section at the end of the SCAT assessment to give test-takers enough time between the *immediate memory* testing and the recall portion to effectively test memory. It is important to note that the words used in the immediate memory section are also used in the recall section. More weight is given to the immediate recall section because patients are to recall the list three times while the *delayed* recall only requires a single word list. One key reason the SCAT3 was revised and updated was that research indicated a ceiling effect on the immediate memory and delayed recall portions of the exam because athletes were only given a five-item word list to recall (Echemendia et al., 2017a). This ceiling effect was limited to the *immediate memory* and *delayed recall* portions of

the assessment, and these were the portions that were revised in the SCAT5. Therefore, these portions of the SCAT5 are the focus of this research.

Significance of the Study

The newest revision of the SCAT extended the number of items tested for *immediate memory* from three attempts of a five-item word list (SCAT3) to three attempts of a ten-item word list (SCAT5). This changed the value of the *immediate memory* section score to 30 points instead of 15, thus placing more value on the total SCAT score and providing more opportunity for patient error in *immediate memory* (see Table 1). Creating more opportunities for variation in patient baseline cognitive data may help clinicians in their patient diagnosis and return-to-play protocols because it holds the potential to help them expose subtle changes in cognitive ability.

Due to the importance of identifying SRCs and establishing safe and effective return-toplay protocols, it is essential to research whether a ceiling effect exists in the ten-item word list for *immediate memory* and *delayed recall* tests. If the ceiling effect remains, then SCAT5 will need to be evaluated by other researchers to determine how other populations respond to the SCAT5 and inform future clinical use.

Table 1

	SCAT3		SCAT5	
Cognitive Sections	Questions	Points	Questions	Points
Orientation	5	5	5	5
Immediate Memory	5	15	10	30
Concentration	5	5	5	5
Delayed Recall	5	5	10	10
Total	20	30	30	50

Number of Questions and Points on SCAT3 and SCAT5

Limitations and Delimitations

There are several limitations to this study regarding the data collection, de-identification of participants, researcher's relationship with participants, engagement of participants, and generalizability. Data for this study were collected over multiple years and compiled using a medical database. Multiple administrators collected the data with no guarantee that they followed the SCAT written instructions. These variations could skew the data analysis in unanticipated ways.

De-identification of information was performed by the medical database report module and age was not a reportable variable. Therefore, birthdates were reported and converted into the participants' ages. Because of the researcher's association with many of the participants, previous knowledge existed of birthdays and the possibility of identification of the participants after de-identification increased. Another limitation focused on the engagement of participants in taking the assessment. Because the assessment was a requirement by the university's athletic administration, it is possible that participants did not give their best effort and therefore, results could be skewed.

This study will not be able to be generalized to the collegiate student-athlete population because the exploratory factor analysis (EFA) methodology is only exploratory in nature and does not answer a research hypothesis.

The delimitation of this study centers on the subjectivity of methodology for EFA. This subjectivity restricts the conclusions that can be drawn from the exploratory analysis on the SCAT.

Definitions of Terms

Baseline assessment – A test administered to an athlete prior to the beginning of an athletic season and prior to injury.

Ceiling effect – A maximum score on a section of an assessment.

Delayed recall – The ability to remember a given set of words after a minimum of five minutes.

Immediate memory – This refers to recalling information after a few seconds and can also be called short-term memory ("Short-term memory," n.d.).

Return-to-play – The criteria that needs to be met by an athlete with a SRC in order to be cleared for full sport participation.

Return-to-learn – The criteria that needs to be met by an athlete with a SRC in order to be cleared to return to classroom activities of reading, homework, notetaking, and listening to lectures.

Sports-related concussion – A traumatic brain injury that may be caused by a direct hit to any part of the body that transmits to the head and results in neurological impairment that can create symptoms that may increase over time but cannot be seen on neuroimaging. (McCrory et al., 2017).

Organization of Study

Chapter 2 outlines the prevalence of SRC within the collegiate setting and the role of baseline testing. This explanation is followed by the history of cognitive testing within the context of SRC injury including the reliability and validity, ceiling effects, and previous analysis of the test structure of the SAC and SCAT. Chapter 3 contains a discussion of the methodology for this proposed study. This chapter includes the research design, the sample size, data collection procedures, statistical analysis decisions, and ethical considerations. Chapter 4 reports the results of two EFAs, the reliabilities of the entire test, and of the corresponding subsections. These results are followed by a description of the floor and ceiling effect differences. Chapter 5 discusses the findings as they relate to the comparison of assessment structure, reliabilities, ceiling effects, and implications for practice.

Chapter 2

Review of the Literature

Sports-related concussion (SRC) awareness has increased over the last two decades, leading healthcare providers to look for the most effective ways to analyze and better current examination, diagnosis, return-to-play, and return-to-learn policies. As this search continues, experts in the field are continuing to revise testing protocols and instruments to increase reliability, validity, sensitivity, and specificity. As a response to this search for the best sideline assessment tool for SRCs, the 5th International Consensus Conference on Concussion in Sport published a fourth revision of the Sport Concussion Assessment Tool (SCAT), called the SCAT5. This revision focused on changing the number of items repeated on the *immediate memory* and *delayed recall* portions of the assessment.

A large body of research exists related to SRC and continues to grow annually. This literature review takes a focused look at SRC with research associated with the following key terms: The Sport Concussion Assessment Tool (SCAT), the Standardized Assessment of Concussion (SAC), epidemiology within the National Collegiate Athletic Association (NCAA), and SRC baseline testing. All research conducted utilizing the SAC or SCAT in any form for post-SRC assessment was excluded from this literature review unless mean scores for non-injured collegiate participants, or baseline data, were recorded. This choice was made to focus this literature review on the baseline psychometric properties of the SCAT5 as compared to the SCAT3 and not on how these tests give evidence to SRC diagnosis or return-to-play decisions. Additionally, it is important to recognize the prominent researchers who focus on baseline testing and its effects on diagnosis, return-to-play, and return-to-learn policies. Included in this list are

researchers such as: Michael McCrea, Steven Broglio, Kevin Guskiewicz, Robert Cantu, Ruben Echemendia, Paul McCrory, William Barr, and Margot Putukian.

This review of literature focuses first on the prevalence of SRC in NCAA athletics to contextualize a discussion of the role and usefulness of baseline testing for SRC assessment, along with the available research on the psychometric properties and ceiling effects for the SAC and the SCAT.

SRC Epidemiology

SRC prevalence is not fully understood in NCAA-affiliated institutions because there is no requirement to provide injury data. Over the last three decades, the NCAA has tried to create a clearer portrait of SRC prevalence through the enactment of an injury surveillance program that began in 1982 with the express purpose of better documenting and understanding all injuries that occur during collegiate sports participation (Dick, Agel, & Marshall, 2007). This program gathers information from a convenience sample of 250 institutions, approximately fifteen percent of all participating institutions that are willing to volunteer their injury data to the NCAA Injury Surveillance System. This percentage is acceptable for a research sample size but requires inferences to be made about the population that may not reflect the true prevalence of SRC.

The first SRC research compiled from the NCAA surveillance program was published in 2007 with data from 1988 to 2004 (Hootman, Dick, & Agel, 2007). Hootman et al. (2007) reported an SRC occurrence rate of .28 per 1000 athletic-exposures (A-E) for all NCAA sports when combining all SRCs during this period. The athletic-exposure or (A-E) designation refers to the number of times an athlete has the possibility of becoming injured due to play. This can occur either through a practice or competition. These A-Es are only counted if the athlete participated in the practice or competition and may vary greatly from institution to institution

because some coaches may only play small numbers of athletes on their rosters and others may play most or all. Given this, SRCs accounted for 5% of all injuries reported during this 25-year interval. Additionally, annual SRC rates were reported that showed variation in the data. From 1988-89, the rate was .17 and remained constant until a significant jump to .26 during the 1995-96 season. Another large increase occurred in 1997-98, from .26 to .32 and continued to increase through the 2001-02 season. Eventually the rate gradually returned to 1997-98 levels in 2003-04. Hootman and colleagues attribute these increases in SRC injury rates to two factors: an increase in the diagnosis and treatment of SRCs but may also account for an increase in SRC injuries.

It is important to note that around the time of the creation of the SAC in 1997, the rates of SRC incidence per A-E increased (McCrea et al., 1998). This increase in incidence may be due to the introduction of this new SRC sideline assessment tool that helped improved detection, such that the rate increase may not ultimately reflect an increase of injury (Hootman et al., 2007); yet the SRC incidence rate continued to rise. In 2005-06, it was reported from the NCAA ISS data, which included 180 NCAA institutions, that SRC rates increased to .43 per 1000 A-E with a practice rate of .28 per 1000 A-E and 1.02 per 1000 A-E during competition. This suggests that SRC are still occurring at a significant rate and therefore, sideline assessment tools need to be psychometrically sound for future generations (Gessel, Fields, Collins, Dick, & Comstock, 2007).

The next epidemiological SRC study was not released until 2015, almost a decade later and reported a very different portrait. The SRC data collected from 2009-2014 estimated from the NCAA Injury Surveillance System convenience sample that 10,560 SRCs occurred annually in collegiate athletics with an overall SRC occurrence rate of 4.47 per 1000 A-E (Zuckerman et al., 2015). Unlike Hootman et al. (2007), Zuckerman et al. (2015) did not report annual SRC occurrence rates. This lack of descriptive statistics about annual SRC incidence rate makes it difficult to discern whether a rise in SRC occurred over time or all at once. These data show a staggering difference compared to the two previous studies that gave overall SRC occurrence rates at .28 and .43 per 1000 A-E. It is possible that the incidence rate of SRC increased exponentially over the last decade but other factors may have influenced these statistics, including an increase in the number of student-athletes, and more dangerous styles of play. This research utilized a convenience sample from the NCAA Injury Surveillance System that only collects data from approximately fifteen percent of the population and therefore may not be a true representation of the population. Additionally, SRC diagnosis has increased over the past 20 years and therefore, researchers concluded that SRCs may not be increasing in number but rather that the diagnosis and reporting of SRCs that would have previously gone undetected has improved (Gessel et al., 2007; Hootman et al., 2007; Zuckerman et al., 2015).

Research shows that an increase in the SRC rate has increased over the last two and a half decades and may not be attributed solely to an increase in the injury rate but may have resulted from more objective, specific, and valid assessment tools. Because researchers believe that assessment tools play a role in the increase of the SRC occurrence rate, it is necessary to explore the tools used to determine these conclusions. To understand how these tools work, an exploration of the research will examine how baseline testing has been utilized within SRC protocols and how these assessments have been improved over the last two decades.

Testing Paradigms

Even though research shows that SRC rates have increased exponentially over the last two and a half decades, researchers believe that SRC rates have only slightly increased and that another variable may be attributed to this change. This change relates to the ability to diagnose SRC as assessment tools have become more sophisticated during this time and may have a greater effect on the occurrence rate. There are many assessment tools that can be utilized in the diagnosis and treatment of SRC. The focus here is on sideline assessments as one improvement added to the SRC protocol to assist clinicians in diagnosing SRC on the field in a timely fashion. To fully understand how the problems associated with psychometric properties of baseline testing occur, it is necessary to review the two testing paradigms: individual-centered standard and criterion-referenced standard (Ragan & Kang, 2007).

Individual-center standard focuses on the individual as compared to themselves and follows two assumptions: the participants' abilities are normally distributed and all ability levels can be measured. This standard is similar to a norm-referenced standard, but instead of comparing the datum to a norm, the athlete is compared to their pre-injury baseline SRC datum. When a SRC is suspected, the clinician will re-test the athlete using the same assessment tool and then compare the results to their baseline score. If the difference is greater than the confidence interval, then mental status change is confirmed (Ragan & Kang, 2007).

In contrast to the individual-center standard, researchers like McCrea et al. (1998) have proposed the use of a criterion-referenced standard. This standard utilizes the mean and standard deviation (SD) scores for a specific population and baseline testing becomes unnecessary (Ragan & Kang, 2007). Therefore, athletes will only be diagnosed with a SRC if their score is more than one SD from the mean value. The problem occurs when the ability level of the athletes differs significantly. For example, if the cut-off score for a SRC diagnosis is 25 and two athletes sustain a SRC, resulting sideline testing may reveal that one scored a 28 and the other a 25, yet both athletes have a SRC. The criterion-reference standard disregards the possibility that one athlete has a higher cognitive ability than the other, and thus this testing measure would require greater cognitive impairment for a SRC to be diagnosed. Therefore, if the assumptions of normality and ability discrimination are met with appropriate test validity, then individual-centered standard should be the choice for neuropsychological testing prior to and following a SRC (Ragan & Kang, 2007).

Baseline Testing. Even though Ragan and Kang (2007) conclude that individualcentered standard should be the choice for SRC testing, it is necessary to acknowledge the evolution of these sideline assessments. Individual-center standard, otherwise known as baseline testing, is a requirement of many SRC assessment tools. This baseline testing is essential because minimal or no normative data had been published prior to the publication of these assessment tools. Therefore, individualized baseline testing became the only way for clinicians to observe changes between pre-injury and post-injury abilities. These individualized baseline tests were used to compare SRC post-injury data to determine the absence or presence of a SRC.

Baseline testing for SRC usually occurs at the beginning of the academic year prior to the start of any school-sanctioned practices and competitions. When baseline testing occurs, it can include many different diagnostic tests including but not limited to SAC, SCAT, BESS, ImPACT, CogSport, and Sway. Test choice is currently determined by individual institutions and not mandated by the NCAA (NCAA Sport Science Institute, 2017). Presently, NCAA institutions are requested by the NCAA Sport Science Institute to have a SRC management plan that is available to the public through paper and web-based interfaces that includes institutional procedures for SRC education, pre-participation assessments, recognition and diagnosis of SRC, post-SRC management, returning athletes to competition, and returning students to the classroom. The specific request by the NCAA is that baseline testing be performed for all athletes prior to the start of their season. This baseline testing should include a minimum of four

basic components: a medical history related to brain injury or SRC, symptom evaluation, cognitive testing, and balance testing (NCAA Sports Science Institute, 2017).

These types of tests give clinicians an individualized comparison to utilize when an athlete sustains a SRC. This comparison can help with the initial diagnosis of the injury by providing a point of reference for the cognitive, balance, and neuropsychological abilities of the athlete. It is important to note that most SRC diagnoses occur on the field and therefore sideline assessment tools have been created for this specific reason, including the SAC and the SCAT. These sideline assessments are endorsed by the NCAA Sport Science Institute (2007) and were created to help clinicians overcome their lack of equipment, time, and testing atmosphere (McCrea et al., 2007).

Baseline testing is not a new concept and was first utilized for SRC in a research protocol created by Jeffery Barth and colleagues (1989) to determine neuropsychological and psychosocial changes following SRC in collegiate football players using Gronwell's PASAT. The PASAT examines concentration, attention, and immediate memory recall through an auditory numeric material manipulation (Barth et al., 1989). This baseline testing protocol is still currently utilized and can be seen in the research that focuses on the creation of normative data, and identifying the validity and reliability of SRC assessment tools (Echemendia & Julian, 2001).

As previously mentioned, some assessment tools require baseline SRC assessments to help determine the presence of a SRC (Benedict et al., 2015; McCrea et al., 1998; McElhiney et al., 2014; Sufrinko et al., 2017). While these instruments have shown some relationship in diagnosing SRCs, no single test comprehensively assesses all aspects of a SRC. As a result, clinicians are cautioned to use clinical judgment while using these instruments for diagnosis and return-to-play criteria (Echemendia et al., 2017a). Some tools that have been utilized, whether they require a baseline assessment or not, include the Standardized Assessment of Concussions (SAC) (McCrea, 2001b), Sport Concussion Assessment Tool (SCAT) (Chin et al., 2016), King-Devick test (Brommer et al., 2016), Immediate Post-Concussion Assessment and Cognitive Testing (ImPACT) (Allen & Gfeller, 2011), CogSport (Collie et al., 2006), and Balance Error Scoring System (BESS) (Yengo-Kahn et al., 2016).

There are conflicting data on the usefulness of baseline testing. A majority of the research has a positive view of baseline testing with only a few impartial studies to support its use. These are accompanied by expressions of concerns about the psychometric properties of the tests, themselves. Research that views baseline testing positively focuses on the usefulness of the data when comparing it to post-injury scores. According to McCrea (2001a), baseline testing is the preferred method of pre-screening athletes because it provides greater control of variability from variables that may influence the pre-injury and post-injury assessment. Some examples of these variables would include learning disabilities and previous history of SRC. During the same year the 1st International Symposium on Concussion in Sport was held in Vienna, baseline testing was agreed to be beneficial and needed even though these experts acknowledged the limitations of SRC assessments (Aubry et al., 2002). It is important to remember that at this time the SAC was currently in use but the SCAT's creation was still pending. This view of baseline testing remained the same three years later at the 2nd International Symposium on Concussion in Sport in Prague but in response to the lack of a consensus of a psychometrically strong sideline assessment tool, the SCAT was created (McCrory et al., 2005). Over the last decade, additional research has placed baseline testing in a positive light with regards to exertional testing of the SAC (Koscs, Kaminski, Swanik, & Edwards, 2009), variations in Post Concussion Symptom

Scale (PCSS) (Guskiewicz et al., 2013), disparities in post-injury care (Kirschen et al., 2014), standard of care (Broglio et al., 2014; Pachman & Lamba, 2017), and diagnostic accuracy (King, Brughelli, & Hume, 2014).

Even though there are researchers who view SRC baseline testing as favorable, there are still discussions that sideline assessments (i.e. SAC and SCAT) are not useful. The first mention of issues with baseline testing were published in 2001 along with many of the other studies on SRC. One major concern was that serial testing could cause problems with practice effects, player motivation, and non-injured athlete comparisons (Echemendia & Julian, 2001). Specific issues observed with SAC implementation included athletes rehearsing the recall lists months prior to assessment, changes in the rate of digit presentation (if faster than the expected rate of one second per number, then outcomes change), and ceiling effects. Echemendia and Julian concluded in 2001 the SAC should not be used as a clinical tool and only for research purposes until it could be properly validated. Other research with a negative view of baseline testing focuses on the item difficulty of the SAC, stating that baseline testing is not beneficial if the instrument does not differentiate between a wide variety of abilities. Following the analysis of item difficulty, it was determined that most of the items were too easy and therefore did not reflect the variation in abilities necessary to be a valid test that would warrant baseline testing (Ragan et al., 2009).

Other critiques for baseline testing is in the usefulness of these assessments for physicians when performing a physical exam (Matuszak, McVige, McPherson, Willer, & Leddy, 2016). These researchers suggest that alternative tools should be utilized to determine the mental status of patients, yet upon further inspection of the appendices provided with this research, the mental status testing utilized the same exact principles of the SAC assessment with the only differences being slight variations in words and numbers. The reality is that many physicians screening patients for SRC are under many of the same time constraints as athletic trainers engaging in sideline assessments, and therefore the practicality of utilizing a more detailed instrument is limited.

In comparison to general physician guidelines, assessment trends in SRC evaluation for neuropsychologists were reported that only fifteen percent utilize baseline testing and 92 percent will evaluate a SRC post-injury without baseline data (Lemonda, Tam, Barr, & Rabin, 2017). The research did not explore the reasoning behind this choice for the minimal use of baseline testing but Lemonda et al. speculated that neuropsychologists have numerous tests that can be utilized during office visits that have been validated and include normative data across specific age groups but may only have one version of the test, making serial testing imprudent. Additionally, they discussed the possible difficulty of athletes receiving medical reimbursements for these tests. This ability to use precise instrumentation, reimbursement opportunities, and lack of serial testing options may play a role in a neuropsychologist's choice to engage in baseline testing.

When Chin and colleagues (2016) were assessing reliability and validity for the SCAT3, they concluded that there were numerous variables that could inhibit proper baseline testing and therefore it was better to use normative data that had been carefully screened to use as a comparison when evaluating SRC incidences. Issues discussed included testing environment, athlete's motivation, testing resources, and the fact that the general body of research is ambiguous in its attempt to show the value of baseline testing. In contrast to this negative perspective on baseline testing, it is important to remember that a majority of the research sees baseline testing in a positive light.
As previously discussed, there is research that fully supports baseline testing and research that sees only the flaws. Yet other research presents the argument neutrally. According to a systematic review by YeYengo-Kahn et al. (2016), there needs to be more research completed on SCAT with and without the baseline testing present to determine if the SCAT can detect a SRC in either circumstance. Currently, there are no published studies on utilizing only normative data to diagnose an SRC. Other groups report that baseline testing can be helpful, but is not necessary for sideline SRC management (Hyden & Petty, 2016). In a systematic review by Echemendia et al. (2017a), it was concluded that symptom checklist, SAC, and mBESS were useful to clinicians for immediate diagnosis of SRCs with or without baseline measurements. It is important to remember that it has been three decades since the creation of the SAC and a little over a decade since the creation of the SCAT. During this time, normative data options have increased and more and more clinicians are utilizing these assessment tools.

Normative Data. As the arguments in favor of, negative to, or neutral toward baseline testing continue, it is essential to understand how normative data plays a role in this discussion. Normative data, also called norms, utilize a large sample dataset that reflect the intended population to determine test score estimates of population values following administration of an assessment (Zimmerman, 2011). Norms will be presented for both the SAC and the various versions of the SCAT because these are the two sideline assessments utilized for on-field assessment for SRC. It is important to note that Ragan and Kang (2007) refer to the data as criterion-referenced standard and not as normative data. Their argument is that true normative data are met including the normal distribution of all scores and all abilities would be present. This literature review only explores research related to baseline norms for non-concussed athletes

because the intent is to show how the structure of the baseline SRC assessment tool is affected by the increase in items.

The first normative data published on the SAC only established norms for male high school and college football athletes (N = 568). Presented below in Table 2, researchers reported a total score mean and corresponding standard deviation (SD) in addition to the four sections: *orientation, immediate memory, concentration,* and *delayed recall* for all non-concussed participants (McCrea et al., 1998). These data did not differentiate mean scores between high school and college athletes.

Table 2

		SAC
Cognitive Sections	Mean	Standard Deviation
Orientation	4.82	.43
Immediate Memory	14.51	.98
Concentration	3.40	1.27
Delayed Recall	3.84	1.11
Total Score	26.58	2.23

Normative Data for Non-Concussed High School and Collegiate Football Players

Note: Adapted from "Standardized Assessment of Concussion (SAC): On-Site Mental Status Evaluation of an Athlete," by M. McCrea, J. P. Kelly, C. Randolph, J. Kluge, E. Bartolic, G. Finn, B. Baxter, 1998, *Journal of Head Trauma Rehabilitation*, 13, p. 32. Copyright 1998 by Aspen Publishers, Inc.

Normative data for the SAC that includes both genders are available but there is a large disparity between the male and female sample sizes with 88% of the sample (N = 517) as male and only 12% female (N = 73). As can be observed in Table 3, there are minimal differences between mean and SD scores across genders. When comparing overall collegiate data to McCrea et al. (1998), the values are nearly identical. It is important to point out that data was also

presented by sport but this data included all education levels including junior high and senior high athletes. As a result, these data cannot be utilized to generalize this data for collegiate athletic participants. It is also important to note that 81% of the male subjects across education level were football players (McCrea et al., 2007) and therefore, the data is skewed toward male football players and does not reflect all other sports and genders. Additionally, these data came from the third edition of the *Standardized Assessment of Concussion: Manual for*

Administration, Scoring and Interpretation and the date of the data collected for this normative data could not be established. It is possible that this is the original normative data collected prior to the first edition which was published in 1998. If this is the case, these data are possibly out of date and need to be re-evaluated in the twenty-first century. At the very least, a larger sample size should be utilized to draw conclusions about female collegiate athletes.

Table 3

Normative Data for Non-Concussed Collegiate Mixed-Gender Athletes

	SAC				
Cognitive Sections	Mean	Standard Deviation			
Orientation	4.85	.39			
Immediate Memory	14.52	.94			
Concentration	3.57	1.17			
Delayed Recall	3.57	1.07			
Total Score	26.86	2.04			

Note: Adapted from "Standardized Assessment of Concussion: Manual for

Administration, Scoring and Interpretation," by M. McCrea, J. P. Kelly, C. Randolph,

J. Kluge, E. Bartolic, G. Finn, B. Baxter, 2007. Copyright 2007 by CSMi Medical Solutions.

Normative data for the SCAT were compiled from 2005-07 about a mixed-gendered sample of college athletes (Shehata et al., 2009). The data were not reported in a mean and SD format as previously observed in the SAC norms, but were presented as the percentage of participants who completed the task. Reported data combined the number of correct answers to the following sections: PCSS, immediate 5-word recall, delayed 5-word recall, months in reverse, and digits backward. Within this data set, no total score for the SCAT were calculated or reported. Ninety-six percent of the sample (N = 249) successfully repeated all five words. When separated into gender, 98.3% of women (N = 60) and 95.8% of men (N = 189) completed this task perfectly. *Delayed recall* scores were significantly less with 36.9% of all participants successfully remembering all words. The greatest contrast in this section is that only 29.6% of men had perfect recall compared to 60% of women. These data suggested that the test is a poor

measure of cognitive ability if 98% of women and 95% of men hit the ceiling on the SRC assessment for *immediate memory* and 36% for *delayed recall*.

SCAT2 and SCAT3 normative data for college athletes from a wide variety of sports who were 18-23 years old show no differences in athletes with and without a SRC history (Zimmer, Marcinak, Hibyan, & Webbe, 2015). Zimmer et al. (2015) describe the normative data split by gender with a female mean as 91.65 with a SD of 5.58 for total SCAT2 scores. The men's mean values were 90.83 with a SD of 5.6. This standard deviation is important when utilizing this normative data post-injury. The researchers also divided the data into pertinent sections of the SCAT2 assessment that could be evaluated separately, including the symptom score, SAC, and balance. Mean scores were provided for all portions of the assessment, but only SAC mean scores are provided because of the focused nature of this literature review. SAC mean scores for females were 27.63 with a SD of 1.87 and males' mean SAC scores were 26.97 with a SD of 2.05. These values reflect the previous work of McCrea and colleagues with individual female scores one point higher than males (McCrea et al., 1998; McCrea et al., 2007). Zimmerman et al. (2015) exhorts clinicians to be cautious in their diagnosis of a SRC with a score more than 1 SD from the mean and consider that real impairment is present if more than 1.5 SD from the mean when comparing post-injury SCAT assessments to the normative data (Zimmer et al., 2015). Currently, no normative data has been published for the SCAT5.

Sport Concussion Assessment Test

The Sport Concussion Assessment Tool, or SCAT, is one of the more common tools used to assess SRCs and is endorsed by practitioners (Aubry et al., 2002; McCrory et al., 2013). There are multiple versions of the SCAT: SCAT, SCAT2, SCAT3, and SCAT5. It is important to note

28

that there was no SCAT4 revision published. The two most recent versions are the SCAT3 and SCAT5 published in 2013 and 2017, respectively.

The format of the original SCAT included sections for Signs, Memory (modified Maddocks questions), Post Concussion Symptom Scale (PCSS), cognitive assessment, and neurological screening. The sign section included three questions in respect to loss of consciousness, convulsions, and balance impairments. The memory section used a modified version of the Maddocks questions while the cognitive assessment includes three subsections. The subsections included a one-time only 5-item word list (the practitioner could come up with any five words of their choice) for immediate recall, a recitation of the months of the year in reverse order, and the last section asks participants to recall numbers in reverse-order. This last subsection begins with three numbers and ends with six. Participants are given two chances to pass any given level before moving on to the next set of numbers (McCrory et al., 2005).

In 2009, the second modification of the SCAT occurred, named the SCAT2. Modifications included were the addition of the Glasgow Coma Scale (GCS), three trials for the *immediate memory* section, alternate word lists for the immediate and *delayed recall* sections, and a modified version of the Balance Error Scoring System (McCrory et al., 2009). In 2013, the third revision, SCAT3, had very few modifications but included an additional balance testing option and a neck examination section (McCrory et al., 2013). The last modification occurred in 2017, named the SCAT5. The major change to this version was in the number of words repeated for the *immediate memory* and *delayed recall* sections. All the previous versions of the SCAT used a 5-item word list and this was increased to an optional 10-item word list for the SCAT5 (Echemendia et al., 2017b). The scoring of these versions has changed over time with the addition of various sections. The first version of the SCAT did not provide a total score (McCrory et al., 2005) while the SCAT2 had a maximum of 100 points (McCrory et al., 2009), and SCAT3 excludes a maximum score (McCrory et al., 2013). The SCAT5 does not provide a total score either and therefore, it is difficult to compare normative data from version to version (Davis et al., 2017).

In the most recent systematic review focused on sideline screening, researchers concluded that the SCAT is "the most-well established and rigorously developed instrument for sideline testing" (Patricios et al., 2017, p. 893). When the SCAT was created, the goal was for patient education and SRC assessment by healthcare providers (McCrory et al., 2005), which is similar to the SAC goal of creating a SRC assessment for athletic trainers and other healthcare professionals (McCrea et al., 1998). This shows that both assessments were created for healthcare providers and in an effort to simplify the SRC assessment thus making SAC a natural subsection of the SCAT.

Even though researchers report that SCAT tests are the most rigorously developed sideline assessment tool (Patricios et al., 2017), it is necessary to understand that the quality of the instrument has been called into question over the years. Specific issues with the SCAT include the limitations of a short sideline assessment in both content and length, along with the challenge of creating a psychometrically-viable test for clinicians who have no previous experience with neuropsychological testing (McCrea, 2001a). As a result of the condensed assessment, a ceiling effect has been noted, specifically in the SAC portion of the SCAT. Researchers believe that this ceiling effect exists because the item difficulty is quite low (Ragan & Kang, 2007). Therefore, Ragan and Kang (2007) concluded that observed ability cannot reflect the true ability of an athlete, much less of many athletes. This skewness in both the SAC

and SCAT versions is pronounced, and mostly likely not related to a lack of normality in cognitive function. Instead, researchers believe it is related more to the lack of item difficulty.

Other issues with SCAT testing include practice effects, reliability, and lack of sensitivity and specificity. When utilizing the SCAT assessments for baseline testing and post-injury assessment, serial testing is required. Serial testing requires that athletes to be testing annually and in the case of SRC injury, daily until baseline values are achieved. Therefore, one athlete could have taken all three versions of the SCAT prior to injury. This is problematic when there are only three sets of words or numbers as seen on the SCAT3 (McCrory et al., 2013). In other words, athletes who are tested annually and post-injury will have a familiarity with these words because there is limited variation. Other ways that athletes can practice include participants rehearsing the months of the year in reverse order or inflating PCSS to have a higher baseline score and therefore have a lower chance of being diagnosed with a SRC, but there is no research to substantiate this claim. Other practice effects have looked at changes in SAC values for preseason, mid-season, and post-season where no significant differences were determined in nonconcussed collegiate football players (Miller, Adamson, Pink, & Sweet, 2007).

Since the SCAT was created as a compilation of already-existing assessment tools, there is little data on the reliability of the entire test. Therefore, data are published by assessment section and are presented as such in literature. The overall reliability of the SCAT has not been documented but the reliability of the SAC has. It has been reported anywhere between .42 and .71 (Guskiewicz et al., 2013). This value describes the internal consistency of the instrument and good values can vary but .70 or above is considered acceptable (Laerd Statistics, 2015a).

Sensitivity and specificity for SAC were also calculated based on an all-male sample of high school and college football players, which is problematic because these values do not reflect the entire population of athletes including females and non-football male athletes. Therefore, the generalizability of the data to females and other sports is limited. Barr and McCrea concluded that a one-point decrease between baseline and post-SRC score determines the presence of a SRC. Following this conclusion, they measured this same data set for sensitivity and specificity, which was .94 and .76 respectively (Barr & McCrea, 2001). Sensitivity reveals the true positive rate, meaning that there is a 94% chance of successfully detecting a SRC when using the SAC. Specificity is the true negative rate, meaning that there is a 76% chance of not having a SRC when the assessment is used. The same year, McCrea (2001a) published another study with what appears to be the exact same sample where he reported a slightly higher sensitivity at .95 and the same specificity as Barr and McCrea (2001). It has also been reported that sensitivity was as high as .80 and specificity was between .89 to .98 over the seven days that collegiate football players were evaluated using the SAC (McCrea et al., 2005). These specificity and sensitivity values show that there is a good chance that the SAC will accurately diagnose a concussion but is less likely to properly clear athletes of having sustained a SRC. Additionally, Guskiewicz et al. (2013) reported that sensitivity ranges from .80 to .94 with the highest occurring during the first 48 hours. Specificity was reported from .76 to .91. Therefore, research concluded that a change in mental status is noted with a 2 to 4-point change below baseline values which means that little variation in testing score is needed to diagnose a SRC. This is problematic because there is minimal variation in the cognitive abilities and if the small nuisances in the score are inaccurate, then a SRC could be misdiagnosed.

The last significant issue with the SCAT is the choice of words utilized during the *immediate memory* and *delayed recall* portions of the SCAT assessment. Only one published article is available focused on word choice and the SAC. McElhiney et al. (2014) utilized a 10-

item list with one repetition, instead of using 5-item word list repeated three times, with a wide variation in words, thus not changing the overall total score for the SAC. These words (penguin, magazine, tornado, luggage, splinter, cottage, mushroom, vehicle, demolish, and gutter) were determined to be psychometrically sound. But although these words were found to be of value, none were found within the SCAT5. This suggests that the current words utilized on the SCAT5 may not be difficult enough to be considered psychometrically sound. Following the release of the SCAT5, no other research on repeating of the 10-item lists repeated three times presently exists.

In the newest revision of the SCAT, the SCAT5, experts hope to improve the validity, reliability, sensitivity, and specificity of the assessment while reducing ceiling effects by improving item-difficulty (Echemendia et al., 2017b). The SAC, SCAT3, and SCAT5 assessments have three forms that utilize different words and numbers. Traditionally they are named Form A, Form B, and Form C. When evaluating item difficulty through an item-analysis, a majority of the SAC had unacceptable psychometric properties when evaluated by forms, meaning that the words were too easy. Each form's acceptable item percentages are as follows: Form A = 33%, Form B = 30%, Form C = 27% with 76% of the items being too simplistic (Ragan et al., 2009). Therefore, the items need to be more difficult by changing the word lists and adding more words. Currently, there is no research performed on the SCAT5 with regards to psychometric properties.

Conclusion

Experts in SRC want clinicians and other healthcare professionals to have a psychometrically sound sideline assessment tool for the diagnosis of SRC. This is essential because the number of SRC in collegiate athletics is has increased over the past ten years,

although questions still exist as to whether it is from an increase in more sound sideline assessment tools, from an increase in student-athlete participation, or from more dangerous techniques utilized during practice and competitions. Though most experts agree that baseline testing is the best standard of care, it is important to note that there are numerous questions about the ability of these sideline tests to accurately detect the presence of a SRC.

The SCAT5, as the newest version of the SCAT offers some changes with the number of *immediate memory* and *delayed recall* words on the assessment. The goal is to help improve the item-difficulty, validity, reliability, sensitivity, and specificity of the assessment. Because no research has been performed on the SCAT5, this research determined whether the factor structure of this assessment remains the same as compared to the SCAT3 or if the additional words changed the structure of the exam, therefore affecting the validity, reliability, sensitivity, and specificity of the instrument.

Chapter 3

Methodology

This chapter outlines the methodology for this study. This was a quantitative, nonexperimental, and exploratory research study on the structural differences between the SCAT3 and SCAT5, including an emphasis on the proportion of ceiling effects found specifically within the *immediate memory* and *delayed recall* portions of the assessment for both versions of the test. The purpose of this study was to answer the following research questions:

RQ1. Are there differences in the factor structure and internal reliability between the SCAT3 and SCAT5 subsections of *immediate memory* and *delayed recall* assessments in an analysis of baseline data?

RQ2. Are there differences in the proportions of floor/ceiling effects for the SCAT3 and SCAT5 subsections of *immediate memory* and *delayed recall* assessments in an analysis of baseline data?

To answer these research questions, I analyzed historical SRC data from one NCAA Division III institution with a variety of sports. All athletes who have completed both SCAT3 and SCAT5 assessments at the chosen institution were included in the data set regardless of gender, sport, or age. I analyzed the data using exploratory factor analysis (EFA), Cronbach's alpha, and proportions of the correct responses. The remainder of this chapter provides the specific details of the methodology.

Research Design

This quantitative study used a non-experimental, exploratory method to determine if differences exist in the factor structure, internal reliability, and ceiling/floor effects of the SCAT3 and SCAT5. I utilized exploratory factor analysis (EFA) on the SCAT3 and SCAT5 to

determine the underlying structure of the assessments. The goal of EFA is to reduce the number of variables (or items) in order to determine what kind of underlying factor structure exists (Laerd Statistics, 2015b; Osborne, 2014). This method helped determine the extent to which the five-item and ten-item word lists in the *immediate memory* and *delayed recall* portions were similar in latency. Additionally, internal reliability was assessed using Cronbach's alpha to determine how consistent respondents were across items and formats. Proportions of floor and ceiling effects were also reported.

Sampling and Participation. Data were collected from a large and established secondary data set created by an NCAA Division III institution's sports medicine staff compiled over multiple years of SRC baseline testing. The secured secondary data set was kept in a HIPPA-approved, password-protected environment. The data set came from an institution whose policy required all contact and collision sports athletes to receive annual baseline SRC testing before clearance to participate in university-sponsored athletic activities. Subsequently, the data set encompassed SRC data for athletes who played a variety of sports: football, volleyball, basketball, soccer, baseball, softball, lacrosse and track and field. In this study, all participants who completed the 2016 (SCAT3) and/or 2017 (SCAT5) baseline testing were included in the data set.

Participants for SCAT3 (N = 416) and SCAT5 (N = 395) met the recommended minimum sample size of 10 subjects per variable (Laerd Statistics, 2015b), because a total of 30 variables (words) are used on the SCAT3 and the SCAT5, as evidenced in Table 4. Table 5 indicates how a minimum of 300 participants should be used, to correspond with 30 variables.

Table 4

Example of Word Lists used on for immediate memory and delayed recall on SCAT3 and SCAT5

	SCAT3		SCAT5					
List A	List B	List C	List A		List B		List C	
Elbow	Candle	Baby	Finger	Candle	Jacket	Dollar	Baby	Elbow
Apple	Paper	Monkey	Penny	Paper	Arrow	Honey	Monkey	Apple
Carpet	Sugar	Perfume	Blanket	Sugar	Pepper	Mirror	Perfume	Carpet
Saddle	Sandwich	Sunset	Lemon	Sandwich	Cotton	Saddle	Sunset	Saddle
Bubble	Wagon	Iron	Insect	Wagon	Movie	Anchor	Iron	Bubble

Table 5

Number of variables and appropriate sample size for EFA

	SCAT3	SCAT5
Number of Word Lists	3	3
Words per List	5	10
Sample Size per Variable	10	10
Total Sample Size Needed	150	300

Variables. There are three versions of each SCAT: A, B, and C. Each patient was only given one word-list during the examination and words were recalled in any order remembered. This included a five-item word list for SCAT3 and a ten-item word list for the SCAT5. A total of fifteen usable words for the SCAT3 were available when combining all three lists. A total of thirty useable words were available when combining the three lists for the SCAT5. Because of the limited variation of the words between the SCAT3 and SCAT5, fifteen of the 45 words from the

SCAT3 were utilized on the SCAT5, thus creating 15 duplicate variables. This resulted in a final variable list of 30 words. Therefore, 30 variables were utilized in this study as can be seen in Table 5, even though there are 45 words in the table. These variables were used for both the *immediate memory* and *delayed recall* potions of the SCAT3 and SCAT5.

Data were compiled from a secure medical database and de-identified through a report function within the medical database except for the participants' birthdays to determine age. Once the data were cleaned, it was analyzed using an EFA to determine the underlying structure of the assessments and the proportion of ceiling effects were calculated from the data set.

Timeline. This timeline outlines the various stages of this research:

1. February 2018, data use approval obtained (see Appendix C)

2. Late February 2018, research proposal accepted.

3. Early March 2018, IRB approval was obtained (See Appendix D).

4. Middle of March 2018, data was retrieved and cleaned from Sportsware Online for the SCAT3 and SCAT5 *immediate memory* and *delayed recall* sections along with age, academic standing, and race/ethnicity. During this process, the data was de-identified to protect the confidentiality of the participants and limit researcher bias.

5. Late March 2018, statistical analysis of data was performed.

6. Early April 2018, chapters 4 and 5 were written.

7. Late April 2018, dissertation defense was completed.

Data Analytics. RQ1. Are there differences in the factor structure and internal reliability between the SCAT3 and SCAT5 subsections of *immediate memory* and *delayed recall* assessments?

The data analysis included:

- 1. Cleaning the data
- 2. Choosing an extraction method
- 3. Determining the number of factors for analysis
- 4. Determine appropriate rotational method
- 5. Interpreting results
- 6. Cronbach's alpha was run to determine internal reliability

The following provides greater detail on the rationale and processes of the EFA. I cleaned the data to limit bias or derailment of analysis by specifically looking for and eliminating types of mis-responses and missing data (Osborne, 2014). Following the cleaning, an extraction method was determined and a principle component analysis factoring method was utilized (Osborne, 2014). Assumptions for factor analysis were evaluated including multicollinearity, Bartlett's test of sphericity, and Kaiser-Meyer-Olkin (KMO). Other extraction techniques have difficulty meeting these assumptions and show little benefit when used over these previous techniques (Osborne, 2014).

Next, I determined the number of factors that should be analyzed followed by a decision to use an oblique rotation because a correlation existed among the variables. This step increased the ease of interpreting the results of EFA (Osborne, 2014). There are two different types of rotations: orthogonal and oblique. Each depends on the correlation between factors. This correlation, if any, was determined, therefore, an oblique rotation was applied (Osborne, 2014). The fifth step in the EFA process was to interpret the results. It was essential to determine if the results were sensible. EFA aims to create results that are meaningful but sometimes fails to provide functional and practical application with respect to an instrument's conceptual or theoretical framework. It is possible for EFA to provide results, but they may not be useful to the researcher based on the framework; therefore, careful scrutiny of these results was imperative (Osborne, 2014). Additionally, reliability of the EFA was determined by running a Cronbach's alpha. The final step in the EFA was to see if the results could be replicated through an additional EFA or CFA analysis to help determine the strength and relevance of the findings (Osborne, 2014). This step was beyond the scope of this research and was not included in the analysis.

The data set was created by the institution's sports medicine staff and included one subset of NCAA Division III athletics. Additionally, the decision to use EFA to examine the data means this study cannot be generalized to all NCAA Division III institutions because exploratory factor analysis is used to explore data and not used to confirm a hypothesis (Osborne, 2014). Nevertheless, the information obtained from this research adds to the current body of literature by determining if differences exist in the assessment structures and in the proportion of ceiling effects for the SCAT3 (with a five-item word list) compared with the SCAT5 (with a ten-item word list). This is important to determine because the structure of the instrument is essential to the validity and reliability of the test. Additionally, the presence of ceiling effects is inversely associated with the number of test-items; i.e., presence is more likely with tests of fewer items. If there are high proportions of ceiling effects in one version, then it is plausible that a high grouping of scores is a result of test-format defects, leading to construct-irrelevant invariance. In other words, detecting true differences in a student-athlete's ability to accurately recall words becomes problematic. Thus, athletes who are suspected of sustaining a SRC but score highly because of the ceiling effect due to immediate memory/delayed recall bias, have their mental and physical health endangered if they are truly concussed but test as a false-negative.

RQ2. Are there differences in the proportions of floor/ceiling effects for the SCAT3 and SCAT5 subsections of *immediate memory* and *delayed recall* assessments?

To analyze the second research question, I identified the number of participants with either zero or 15 correct responses on the SCAT3 *immediate memory* portion. The zero showed a floor effect, meaning that the participant could not score lower. In contrast, 15 correct responses showed a ceiling effect meaning that the participant could not score higher. These scores were summed and differences in average score were examined. Additionally, the same procedures were followed for the SCAT5, with the exception that the ceiling score was 30 based on the three attempts of the 10-item construct. Floor and ceiling effects were then calculated for *delayed recall* which was zero and five for SCAT3 and zero and ten for SCAT5. I ran a chi-square test to determine statistical significant differences between groups on an independent sample. An independent sample was created because there is no statistical test for differences of proportions that exists for dependent groups. The independent group was created by determining all participants who had the same sport, age, and birthdate. Once these commonalities were discovered, the participant was excluded from the sample.

Ethical Issues

At the time of this study, I was a full-time employee of the institution where the data was collected. My role as assistant athletic trainer was to collaborate with the other sports-medicine staff members to review medical policies, review student-athletes medical history, oversee and participate in compiling baseline testing, clearing them for participation, evaluating and rehabilitating sports-related injuries and general medical conditions, and provide practice and game coverage. Because my employment encompassed numerous tasks, I had a personal working relationship with many of the student-athletes who participated in this study. Though I

made every effort to treat the data in an unbiased manner, particularly by having the medical database de-identify the data prior to statistical analysis, there was the possibility that this relationship could constitute a conflict of interest.

As a member of the sports-medicine team at the participating institution, I helped to collect approximately 10 percent of the data contained in the database analyzed. Therefore, I had some prior knowledge of participants' names and scores. I addressed this issue by maintaining transparency with my committee about knowledge. I reduced my ability to identify participants by running a report through the medical database to de-identify the data, except for date of birth to determine age, before data analysis procedures began. Finally, I structured the study so as to quantitatively analyze the data in such a way that did not incentivize my knowing participants' identifies/scores.

Permission for the use of the data set was obtained from the Director of Sports Medicine at the given institution and IRB approval was obtained prior to the compilation of data.

Chapter 4

Results

The purpose of this study was two-fold: to explore the factor structure of the Sport Concussion Assessment Tool (SCAT) versions three (SCAT3) and five (SCAT5) during baseline testing. Additionally, this study sought to determine and compare the proportion of floor and ceiling effects of the *immediate memory* and *delayed recall* portions of these assessment tools.

The research questions guiding this study were:

RQ1. Are there differences in the factor structure and internal reliability between the SCAT3 and SCAT5 subsections of *immediate memory* and *delayed recall* assessments in an analysis of baseline data?

RQ2. Are there differences in the proportions of floor/ceiling effects for the SCAT3 and SCAT5 subsections of *immediate memory* and *delayed recall* assessments in an analysis of baseline data?

To answer the first research question, this chapter reports the demographic information of participants including gender, sport, and age. This is followed by a presentation of exploratory factor analyses (EFAs) as well as internal consistency estimates for both SCAT versions. In answer to the second research question, proportions for the floor and ceiling effects for the *immediate memory* and *delayed recall* portions of the assessment are reported.

Participants

Two groups of participants were utilized for this study. For RQ1, all participants at George Fox University who took the SCAT3 (N = 416) or the SCAT5 (N = 395) assessment were included in two separate EFAs. Gender distributions for the SCAT3 were 33.4% (N = 139) female and 66.6% (N = 277) male. The SCAT5 had a distribution of 35.7% (N = 141)

female and 64.3% (N = 254) male. The SCAT3 and SCAT5 were administered to participants in August 2016 and 2017, respectively. Sports participation for SCAT3 and SCAT5 are presented in Table 6 with the largest participation being football for both assessments. For the SCAT3, soccer, baseball, and basketball were the next three most-played sports, while for the SCAT5, track & field, soccer, and baseball were the top four most-played sports represented in the sample. The ages of participants for the SCAT3 ranged from 18 to 26 (M = 20.91, SD =1.456) while the age range for SCAT5 participants was 19 to 24 (M = 19.94, SD = 1.417).

Table 6

SC	CAT3	SC	AT5
Sport	Count (%)	Sport	Count (%)
Football	143(34.4)	Football	120(30.4)
Soccer	60(14.4)	Track & Field	54(13.7)
Baseball	55(13.2)	Soccer	53(13.4)
Basketball	46(11.1)	Baseball	42(10.6)
Track & Field	36(8.7)	Basketball	32(8.1)
Softball	32(7.7)	Tennis	26(6.6)
Lacrosse	18(4.3)	Softball	24(6.1)
Volleyball	15(3.6)	Volleyball	16(4.1)
Other	10(2.4)	Other	12(3.0)
Cross Country	1(.2)	Lacrosse	11(2.8)
		Cross Country	5(1.3)
Total	416 (100)		395 (100)

Sport Demographics for SCAT3 and SCAT5 Sample

For RQ2, participants who took both SCAT3 and SCAT5 were eliminated from the sample to create two distinctive, independent samples. This was accomplished by matching participants' gender, birthdate, and sport. If all three were identical on both lists, those data were deleted from the sample. This decision was made because there are no statistical methods to determine statistical differences of proportions for dependent groups. Therefore, independent

samples were created for SCAT3 (N = 224) and SCAT5 (N = 203). Gender distributions for these samples were as follows: SCAT3 had 33.9% (N = 76) female and 66.1% (N = 148) male while the SCAT5 had 37.9% (N = 77) female and 62.1% (N = 126) male. Sports participation for SCAT3 and SCAT5 are presented in Table 7 with the most participants engaged in football for both assessments. Within the SCAT3 sample, baseball, basketball, and soccer were the next three highest participation rates, while the SCAT5 indicated football was followed by track & field, baseball, and tennis. The ages of participants for the SCAT3 ranged from 20 to 26 (M =21.15, SD = 1.621) while the age range for SCAT5 participants was 19 to 24 (M =19.24, SD = 1.252).

Table 7

	SCAT3	S	CAT5
Sport	Count (%)	Sport	Count (%)
Football	66(29.5)	Football	46(22.7)
Baseball	38(17.0)	Track & Field	32(15.8)
Basketball	29(12.9)	Baseball	27(13.3)
Soccer	28(12.5)	Tennis	26(12.8)
Track & Field	21(9.4)	Soccer	21(10.3)
Softball	18(8.0)	Basketball	15(7.4)
Lacrosse	11(4.9)	Other	10(4.9)
Other	6(2.7)	Softball	9(4.4)
Volleyball	6(2.7)	Volleyball	9(4.4)
Cross Country	1(.4)	Cross Country	4(2.0)
		Lacrosse	4(2.0)
Total	224 (100)		203 (100)

Sport Demographics for SCAT3 and SCAT5 Independent Sample

Exploratory Factor Analysis for SCAT3

An EFA was conducted, using IBM SPSS version 24, for the Standardized Assessment of

Concussion (SAC) portion of the SCAT3 that measures orientation, immediate memory,

concentration, and *delayed recall* for a sample of 416 NCAA Division III collegiate athletes. Statistical assumptions for the use of EFA were assessed prior to the analysis including multicollinearity, overall Kaiser-Meyer-Olkin (KMO), and Bartlett's test of sphericity. Once these assumptions determined the appropriateness for factorization of an initial EFA, principle component analysis extracted the factor structure with a single varimax rotation. The solution sought to extract four components based on the original four sections of the SAC portion (*orientation, immediate memory, concentration,* and *delayed recall*) of the SCAT3. Following the analysis of this factorization, an oblique rotation was applied to determine if the factor loadings changed among the four components.

Multicollinearity was the first assumption assessed in the EFA process. This was assessed by using the determinant which was reported as .04. If the determinant is above .00001, factorization can occur (Laerd Statistics, 2015b). Therefore, multicollinearity does not occur for this data set. This means that the variables do not have high correlations amongst themselves thus eliminating the potential for multiple variables measuring exactly the same thing (Mertler & Vannatta, 2010). This was additionally shown through the inspection of the correlation matrix. Following the inspection, it was determined that at least one correlation coefficient was greater than 0.3 for all variables (see Appendix E) and none were highly correlated, thus the assumption was met.

The second assumption states that Bartlett's test of sphericity was statistically significant. The SCAT3 EFA showed Bartlett's test of sphericity ($\chi^2(435) = 1304.73, p < .0000$) to be statistically significant. The null hypothesis was that the correlation matrix and identity matrix are identical. Having identical matrices can be a problem since it practically means that there were not a sufficient number of correlations (and therefore structure) to the underlying latent variables. Given that Bartlett's test was statistically significant, the null hypothesis was rejected, implying that the two matrices were indeed different. This indicated that the data factorization for structure or dimension was warranted.

The last assumption states that a moderate overall KMO measure should be .60 or above. This statistic examined sampling accuracy, specifically sample size per variable and the proportion of variance in the variables that might have common variance. This EFA reported a proportion of 0.57 which is "miserable" by some standards (Dziuban & Shirkey, 1974, p. 359) and is a violation of the assumption of sampling adequacy. Nevertheless, and consistent with other research, the EFA was run given that: (1) the other two assumptions were met, and (2) the sample size (416) was appropriate for factoring 30 variables at a ratio of close to 13:1 subjects to variables (Osborne, 2014).

EFA revealed that 13 factors had eigenvalues greater than one. Based on the four sections of the SCAT3 assessment (*orientation, immediate memory, concentration,* and *delayed recall*) the EFA was conducted using four factors which accounted for 8.58%, 6.27%, 5.89%, and 5.52% of the total variance, respectively. These values are presented in Table 8. A scree plot is used to determine the number of components used to conduct the EFA and gives a graphical depiction that should bend and level off (Mertler & Vannatta, 2010). Figure 1 depicts the scree plot for these data, which did not accurately depict the four factors because there was no true bend in the line and leveling off did not occur. The ambiguity of the line bend made it difficult to determine if four factors were the best choice for the SCAT3 EFA. Additionally, since no leveling of the line occurs, it was difficult to determine if the remaining factors were equal in size (Mertler & Vannatta, 2010). The implications associated with these issues are discussed further in the structure section of Chapter 5.

The four-factor solution explained 26.2% of the total variance, therefore an oblique rotation was applied to determine if changes to factor loading occurred. After the oblique rotation was applied, the variables did not load in the way that the assessment tool intended. This means that not all of the *orientation, immediate memory, concentration,* and *delayed recall* questions loaded into separate factors. In many of the factors, there were multiple variables from 2 or more assessment categories. The factor loadings of the rotated solution are presented in Table 9. All factor loadings less than 0.2 were suppressed when the EFA was conducted, which accounts for the blanks in Table 9.

Table 8

Eigenvalues for SCAT3 EFA using Principle Component Analysis

Initial Eigenvalues		values		Rotation			
Factor	Total	% Variance	Cumulative % Total		% Variance	Cumulative %	Total
1	2.575	8.584	8.584	2.575	8.584	8.584	2.331
2	1.882	6.274	14.857	1.882	6.274	14.857	1.917
3	1.767	5.891	20.748	1.767	5.891	20.748	1.886
4	1.656	5.521	26.269	1.656	5.521	26.269	1.747



Figure 1. SCAT3 scree plot from principle component analysis and oblique rotation.

EXAMINATION OF SCAT5 BASELINE TESTING

Item		Fac	tors	
(N = 30)	F1	F2	F3	F4
C1	0.680			
IM3_5	0.645			
IM3_2	0.604			
IM3_4	0.568			
IM2_2	0.499			
IM2_5	0.467			
IM2_4	0.415			
O_Date				
IM3_3				
DR1		0.628		
DR5		0.617		
DR2		0.587		
DR4		0.586		
DR3		0.424		
O_Month		0.264		
O_Time		0.205		
O_Year				
IM2_3			0.631	
IM1_2			0.628	
IM1_3			0.591	
IM1_4			0.562	
O_Day			0.239	
IM1_5			0.225	
C3				0.790
C4				0.653
C2				0.585
C_Months				0.353
IM3_1				
IM1_1				
IM2_1				

Table 9 Structure Matrix Factor Loading for SCAT3

Note Extraction Method: Principal Component Analysis Rotation Method: Oblimin with Kaiser Normalization

Reliabilities for SCAT3

Cronbach's alpha was utilized to examine the internal consistency of the SAC portion of the SCAT3. The internal consistency values reported in this chapter are based off the categories found in Table 10 (Manerikar & Manerikar, 2015). The following sections explore the

reliabilities and factor name assignments. Additionally, this section is broken into two categories because the variables did not load on the exact factors as outlined in the SCAT3 assessment. Therefore, the first section reports the reliabilities for the variables that loaded on the factors during the EFA; the second section reports idealized reliabilities. Idealized reliabilities are the result of variables if they are divided into their assigned sections (i.e. *orientation, immediate memory, concentration,* and *delayed recall*) along with a total internal consistency measure for the entire SAC assessment.

Table 10

Cronbach's Alpha Classifications

Cronbach's Alpha	Internal Consistency				
$\alpha \ge .9$	Excellent (High-Stakes testing)				
$0.7 \le \alpha < .9$	Good (Low-Stakes testing)				
$0.6 \le \alpha < .7$	Acceptable				
$0.5 \le \alpha < .6$	Poor				
$\alpha < .5$	Unacceptable				

Adapted from Manerikar, V., & Manerikar, S. (2015). Cronbach 's Alpha. Aweshkar Research Journal, 19(1), 117–119.

Factor naming and structure. Factor labels were determined by observing the similarities of variables within the factor loading found in Table 9. This table shows that the four assigned sections of the SCAT3 were not all represented. It was determined that *immediate memory* loaded onto two separate factors and therefore, two factors have been labeled as such. The implications of this are further discussed in the structure section of Chapter 5.

Factor 1 – Immediate Memory 1. Internal consistency for immediate memory 1 was poor ($\alpha = 0.589$) and it does not increase if any of the items are removed, as evidenced in Table 11.

Table 11

Reliability Scale for SCAT 3 Factors

		Scale			
		Scale Mean	Variance if	Corrected	
	Internal	if Item	Item	Item-Total	α , if Item
Variable Name	Consistency	Deleted	Deleted	Correlation	Deleted
Immediate Memory 1	0.589				
C1		5.88	0.219	0.428	0.543
IM3_5		5.88	0.199	0.417	0.520
IM3_2		5.88	0.210	0.351	0.544
IM3_4		5.88	0.208	0.311	0.552
IM2_2		5.90	0.187	0.273	0.568
IM2_5		5.91	0.166	0.284	0.584
IM2_4		5.89	0.191	0.304	0.552
Delayed Recall	0.512				
DR1		5.19	0.998	0.371	0.408
DR5		5.06	1.175	0.326	0.442
DR2		5.14	1.099	0.295	0.451
DR4		5.11	1.097	0.345	0.427
DR3		5.19	1.137	0.197	0.505
O_Month		4.94	1.488	0.111	0.519
O_Time		4.99	1.412	0.087	0.523
Immediate Memory 2	0.367				
IM2_3		4.69	0.380	0.266	0.313
IM1_2		4.70	0.345	0.280	0.275
IM1_3		4.71	0.336	0.246	0.283
IM1_4		4.75	0.295	0.222	0.286
O_Day		4.70	0.390	0.075	0.374
IM1_5		4.82	0.262	0.105	0.441
Concentration	0.530				
C3		2.06	0.543	0.509	0.250
C4		2.36	0.639	0.370	0.410
C2		1.77	0.893	0.316	0.480
C_Months		1.87	0.892	0.132	0.599

Factor 2 – Delayed Recall. The internal consistency of factor 2 is poor $\alpha = 0.512$. If two variables, O_Month and O_Time, were removed, the internal consistency increased but not enough to change the category strength to acceptable.

Factor 3 – Immediate Memory 2. This factor was named immediate memory 2 because there were more loadings of immediate memory than orientation. Internal consistency was unacceptable $\alpha = 0.367$. It should be noted that if the variable O_Day was removed, then the internal consistency would increase but not enough to change the "poor" status.

Factor 4 – Concentration. All but one of the concentration variables loaded onto factor 4. Internal consistency was poor $\alpha = 0.530$ as seen in Table 11. It should be noted that if the variable C_Months was removed, the internal consistency increases, resulting in the factor being acceptable.

Idealized Reliabilities. As mentioned above, the variables from each of the four assessment sections (*orientation, immediate memory, concentration,* and *delayed recall*) did not load as expected onto a single corresponding factor. Therefore, the reliabilities presented here include the reliabilities if all variables had loaded onto the expected corresponding factor. This choice was made to determine if the reliabilities of the actual assessment sections are appropriate, regardless of factor loading.

Orientation. As seen in Table 12, the internal consistency for the *orientation* subsection was negative, $\alpha = -0.084$. A negative internal reliability can mean two things, either the researcher's coding is incorrect or the participants scores are sporadic. If the researcher does not code the responses properly, for instance, instead of a "1" it should be a "5," then this phenomenon can occur. This is unlikely for this research because the values were "0" and "1." Additionally, it is possible that participants' scores show high variability, which means that the

items do not group together because they do not belong together. This is the most likely reason

because the *orientation* variables loaded on all four factors.

Table 12

Reliability Scale for SCAT3 Idealized Factors

		Scale				
		Scale Mean	Variance if	Corrected		
	Internal	if Item	Item	Item-Total	α , if Item	
Variable Name	Consistency	Deleted	Deleted	Correlation	Deleted	
Orientation	-0.084					
O_Day		3.91	0.086	-0.070	0.009	
O_Date		3.88	0.102	-0.008	100	
O_Month		3.87	0.122	-0.027	078	
O_Year		3.92	0.077	-0.035	068	
O_Time		3.87	0.122	-0.027	078	
Immediate Memory	0.485					
IM1_1		13.56	0.743	0.023	0.490	
IM1_2		13.58	0.663	0.230	0.453	
IM1_3		13.59	0.647	0.226	0.452	
IM1_4		13.63	0.610	0.182	0.471	
IM1_5		13.70	0.527	0.192	0.494	
IM2_1		13.56	0.753	-0.025	0.491	
IM2_2		13.58	0.678	0.189	0.463	
IM2_3		13.57	0.694	0.261	0.456	
IM2_4		13.58	0.659	0.310	0.438	
IM2_5		13.60	0.608	0.345	0.416	
IM3_1		13.56	0.748	0.005	0.491	
IM3_2		13.56	0.719	0.153	0.474	
IM3_3		13.56	0.748	0.005	0.491	
IM3_4		13.57	0.714	0.153	0.473	
IM3_5		13.57	0.699	0.233	0.460	
Concentration	0.503					
C1		2.69	1.141	0.077	0.530	
C2		2.77	0.910	0.311	0.436	
C3		3.06	0.554	0.512	0.231	
C4		3.35	0.652	0.371	0.374	
C_Months		2.87	0.905	0.136	0.537	
Delayed Recall	0.537					
DR1		3.25	0.901	0.358	0.444	
DR2		3.20	0.989	0.294	0.486	
DR3		3.25	1.007	0.217	0.537	
DR4		3.17	0.997	0.332	0.464	
DR5		3.12	1.061	0.329	0.472	

Immediate Memory. Of the 15 *immediate memory* variables six loaded onto factor 1. As evidenced in Table 12, the internal consistency of the 15 variables was unacceptable ($\alpha = 0.485$). Note that if five of the variables were removed (IM1_1, IM1_5, IM2_1, IM3_1, and IM3_3), internal consistency increases, but does not change the category strength.

Concentration. The internal consistency for *concentration*, as seen in Table 12, is poor $(\alpha = 0.503)$. If the two variables of C1 and C_Months were removed, internal consistency improves but not in category strength.

Delayed Recall. The internal consistency was poor ($\alpha = 0.537$).

Total. The internal consistency for all variables in the SAC portion of SCAT3 was poor ($\alpha = 0.525$) as is evidenced in Table 13. It should be noted that if seven items were removed, the internal consistency increases but remains poor.

Table 13

Reliability Scale for SCAT3 All SAC Items

Internal Consistency ($\alpha = 0.525$)						
	,		Corrected			
	Scale Mean if	Scale Variance	nce Item-Total α , if Item			
Item $(N = 30)$	Item Deleted	if Item Deleted	Correlation	Deleted		
O_Day	26.14	4.013	0.071	0.524		
O_Date	26.12	4.070	0.048	0.525		
O_Month	26.10	4.079	0.141	0.522		
O_Year	26.15	4.016	0.060	0.526		
O_Time	26.10	4.137	-0.065	0.529		
IM1_1	26.10	4.110	0.018	0.526		
IM1_2	26.12	4.036	0.089	0.522		
IM1_3	26.13	4.004	0.107	0.520		
IM1_4	26.17	3.948	0.102	0.522		
IM1_5	26.24	3.818	0.128	0.520		
IM2_1	26.10	4.126	-0.022	0.527		
IM2_2	26.12	4.005	0.145	0.518		
IM2_3	26.11	4.047	0.149	0.519		
IM2_4	26.12	3.977	0.216	0.513		
IM2_5	26.13	3.924	0.204	0.511		
IM3_1	26.10	4.118	0.003	0.527		
IM3_2	26.10	4.050	0.164	0.519		
IM3_3	26.10	4.118	0.003	0.527		
IM3_4	26.11	4.056	0.126	0.521		
IM3_5	26.11	4.037	0.171	0.518		
C1	26.10	4.084	0.124	0.522		
C2	26.18	3.854	0.172	0.512		
C3	26.47	3.411	0.268	0.491		
C4	26.76	3.367	0.309	0.480		
C_Months	26.28	3.753	0.145	0.518		
DR1	26.35	3.500	0.266	0.492		
DR2	26.30	3.742	0.140	0.519		
DR3	26.35	3.751	0.108	0.528		
DR4	26.26	3.573	0.287	0.489		
DR5	26.21	3.769	0.197	0.507		

Exploratory Factor Analysis for SCAT5

A similar EFA was conducted for the SAC portion of the SCAT5, utilizing the same

criteria outline for the SCAT3. An EFA was conducted for the orientation, immediate memory,

concentration, and *delayed recall* subsections of the SCAT5 for a sample of 395 NCAA Division III collegiate athletes.

Multicollinearity was the first assumption assessed in the EFA process. This was assessed by using the determinant which was reported as .000003. If the determinant is above .00001, factorization can occur. Therefore, multicollinearity does occur for this data set and the assumption was not met. Multicollinearity within a dataset can be a result of high correlations existing between the variables and there is the potential that variables exist that measure exactly the same thing (Mertler & Vannatta, 2010). Upon inspection of the correlation matrix (see Appendix F) no correlations were above 0.85. However, numerous correlations were observed below 0.15, meaning there were correlations among the variables that were not sufficient to show a lack of multicollinearity. Additionally, it was determined that at least one correlation coefficient was greater than 0.3 for all variables (see Appendix F) and none were highly correlated. Therefore, there was conflicting data on if the multicollinearity assumption was met.

The second assumption states that Bartlett's test of sphericity is statistically significant. The SCAT5 EFA showed Bartlett's test of sphericity ($\chi^2(1225) = 4767.42, p < .0000$) to be statistically significant. The null hypothesis is that the correlation matrix and identity matrix are identical. Having identical matrices is a problem since it practically means that there are not a sufficient number of correlations and therefore structure to the underlying latent variables. Given that Bartlett's test was statistically significant, the null hypothesis was rejected, implying that the two matrices are indeed different. This indicated that the data can be factorized for structure or dimension.

The last assumption states that a moderate overall KMO measure should be .60 or above. This EFA reported a 0.69 which is near "middling" by some standards (Dziuban & Shirkey, 1974) and therefore the assumption is met. Even though the determinant was low, an EFA was still conducted based on Bartlett's test of sphericity being significant and the higher KMO value which implies that structure does exist.

EFA revealed that 17 factors had eigenvalues greater than one. Based on the four sections of the SCAT5 assessment (*orientation, immediate memory, concentration,* and *delayed recall*) the EFA was conducted using four factors which accounted for 9.78%, 6.65%, 5.89%, and 5.43% of the total variance, respectively. These values can be viewed in Table 14. A scree plot is used to determine the number of components used to conduct the EFA and gives a graphical depiction that should bend and level off (Mertler & Vannatta, 2010). Figure 2 depicts the scree plot, which did not accurately depict the four factors because there was no true bend in the line and no leveling off occurred. This made it difficult to determine if four factors were the best choice for the SCAT5 assessment EFA because no true bend existed. Additionally, since no leveling of the line occurred it was difficult to determine if the remaining factors were equal in size (Mertler & Vannatta, 2010). These issues are discussed further in the structure section of Chapter 5.

The four-factor solution explained 26% of the total variance, therefore an oblique rotation was applied to determine if changes to factor loading occurred. The variables did not load in the way that the assessment tool intended. This means that not all the *orientation, immediate memory, concentration,* and *delayed recall* questions loaded into separate factors. In many of the factors, there were multiple variables from two or more assessment categories. The factor loadings of the rotated solution can be found in Table 15. All factor loadings less than 0.2 were suppressed when the EFA was conducted, which accounts for all blanks in Table 15.

Table 14

Eigenvalues for SCAT5 EFA	using Principle	Component	Analysis
---------------------------	-----------------	-----------	----------

Factor	Initial Eigenvalues		Extraction Loadings			Rotation	
	Total	% Variance	Cumulative %	Total	% Variance	Cumulative %	Total
1	4.895	9.789	9.789	4.895	9.789	9.789	3.306
2	3.327	6.654	16.444	3.327	6.654	16.444	2.189
3	2.716	5.431	21.875	2.716	5.431	21.875	2.041
4	2.055	4.110	25.985	2.055	4.110	25.985	2.038



Figure 2. SCAT5 scree plot from principle component analysis and oblique rotation.
Table 15

Item	Factors				
(N = 30)	F1	F2	F3	F4	
IM2_2	0.552				
DR4	0.513				
IM3_4	0.481				
IM3_3	0.474				
IM2_4	0.467				
IM3_5	0.464				
DR5	0.432				
IM1_2	0.432				
IM1_1	0.429				
IM2_1	0.414				
IM1_3	0.401				
IM2_3	0.394				
C4	0.380				
IM3_2	0.375				
IM2_5	0.352				
DR3	0.323				
C3	0.296				
DR2	0.286				
IM3_10		0.679			
IM1_10		0.610			
IM2_10		0.607			
IM2_9		0.566			
DR10		0.514			
IM3_9		0.451			
IM2_8		0.427			
IM1_9		0.425			
IM1_8		0.353			
DR9		0.349			
C_Months		0.251			

Factor Loading using Structure Matrix for SCAT5

 $\label{eq:component} \mbox{Extraction Method: Principal Component Analysis. Rotation Method: Oblimin with Kaiser Normalization. Factor loadings < 0.2 were suppressed.$

Table 15 (continued)

Item		Factors				
(N = 30)	F1	F2	F3	F4		
O_Month			-0.902			
O_Year			-0.849			
O_Day			-0.714			
C1			-0.641			
O_Time			-0.593			
O_Date			-0.376			
IM3_1			-0.373			
C2			-0.262			
DR7				-0.564		
IM3_7				-0.562		
DR6				-0.499		
IM3_6				-0.478		
IM2_6				-0.460		
DR8				-0.438		
IM1_4				-0.427		
IM3_8				-0.413		
IM1_6				-0.383		
IM2_7				-0.336		
IM1_7				-0.311		
DR1				-0.294		
IM1 5						

Factor Loading using Structure Matrix for SCAT5

Extraction Method: Principal Component Analysis. Rotation Method: Oblimin with Kaiser Normalization. Factor loadings < 0.2 were suppressed.

Reliabilities for SCAT5

Cronbach's alpha was utilized to examine the internal consistency of the SAC portion of the SCAT5. The following sections explore the reliabilities and factor name assignments. Additionally, this section is broken into two categories because the variables did not load on the exact factors as outlined in the SCAT5 assessment. Therefore, the first section reports the reliabilities for the variables that loaded on the factors during the EFA and the second section reports the idealized reliabilities. The idealized reliabilities are the result of the variables if they were divided into their assigned sections (i.e. *orientation, immediate memory, concentration,* and *delayed recall*) along with a total internal consistency measure for the entire SAC assessment.

Factor naming and structure. Factor labels were determined by observing the similarities of variables within the factor loading in Table 15. Table 15 demonstrates the four assigned sections of the SCAT3 were not all represented. It was determined that *immediate memory* loaded onto two separate factors and therefore, two factors were labeled as such. The implications for this are further discussed in the structure section of Chapter 5.

Factor 1 – Immediate Memory 1. Of the 30 *immediate memory* variables, 12 loaded onto factor 1. The variables that loaded were the first five words in each list. As indicated in Table 16, the internal consistency was good ($\alpha = 0.722$).

Table 16

Reliability Scale for SCAT 5 Factors

	Scale				
		Scale Mean	Variance if	Corrected	
	Internal	if Item	Item	Item-Total	α , if Item
Variable Name	Consistency	Deleted	Deleted	Correlation	Deleted
Immediate Memory 1	0.722				
IM2_2		12.78	8.672	0.371	0.705
DR4		12.87	8.463	0.385	0.702
IM3_4		12.82	8.639	0.348	0.706
IM3_3		12.80	8.644	0.369	0.705
IM2_4		12.92	8.441	0.364	0.704
IM3_5		12.77	8.765	0.341	0.707
DR5		12.89	8.610	0.313	0.709
IM1_2		12.79	8.818	0.300	0.711
IM1_1		12.68	9.067	0.353	0.710
IM2_1		12.70	9.028	0.317	0.711
IM1_3		13.05	8.488	0.311	0.710
IM2_3		12.86	8.746	0.277	0.713
C4		13.20	8.514	0.304	0.711
IM3_2		12.74	8.990	0.270	0.714
IM2_5		12.91	8.741	0.253	0.716
DR3		12.91	8.806	0.230	0.718
C3		12.93	8.810	0.219	0.719
DR2		12.84	8.952	0.202	0.720
Immediate Memory 2	0.693				
IM3_10		6.10	5.274	0.526	0.646
IM1_10		6.38	5.277	0.388	0.664
IM2_10		6.21	5.300	0.420	0.659
IM2_9		6.19	5.286	0.440	0.656
DR10		6.27	5.359	0.368	0.668
IM3_9		6.14	5.504	0.363	0.669
IM2_8		6.34	5.504	0.286	0.682
IM1_9		6.33	5.597	0.246	0.689
IM1_8		6.55	5.628	0.252	0.687
DR9		6.35	5.488	0.292	0.681
C_Months		6.07	5.937	0.183	0.694
Orientation	0.673				
O_Month		4.84	0.280	0.747	0.585
O_Year		4.85	0.278	0.642	0.592
O_Day		4.86	0.261	0.523	0.596
C1 .		4.85	0.298	0.435	0.637
O_Time		4.87	0.250	0.389	0.639
O_Date		4.92	0.214	0.268	0.772

Table 16 (continued)

			Scale		
		Scale Mean	Variance if	Corrected	
	Internal	if Item	Item	Item-Total	α , if Item
Variable Name	Consistency	Deleted	Deleted	Correlation	Deleted
Delayed Recall	0.649				
DR7		5.92	5.527	0.384	0.612
IM3_7		5.80	5.540	0.422	0.606
DR6		5.97	5.548	0.369	0.614
IM3_6		5.90	5.546	0.380	0.612
IM2_6		6.03	5.689	0.305	0.626
DR8		5.97	5.720	0.292	0.629
IM1_4		6.03	5.867	0.227	0.641
IM3_8		5.82	5.864	0.258	0.635
IM1_6		6.30	6.079	0.218	0.640
IM2_7		5.96	5.919	0.205	0.645
IM1_7		6.11	5.922	0.214	0.643
DR1		5.66	6.148	0.207	0.642

Reliability Scale for SCAT 5 Factors

Factor 2 – Immediate Memory 2. All words towards the end of the *immediate memory* list loaded on factor 2. Only one *concentration* variable loaded on this factor and the remainder were spread out among the other factors. As evidenced in Table 16, the internal consistency for *immediate memory 2* was acceptable ($\alpha = 0.693$). It is important to note that if the C_Months variable was removed the internal consistency would increase slightly but not enough to change the category strength.

Factor 3 – Orientation. All five of the *orientation* variables loaded onto factor 3. Internal consistency was acceptable ($\alpha = 0.673$) as indicated in Table 16. It is important to note that if the date question was removed, the internal consistency would increase to good ($\alpha = 0.772$) because eliminating some variables may help improve the internal consistency.

Factor 4 – Delayed Recall. Only four of the 10 *delayed recall* variables loaded on factor 4. The others were dispersed among the first two factors with four variables loading on the first factor and two variables loading on the second factor. The other factors that loaded in this area were *immediate memory* variables in the middle of the word list. The internal consistency presented in Table 16 and was acceptable ($\alpha = 0.649$).

Idealized Reliabilities. As mentioned above, the variables from each of the four assessment sections (*orientation, immediate memory, concentration,* and *delayed recall*) did not load as expected onto a single corresponding factor. Therefore, the reliabilities presented here include the reliabilities if all variables had loaded onto the expected corresponding factor. This choice was made to determine if the reliabilities of the actual assessment sections are appropriate, regardless of factor loading.

Orientation. As indicated in Table 17, the internal consistency is acceptable for the *orientation* subsection ($\alpha = 0.637$). It should be noted that if the O_Day variable within the *orientation* section is removed, the internal consistency would increase to good ($\alpha = 0.752$).

Table 17

		Scale			
		Scale Mean	Variance if	Corrected	
	Internal	if Item	Item	Item-Total	α , if Item
Variable Name	Consistency	Deleted	Deleted	Correlation	Deleted
Orientation	0.637				
O_Day		3.93	0.165	0.267	0.752
O_Date		3.87	0.213	0.504	0.541
O_Month		3.85	0.232	0.700	0.536
O_Year		3.88	0.202	0.377	0.591
O_Time		3.86	0.230	0.603	0.544
Immediate Memory	0.617				
IM1_1		18.94	13.661	0.194	0.610
IM1_2		19.05	13.617	0.108	0.615
IM1_3		19.31	13.271	0.151	0.612
IM1_4		19.41	13.334	0.131	0.614
IM1_5		19.46	13.726	0.025	0.626
IM1_6		19.68	13.639	0.089	0.617
IM1_7		19.49	13.469	0.099	0.618
IM1_8		19.54	13.503	0.096	0.618
IM1_9		19.31	13.663	0.041	0.624
IM1_10		19.37	13.360	0.123	0.615
IM2_1		18.96	13.519	0.229	0.607
IM2_2		19.05	13.381	0.196	0.608
IM2_3		19.12	13.417	0.146	0.612
IM2_4		19.19	13.148	0.209	0.606
IM2_5		19.17	12.982	0.264	0.600
IM2_6		19.41	12.867	0.264	0.599
IM2_7		19.34	13.249	0.155	0.612
IM2_8		19.32	12.874	0.263	0.599
IM2_9		19.17	13.241	0.184	0.608
IM2_10		19.20	13.408	0.128	0.614
IM3_1		18.92	13.755	0.183	0.612
IM3_2		19.00	13.536	0.172	0.610
IM3_3		19.06	13.362	0.195	0.608
IM3_4		19.09	13.185	0.241	0.603
IM3_5		19.04	13.217	0.269	0.602
IM3_6		19.28	12.715	0.316	0.593

Reliability Scale for SCAT5 Idealized Factors

Table 17 (continued)

			Scale		
		Scale Mean	Variance if	Corrected	
	Internal	if Item	Item	Item-Total	α , if Item
Variable Name	Consistency	Deleted	Deleted	Correlation	Deleted
Immediate Memory	0.617				
IM3_7		19.18	12.906	0.285	0.597
IM3_8		19.20	12.912	0.277	0.598
IM3_9		19.12	13.219	0.211	0.606
IM3_10		19.09	13.257	0.216	0.605
Concentration	0.453				
C1		2.85	1.040	0.158	0.461
C2		2.92	0.864	0.283	0.383
C3		3.15	0.616	0.340	0.310
C4		3.42	0.604	0.305	0.349
C_Months		3.01	0.812	0.183	0.438
Delayed Recall	0.491				
DR1		5.74	3.267	0.319	0.436
DR2		5.81	3.315	0.225	0.458
DR3		5.87	3.326	0.181	0.471
DR4		5.84	3.361	0.176	0.473
DR5		5.86	3.337	0.181	0.471
DR6		6.06	3.195	0.217	0.460
DR7		6.01	3.228	0.202	0.465
DR8		6.05	3.129	0.257	0.446
DR9		6.04	3.349	0.129	0.490
DR10		5.95	3.297	0.171	0.475

Reliability Scale for SCAT5 Idealized Factors

Immediate Memory. The internal consistency evidenced in Table 17 was acceptable ($\alpha = 0.617$). It should be noted that if *immediate memory* variables 1_5 and 1_9 were removed, the internal consistency would increase but would not change in relation to category strength.

Concentration. The internal consistency evidenced in Table 17 is unacceptable for the concentration section of the SCAT5 ($\alpha = 0.453$). If the first *concentration* variable of C1 was removed, the internal consistency would increase but not enough to change the strength category.

Delayed Recall. The internal consistency evidenced in Table 17 was unacceptable ($\alpha = 0.491$).

Total. The internal consistency evidenced in Table 18 was good ($\alpha = 0.764$) for all variables in the SAC portion of SCAT5. If three items (IM1_5, IM1_8, IM1_9) were removed, the internal consistency increases but remains as good internal consistency.

Table 18

Reliability Scale for SCAT5 All SAC Items

Internal Consistency ($\alpha = 0.764$)					
-	Corrected Item-				
	Scale Mean if	Scale Variance	Total	α , if Item	
Item $(N = 50)$	Item Deleted	if Item Deleted	Correlation	Deleted	
O_Day	34.22	33.797	0.158	0.762	
O_Date	34.16	33.982	0.226	0.762	
O_Month	34.15	34.034	0.328	0.762	
O_Year	34.17	34.007	0.154	0.762	
O_Time	34.15	34.046	0.272	0.762	
IM1_1	34.20	33.683	0.228	0.760	
IM1_2	34.31	33.641	0.134	0.763	
IM1_3	34.57	33.078	0.184	0.762	
IM1_4	34.67	33.110	0.176	0.762	
IM1_5	34.72	33.846	0.049	0.767	
IM1_6	34.94	33.634	0.126	0.763	
IM1_7	34.75	33.471	0.118	0.764	
IM1_8	34.80	33.607	0.099	0.765	
IM1_9	34.57	33.692	0.076	0.766	
IM1_10	34.63	33.321	0.139	0.764	
IM2_1	34.22	33.534	0.238	0.760	
IM2_2	34.31	33.290	0.217	0.760	
IM2_3	34.38	33.369	0.165	0.762	
IM2_4	34.45	32.750	0.266	0.758	
IM2_5	34.44	32.790	0.263	0.758	
IM2_6	34.67	32.612	0.265	0.758	
IM2_7	34.60	33.246	0.153	0.763	
IM2_8	34.58	32.599	0.269	0.758	
IM2_9	34.44	33.043	0.213	0.760	
IM2_10	34.46	33.259	0.167	0.762	
IM3_1	34.18	33.864	0.205	0.761	
IM3_2	34.26	33.494	0.202	0.761	
IM3_3	34.32	33.330	0.200	0.761	
IM3_4	34.35	32.968	0.266	0.758	
IM3_5	34.30	33.102	0.272	0.758	
IM3_6	34.54	32.203	0.346	0.754	
IM3_7	34.44	32.364	0.343	0.755	
IM3_8	34.46	32.645	0.283	0.757	

0 - (1)

Table 18 (continued)

1 0

Reliability Scale for SCAT5 All SAC Items

Internal Consistency ($\alpha = 0.764$)						
			Corrected Item-			
	Scale Mean if	Scale Variance	Total	α , if Item		
Item $(N = 50)$	Item Deleted	if Item Deleted	Correlation	Deleted		
IM3_9	34.38	32.907	0.260	0.758		
IM3_10	34.35	33.039	0.251	0.759		
C1	34.15	33.955	0.350	0.761		
C2	34.22	33.578	0.228	0.760		
C3	34.46	32.985	0.219	0.760		
C4	34.72	32.394	0.309	0.756		
C_Months	34.31	33.272	0.218	0.760		
DR1	34.30	32.953	0.301	0.757		
DR2	34.36	32.973	0.255	0.758		
DR3	34.43	32.977	0.228	0.760		
DR4	34.39	32.717	0.293	0.757		
DR5	34.42	32.660	0.296	0.757		
DR6	34.62	32.339	0.314	0.756		
DR7	34.56	32.485	0.291	0.757		
DR8	34.61	32.279	0.325	0.755		
DR9	34.60	33.098	0.179	0.762		
DR10	34.51	32.575	0.283	0.757		

To summarize the findings of RQ1, the SCAT3 and SCAT5 factor structure loaded differently than what the original SAC researchers intended. This means that the intended factors of *orientation, immediate memory, concentration,* and *delayed recall* did not load the variables corresponding to their intended subsection. This indicates that the structure of the assessment needs further scrutiny to determine if there are other latent variables that are being assessed. Additionally, the reliabilities of the subsections and overall assessment are substantially increased for the SCAT5, compared to SCAT3. This means that the increase in variables does appear to have strengthened the internal consistency of the assessment. The next section of this chapter reports the findings pertaining to RQ2.

Floor and Ceiling Proportions

The SCAT3 (N = 224) showed that *immediate memory* scores ranged from 9 to 15 (M = 14.53, SD = .893) and no participants (x = 0) received a floor effect on the *immediate memory* section. Within the *delayed recall* portion, the range was from 0 to 5 (M = 4.04, SD = 1.143) with 2.3% of the participants scoring a zero (N = 5) or floor effect on this portion of the assessment. In contrast, 68.8% (N = 154) of participants received a ceiling effect (x = 15) on the *immediate memory* portion and 43.8% (N = 98) did on the delayed memory portion (x = 5). This means that there was a large section of the sample that received a maximal score or ceiling effect on these sections of the SCAT3.

The SCAT5 (N = 203) showed that *immediate memory* scores ranged from 12 to 29 (M = 19.85, SD = 3.562) with no participants (x = 0) receiving a floor effect on the *immediate memory* section. Within the *delayed recall* portion, the range was from 0 to 10 (M = 6.62, SD = 1.945) with 0.5% of the participants scoring a zero (N=1) or floor effect on this portion of the assessment. Similar to the floor effect, no participants exhibited a ceiling effect (x = 30) on the *immediate memory* portion. A ceiling effect was present 3.9% (N = 8) for the *delayed memory* portion (x = 10). This means that very few participants, if any, exhibited a floor or ceiling effect on the SCAT5.

A Chi-square analysis was run to determine the statistically significant differences for floor and ceiling frequencies across the SCAT3 and SCAT5 *immediate memory* and *delayed recall* sections. Statistical significance was present for *immediate memory* ceiling $(\chi^2(1,427) = 218.290, p < .0000)$ and *delayed recall* ceiling $(\chi^2(1,427) = 90.43, p < .0000)$. No data could be computed for *immediate memory* floor because no participant received a zero on this portion of the assessment. The *delayed recall* floor was not significantly different $(\chi^2(1, 427) = 2.33, p = .127)$. This means that there were differences in the participants that received a ceiling effect in both subsections of the test, suggesting that the SCAT5 had a smaller percentage of maximal scores.

Conclusion

The EFA analysis of the SCAT3 and SCAT5 showed conflicting evidence of the appropriate factor loading based on the four portions of the assessment: *orientation, immediate memory, concentration,* and *delayed recall.* This indicates that the variables provided did not load onto the expected factors and therefore, the test may not be testing what it intends. There is evidence to suggest that the reliability of the assessments is questionable yet there is some improvement to the reliability of the SCAT5 over the SCAT3 because of the new 10-word list format. Additionally, the SCAT5's overall internal consistency increased significantly over the SCAT3. The ceiling effect present in the *immediate memory* and *delayed recall* portions of the test was significantly improved for the SCAT5, yet the floor effects between the two versions remained unchanged. This indicates that participants taking the SCAT5 will be less likely to earn a maximum score on these two subsections of the assessment. Therefore, the assessment seems to provide enough variables to appropriately measure ability in these areas.

Chapter 5

Conclusion

Assessment tools used for the detection of sports-related concussions (SRCs) are essential to athletic trainers and other allied healthcare professionals. Many different assessment tools have been created over the last 20 years and continue to be revised. One of these assessment tools, the Sport Concussion Assessment Tool (SCAT), was recently revised in April of 2017 from the SCAT3 to the SCAT5. The main changes were to the *immediate memory* and *delayed recall* portions of the assessment where the 5-item word list was changed to an optional 10-item word list. With the publication of the SCAT5, this optional 10-item list was provided free of charge to the public. Assessment administrators are not required to utilize the 10-item list but the medical software utilized for the historical dataset only provides the 10-item word list when a SCAT is administered. To this author's knowledge, no published data on the structure of the SCAT5 assessment and its ceiling effects exists.

Subsequently, this study examined the differences in the factor structure and internal reliability between the SCAT3 and SCAT5. The 2017 revision of the Standardized Assessment of Concussion (SAC) portion of the assessment was specifically addressed. This was accomplished through an exploratory factor analysis (EFA) that utilized a principle component analysis with an oblique rotation. Additionally, floor and ceiling effects were analyzed using an independent sample and chi-square to determine if differences in the proportions exist between SCAT versions in the *immediate memory* and *delayed recall* subsections of the assessment.

This chapter explains the implications of this study, including insights into test administration issues and participation, why some EFA assumptions were not met, and why the subsections (*orientation, immediate memory, concentration,* and *delayed* recall) did not load as expected on either version of the SCAT. The ceiling effects are also compared to other historical data. Limitations and future research ideas are also discussed.

Discussion of Findings

This section is partitioned into two: one is focused on the findings as they relate to the EFA on each of the SCAT3 and SCAT5 assessments, and the second is focused on the floor/ceiling effects with respect to the comparison across assessments. It is essential to note that this researcher found no published research utilizing an EFA or confirmatory factory analysis (CFA) on the SAC portion of the SCAT, regardless of assessment version. The dataset for this study was only for baseline testing and therefore only applies to the SCAT assessment tool in this pre-injury testing capacity.

Test administration. There are many variables specific to the test-administration procedures (i.e. time-of-day/season, assessor training, bias) that can create random variance across test-sessions within the dataset. Such randomness can add construct-irrelevant variance to the error terms in the composite scores of the assessment results. This section discusses issues in regard to test sessions, test proctoring, testing time of day and testing environment. Participants were tested annually at three different sessions: either fall sports screenings, all other sports screenings, and upon entrance into the sport, whether by transfer or participation. Therefore, random variances can occur because participants are tested at different times throughout the year. Other issues potentially affecting test outcomes include differences among individual test administrators, such as clarity of voice, or timing of word, question, or number presentation, or errors in reporting. Some test administrators are learning the tool for the first time while others have over a decade of experience. All administrators are given a tutorial by a highly experienced proctor prior to every testing session. The day and time of each screening varies annually and is

dictated by the athletics administration. This can affect both the energy levels and the cognitive abilities of participants. Additionally, on screening days, there were multiple stations operating throughout the gymnasium and the volume fluctuated. Therefore, the testing environment was not quiet and may have affected the cognitive ability of participants. The random variance resulting from these test administration issues likely contributed to the poor factor structures and poor to moderate reliabilities. Future research that better controls for test administration issues may find an improvement in the factor structures and reliabilities.

Participants. The total sample size for both EFAs was similar and gender was approximately equal with both versions. Sports participation was comparable between samples with participation in football representing the largest subset in SCAT results. One difference between the two samples was the number of track-and-field participants and tennis participants both significantly increased from the SCAT3 to the SCAT5. It seems as though this could have been the result of a policy change by the university which required all track-and-field and tennis athletes to engage in baseline concussion assessments. Prior to 2017, only select track-and-field participants (pole value, steeple chase, and hurdles) were required to be assessed. Participants were not included in the SCAT5 sample if administrators did not mark the assessment as "baseline" on the medical software when the assessment was administered. Therefore, it is possible this study worked with an underrepresented sample of multiple sports (i.e. lacrosse, softball, basketball, and soccer) for the SCAT5. Also, the average age of the participants was a year higher for the SCAT3 participants than the SCAT5 participants, yet the standard deviations were almost identical, which accounts for the differences in mean age scores.

The independent samples used to determine if floor/ceiling effects existed had a gender distribution of approximately one-third female, for the SCAT3 with a slightly larger sample of

females for the SCAT5. Differences between the two independent groups with respect to sport of participation were primarily due to the inclusion of tennis on the SCAT5 and a large decrease in lacrosse participants from the SCAT3 to SCAT5. Variability in sports participation is mainly dependent on the number of new and exiting athletes. The average age of participants for the independent sample was a year and a half above the SCAT5 participants' average age. This difference was primarily due to a larger number of SCAT3 participants in the 24 and above grouping, as well as the fact that none fell in the under 19-year grouping. The age group differences may have resulted in a larger disparity between ceiling effects than exists in other samples. As age increases, so do cognitive abilities (Rushton & Ankney, 1996) and potentially the number of testing experiences. Therefore, it is possible that the ceiling effects reported in this study for the SCAT3 are not as elevated for younger populations and that there would have been more participants who achieved a ceiling effect on the SCAT5 if the average age was identical.

Structure of SCAT3 and SCAT5. The following section explores the similarities and differences between the factor structure of the SCAT3 and SCAT5. Specifically, it addresses the issues of violation of assumptions, unexpected factor loading, and reliabilities.

Violation of assumptions. To create a clear picture of the structure of the various versions of the SCAT, it was necessary to examine how neither version of the SCAT met all the assumptions required for an EFA. Only two of the three assumptions for conducting an EFA for the SCAT3 and SCAT5 were observed. Both passed Bartlett's test of sphericity, but the SCAT3 violated the Kaiser-Meyer-Olkin (KMO) assessment and the SCAT5 violated multicollinearity (as observed by the determinant of the correlation matrix). Therefore, an argument could be made that EFA was not the appropriate statistical method to evaluate these assessments. However, the decision to utilize the EFA was made because two assumptions were met in each

instance and neither violated the Bartlett's test for sphericity. Bartlett's is key, because statistical significance implies that the correlation matrix and the identity matrix are not identical. This can be observed by finding the diagonal of the identity matrix. If all the diagonal correlations are 1's and the off diagonals are 0's, this indicates that each variable is only correlated with itself. If this test fails, it shows that the variables may not have been correlated with themselves. Therefore, because the Bartlett's test was passed, this indicates that the data were able to be factorized for structure or dimension.

Factor loading. When comparing the factor structure of the SCAT3 and SCAT5, there are some similarities in total variance and scree plot attributes. The factor structure accounts for approximately 26% of the total variance, which indicates that the factor structure has relatively low explanatory power. As a result, a large portion of the variance within the data cannot be attributed to the actual subsections (orientation, immediate memory, concentration, and delayed *recall*). This is problematic because it is unclear what other factors play a role in the assessments both as an instrument and as subsections. Moving from the 5-item word list to the 10-item word list did not change the total variance between versions, meaning that the structural changes between the SCAT3 and SCAT5 did not improve the explanatory value of the factor structure as it related to the *orientation*, *immediate memory*, *concentration*, and *delayed recall* constructs. The weak total explanatory variance is concerning because it indicates that there is relatively little explanatory power for what the retained factors are able to explain when considering the total number of items. Further research should focus on determining if the explanatory variance can be increased, perhaps by changing either the questions or the word choices. If the explanatory variance increases, then healthcare practitioners can be more confident that the assessment is really testing what it claims.

The scree plots show the appropriate number of factors to be extracted during the EFA by the presence of a bend or elbow in the line on the graph. Both versions were similar in shape but differed in the number of possible bends or eligible factors to be extracted. Both scree plots had no true "bend or elbow" (Mertler & Vannatta, 2010, p. 234) with multiple inflection points for both versions. The SCAT3 showed bends at factors 2 and 5 while the SCAT5 at factors 4, 8, and 14 (see Figure 1 and Figure 2). Therefore, it was difficult to determine the appropriate number of factors to extract. This bend in the plot should have occurred at 4 in both versions to clearly show that four factors was the best option. I did not focus on these numbers to determine the factor extraction because the four subsections of the assessment should have been the four factors extracted (orientation, immediate memory, concentration, and delayed recall). It is important to note that the SCAT5 first bend is at 4, which is the number of factors/subsections that were extracted. It is possible that as the number of items on the assessment increase, the factor structure becomes stronger. Additionally, the scree plot is supposed to level off after the bend and in both assessments, they continue to decline but never level off, which indicates that the variables are not loading perfectly onto the factor structure. An imperfect factor loading may call into question the psychometric properties of the SCAT and whether this test should be utilized for SRC sideline assessment. Further research should focus on determining the appropriate number of factors and possible factors that might be missing from the SCAT5 assessment.

There is a lack of clarity on how items loaded onto the four factors, inferring for instance, that the assessment says that it is testing *immediate memory* but is not. For example, when the *immediate memory* variables are recorded, they loaded onto two different factors with the implication that *immediate memory* variables are testing different constructs. Ideally, all

immediate memory variables would have loaded onto the *immediate memory* factor. The only departure from this occurred with the *delayed recall* factor for SCAT3 and *orientation* factor for SCAT5. In this case, all *delayed recall* variables loaded onto the *delayed recall* factor and the same for *orientation* for SCAT5. Therefore, there is conflicting data on the psychometric properties of both versions of the SCAT which leads to the conclusion that this assessment may not be assessing what it claims. Healthcare practitioners need to determine if this is the most accurate sideline assessment or it another tool needs to be created. The next section explores possibilities for these unique factor loadings as they relate participants' interpretation of the items.

The question that most needs to be addressed is how these items are being interpreted by the examinee. Specifically, it is possible that individuals code for the position of the item in a sequence when given lists of information to recall (Henson, 1999). The coding of the position of the word or number can increase or decrease the effectiveness of recalled information. This is suggested in the disparity of variable loadings based on their location within the word-lists. This means that the order of the words presented may affect how the participant recalls it because the word-list is repeated for three trials for the *immediate memory* subsection. For example, in the SCAT3 *immediate memory* items located at 2, 4, and 5 for the second and third trials load together while all of the variables from the first *immediate memory* trial all load onto a different factor. This could mean that the position of the word within the trial carries more weight than the first trial of the *immediate memory* utilizes short-term memory but the subsequent trials utilized long-term memory. For SCAT5, the groupings of *immediate memory* factor into location as well. In factor 1, most of the first five words are represented except for three variables. In factor 2, all but one of the last three words (i.e. words 8, 9 and 10) load and in factor 4, all words from

location 6 and 7 are present (see Table 15). The net result is that participants may use a mixture of short- and long-term memory during various subsections of this assessment. Therefore, it is possible that they know that because they will need to recall these words at the end of the screening process, they change how the words are coded in their memory, which could account for the unexpected factor loading reported in the EFA.

Another example of unexpected factor loading can be found in how the first concentration item does not load within the concentration factor but loads with immediate *memory 1* with the SCAT3. All other *concentration* items load together which means that it is the only concentration variable that deviates from the intended factor. During the concentration subsection, the participant is given multiple numbers to repeat in reverse order with three values for the first variable up to six for the last. One possible explanation for this is that the first concentration exercise is really testing immediate memory because test-takers have no other numbers to remember at this point in the assessment. After this first question, the participant is expected to 'push' the old number out of short-term memory and then integrate the new number before it can be reversed. This mental exercise requires that participants be careful that none of the old and new numbers overlap. Thus, in a way, long-term and short-term memory must coordinate to complete some of the tasks within the SCAT. So, while the SCAT purports to assess long-term and short-term memory in separate ways through different portions of the test, there appears to be a significant overlap between the tasks, making it harder to diagnose decreases in the various aspects of cognitive function.

Another example of overlap in cognitive function occurs between orientation and longterm memory as evidenced within how the SCAT3 *orientation* items align with *delayed recall*. The possibility is that certain *orientation* items are already stored in long-term memory. An example of this would be the question about the current month. When asked, "What month is it?" it is possible that participants could think in two ways about this question. One way is to just say the current month. But another way would be to consider the month in terms of the past 0-30 days, meaning that if it is more than one day into the month, participants would have to recall what the month is from previous knowledge. Therefore, it is possible that some of the *orientation* items utilize long-term memory, suggesting an inconsistent and unreliable measure of *orientation*. This implies that a healthcare practitioner cannot explicitly state that orientation, short-term or long-term memory is impaired but they could state that there is a cognitive dysfunction.

Reliabilities. Internal consistency was reported in two ways: with the factor loadings and as idealized values. This decision was made based on the lack of clarity in the factor structure matrix, therefore internal consistency was reported for each subsection and for the overall SCAT assessment. This enabled a determination of the strength of the variables' groupings following the EFA. The SCAT3 internal consistency for the immediate memory 2 factor was unacceptable with immediate memory 1, concentration, and delayed recall factors having poor classification. After producing the idealized reliabilities, there was no change in internal consistencies between immediate memory or orientation (unacceptable) and delayed recall or concentration (poor) internal consistency. It is difficult to compare the reliability of the SAC portion of the SCAT3 to previous literature because all previous literature looks only at the reliability based on those who have sustained an SRC and not exclusively with baseline testing (McCrea et al., 1998; McCrea et al., 2007). The overall reliability of the SCAT3 assessment fell within the range found in literature (.42-.71), meaning that the sample used in this study reliably responded to SCAT3

items (Guskiewicz et al., 2013). This does not mean that the reliability of the SCAT3 was good or acceptable, but just that the values for this sample showed a similarity of results.

SCAT5 reliabilities are better than SCAT3 when loaded by factors with *immediate memory 1* as good and *immediate memory 2*, *orientation*, and *delayed recall* as acceptable. When the idealized reliabilities are conducted they significantly decrease the reliabilities to unacceptable for *delayed recall* and *concentration* but remain the same for *immediate memory* and *orientation* as acceptable. This suggests that the subsections in the SCAT are not adequately grouping the variables together and therefore, the structure of the exam is questionable. The overall internal consistency for all variables surpasses the reliabilities reported for the previous version of the SCAT (Guskiewicz et al., 2013). Therefore, it may be concluded that grouping all of the variables together with the SCAT5 provides more internal consistency, not from the factor structure, but from the additional variables found within the *immediate memory* and *delayed recall* subsections of the assessment. It is important to remember that the internal consistency values reported in the literature are based on those participants who have sustained a SRC and not on baseline testing as reported in this study. Therefore, further research needs to determine if these reliabilities remain consistent for post-injury assessments with the SCAT5.

Floor and ceiling effects. There were significant differences in the ceiling effects within the immediate memory and delayed recall portions of the assessment. This indicates that the SCAT5 significantly reduces the ceiling effects seen within the SCAT3. The average score for immediate memory on the SCAT3 was near the maximum score of 15 while the average score for the SCAT5 was more than 10 points away from the maximum. These results indicate it is less likely that participants will score near the maximal threshold on the SCAT5. Within the delayed recall, the same phenomenon occurs suggesting that a ceiling effect is less likely to occur and the

assessment has increased the difficulty of the assessment enough to capture most participants maximal cognitive ability on the tasks requested within the assessment.

There are statistically significant differences between ceiling effects of *immediate memory* and *delayed recall*. This means that by increasing the number of word items on the SCAT5 assessment, participants are less likely to get the maximal score. This allows a greater chance for capturing a more accurate assessment of cognitive ability. Further research should focus on whether increasing the word-item list to an even larger number would adequately reflect the abilities of each participant or if a 10-item word list is ideal.

Limitations

There were several limitations related to this research. The use of an EFA was the main limitation. The goal of an EFA is to explore the factor structure and not to test for differences between assessment structures. The only ways to evaluate differences are to visually inspect the EFA outputs (i.e. graphs and tables) and compare internal consistency values. Therefore, EFAs are somewhat subjective, which limits conclusions to the descriptions and comparisons provided. Additionally, the principle component analysis utilized as the extraction method may not have been the best choice and other methods may have created different results (Osborne, 2014). In hindsight, a better plan would have been to use a maximal likelihood factoring.

Another limitation was that this research only examined baseline data for SCAT3 and SCAT5. No conclusions about SCAT5 usability for SRC diagnosis can be formed because the data was not compiled on participants who were currently concussed or suspected of sustaining a SRC. Additionally, the variation in testing environment may have limited the outcomes, meaning that participants may have had higher scores if they were testing in a quiet environment isolated from their peers and with one or two researchers controlling for the variability between proctors.

This study did not account for athlete history of SRC injury or diagnosis of cognitive disorders. This means that this study excludes any impact on participant scores who have a history of SRC or a cognitive disorder (e.g. post-injury data).

A final limitation was that the participants were not evaluated in either the SCAT3 or the SCAT5 assessment on the truthfulness of their responses. As a result, some participants may have intentionally performed poorly or sandbagged, which influences the results of their assessment score. The reason athletes may sandbag is to have an easier and quicker path to return-to-play following a SRC. This is possible because the scores on SRC assessments are used as a comparison to their post-injury assessments. An athlete is only cleared for participation after a concussion when all post-SRC assessments return to pre-injury baseline scores, the athlete is asymptomatic, and they are able to tolerate intense exercise without the re-occurrence of symptoms. It is possible some athletes choose to mis-represent their symptoms on the pre-test in order to guarantee a quicker return-to-play after an injury. As a result, sandbagging may lead to artificially lower assessment scores. Therefore, it is essential that SRC assessments incorporate questions to prevent sandbagging as an additional safeguard to accurately reflect deficiencies in cognition.

Implications for Practice

Athletic trainers (ATCs) are usually the first healthcare providers to diagnose a SRC. Therefore, each ATC needs to decide what decision criteria they will use to assess and clear athletes for participation following a suspected SRC. Many athletic trainers utilize the SCAT3 or SCAT5 as a portion of their concussion protocol. Based on the structural inconsistencies within both assessments that were revealed in this research, practitioners should consider whether the SCAT assessments should be included within the battery of SRC assessments. Currently, there is no universal protocol utilized by athletic trainers to diagnose a SRC; research certainly points to the need to administer multiple SRC assessments to avoid relying on a single data point (Broglio, Guskiewicz, & Norwig, 2017).

Some researchers claim that objective assessments exist to measure SRCs (Barr & McCrea, 2001; Guskiewicz et al., 2013)and others suggest that subjectivity is an overriding component of all SRC assessments (Broglio et al., 2017; McCrory et al., 2017). This study indicates the possibility that the SCAT is not effectively assessing what it claims to measure, warranting further consideration of the objective and subjective nature of assessing SRCs. Further research on ATCs' decision-making processes may be useful in this regard. Additionally, research on how political pressure from athletes, coaches, and athletic administrators affects ATCs' decisions should also be explored.

Suggestions for Future Research

Future research is essential to help athletic trainers and other allied healthcare professionals determine the best practices for sideline SRC assessment. This research should focus on determining if the total variance described by the assessment can be increased with changing either the questions or the word choice within the SCAT5. This research shows that the SAC assessment accounts for only 26% of the total explanatory variance. This is a weak percentage and needs to be explored to determine if this assessment should even be included in the SCAT5.

A closer look at the appropriate number of factor loadings and possible factors that might be missing from the SCAT5 assessment should also be explored. A deeper examination of the factor loading may reveal other latent variables present within the SCAT5 which may improve the total explanatory variance. Post-injury data collection on the SCAT5 needs to be explored to determine if the internal consistency remains strong for the overall SAC assessment. If the internal consistency is confirmed for post-injury assessment, then further exploration of how short-term and long-term memory are reflected in each SAC questions should be identified.

Finally, it would be worthwhile for researchers to explore whether adding more than a 10-item word list could capture more variance in cognitive ability, or if the 10-item list on the SCAT5 is sufficient. Future research for SCAT5 could also focus on the comparison between baseline testing and post-injury assessment following a SRC to determine if the changes to the SCAT5 actually help in diagnosis and return-to-play decisions.

Conclusion

This study focused on how the structures of the SCAT3 and SCAT5 changed with the addition of a 10-item word list, while also comparing floor/ceiling effects between the two versions of the test. This research showed that the structures of the assessment changed when adding additional words to the assessment. Additionally, the ceiling effect that exists within the SCAT3 disappears in the SCAT5 in the *immediate memory* and *delayed recall* portions of the SAC assessment. This indicates that the SCAT5 is indeed a better assessment of athletes' cognitive abilities, when the optional 10-item list is used. Overall, the internal consistency of the SCAT5 is improved over the SCAT3, although there is still concern related to the uneven loading of the variables. This indicates that the SCAT5 is not assessing what the subsections labels suggest and may not test the variables that the designers intended for the cognitive assessment.

Despite these concerns, this research suggests that all athletic trainers and allied healthcare professionals who currently use the SCAT5 in their return-to-play protocol, should be utilizing the 10-item word list to mitigate the ceiling effects present in the SCAT3. Despite the concerns present within the structure of the SAC portion of the SCAT, healthcare practitioners should still continue to make use of this sideline assessment, until a more psychometrically sound instrument for cognition is presented.

References

- Allen, B. J., & Gfeller, J. D. (2011). The immediate post-concussion assessment and cognitive testing battery and traditional neuropsychological measures: A construct and concurrent validity study. *Brain Injury*, 25(2), 179–191. Retrieved from http://doi.org/10.3109/02699052.2010.541897
- Aubry, M., Cantu, R. C., Dvorak, J., Graf-Baumann, T., Johnston, K. M., Kelly, J. P., ...
 Schamasch, P. (2002). Summary and agreement statement of the first international symposium on concussion in sport, Vienna 2001. *Clinical Journal of Sports Medicine*, *12*(1), 6–11. Retrieved from http://doi.org/10.1136/bjsm.36.1.6
- Barr, W. B., & McCrea, M. (2001). Sensitivity and specificity of standardized neurocognitive testing immediately following sports concussion. *Journal of the International Neuropsychological Society*, 7, 693–702. Retrieved from http://doi.org/10.1017/S1355617701766052
- Barth, J. T., Alves, W. M., Ryan, T. V., Macciocchi, S. N., Rimel, R. W., Jane, J. A., & Nelson,
 W. E. (1989). Mild head injury in sports: Neuropsychologica sequelae and recovery of function. In *Mild Head Injury* (pp. 257–276). New York: Oxford University Press.
- Benedict, P. A., Baner, N. V., Harrold, G. K., Moehringer, N., Hasanaj, L., Serrano, L. P., ...
 Balcer, L. J. (2015). Gender and age predict outcomes of cognitive, balance and vision
 testing in a multidisciplinary concussion center. *Journal of the Neurological Sciences*, *353*, 111–115. Retrieved from http://doi.org/10.1016/j.jns.2015.04.029

- Broglio, S. P., Cantu, R. C., Gioia, G. A., Guskiewicz, K. M., Kutcher, J., Palm, M., & McLeod, T. C. V. (2014). National athletic trainers' association position statement: Management of sport concussion. *Journal of Athletic Training*, 49(2), 245–265. Retrieved from http://doi.org/10.4085/1062-6050-49.1.07
- Broglio, S. P., Guskiewicz, K. M., & Norwig, J. (2017). If you're not measuring, you're guessing: The advent of objective concussion assessments, 52(3), 160–166. Retrieved from http://doi.org/10.4085/1062-6050-51.9.05
- Brommer, R., Fowler, J., Hons, K., Gerwing, L., & Payne, J. (2016). The utility of the KingDevick test as a sideline assessment tool for sport-related concussions: A narrative review. *Journal of Canadian Chiropractic Association*, 60(4), 322–329.
- Chin, E. Y., Nelson, L. D., Barr, W. B., McCrory, P., & McCrea, M. A. (2016). Reliability and validity of the sport concussion assessment tool-3 (SCAT3) in high school and collegiate athletes. *The American Journal of Sports Medicine*, 3(9), 1–11. Retrieved from http://doi.org/10.1177/0363546516648141
- Collie, A., Makdissi, M., Maruff, P., Bennell, K., & McCrory, P. R. (2006). Cognition in the days following concussion: Comparison of symptomatic versus asymptomatic athletes. *Journal of Neurology, Neurosurgery, and Psychiatry*, 77(2), 241–245. Retrieved from http://doi.org/10.1136/jnnp.2005.073155
- Davis, G. A. (2017). Sport concussion assessment tool 5th edition. *British Journal of Sports Medicine*, 0, 1–8. Retrieved from http://doi.org/10.1136/bjsports-2017-097506SCAT5
- Dick, R., Agel, J., & Marshall, S. W. (2007). NCAA injury surveillance system commentaries: Introduction and methods. *Journal of Athletic Training*, *42*(2), 173–182.

- Dziuban, C. D., & Shirkey, E. (1974). When is a correlation matrix appropriate for factor analysis? Some decision rules. *Psychological Bulletin*, 81(6), 358–361. Retrieved from http://doi.org/10.1037/h0036316
- Echemendia, R. J., Broglio, S. P., Davis, G. A., Guskiewicz, K. M., Hayden, K. A., Leddy, J. J.,
 ... Schneider, K. J. (2017). What tests and measures should be added to the SCAT3 and
 related tests to improve their reliability, sensitivity and/or specificity in sideline concussion
 diagnosis? A systematic review. *British Journal of Sports Medicine*, *51*(10), 895–901.
 Retrieved from http://doi.org/10.1136/bjsports-2016-097466
- Echemendia, R. J., & Julian, L. J. (2001). Mild traumatic brain injury in sports:
 Neuropsychology's contribution to a developing field. *Neuropsychology Review*, *11*(2), 69–88.
- Echemendia, R. J., Meeuwisse, W., McCrory, P., Davis, G. A., Putukian, M., Leddy, J., ...
 Herring, S. (2017). The sport concussion assessment tool 5th edition (SCAT5). *British Journal of Sports Medicine*, *51*(5), 848–850. Retrieved from http://doi.org/10.1136/bjsports-2017-097506
- Elkington, L., & Hughes, D. (2016). Australian institute of sport and Australian medical association concussion in sport position statement. Retrieved from https://ama.com.au/sites/default/files/documents/AMA_AIS_Concussion in Sport Position Statement 2015.pdf
- Gessel, L. M., Fields, S. K., Collins, C. L., Dick, R. W., & Comstock, R. D. (2007). Concussions among United States high school and collegiate athletes. *Journal of Athletic Training*, 42(4), 495–503. Retrieved from http://doi.org/10.1016/S0162-0908(08)79294-8

- Guskiewicz, K. M., Register-Mihalik, J., McCrory, P., McCrea, M., Johnston, K., Makdissi, M.,
 ... Meeuwisse, W. (2013). Evidence-based approach to revising the SCAT2: Introducing the SCAT3. *British Journal of Sports Medicine*, *47*(5), 289–293. Retrieved from http://doi.org/10.1136/bjsports-2013-092225
- Hecimovich, M., & Marais, I. (2017). Examining the psychometric properties of a sport-related concussion survey: A Rasch measurement approach. *BMC Research Notes*, *10*(1), 228.
 Retrieved from http://doi.org/10.1186/s13104-017-2559-z
- Henson, R. N. A. (1999). Positional information in short-term memory: Relative or absolute?*Memory & Cognition*, 27(5), 915–927. Retrieved from http://doi.org/10.3758/BF03198544
- Hootman, J. M., Dick, R., & Agel, J. (2007). Epidemiology of collegiate injuries for 15 sports:
 Summary and recommendations for injury prevention initiatives. *Journal of Athletic Training*, 42(2), 311–319. Retrieved from http://doi.org/10.1111/j.1600-0838.2006.00528.x
- Hyden, J., & Petty, B. (2016). Sideline management of concussion. *Physical Medicine and Rehabilitation Clinics of North America*, 27(2), 395–409. Retrieved from http://doi.org/10.1016/j.pmr.2015.12.004
- King, D., Brughelli, M., & Hume, P. (2014). Assessment, management and knowledge of sportrelated concussion: Systematic review. *Sports Medicine*, 44, 449–471. Retrieved from http://doi.org/10.1007/s40279-013-0134-x
- Kirschen, M. P., Tsou, A., Bird Nelson, S., Russell, J. A., & Larriviere, D. (2014). Legal and ethical implications in the evaluation and management of sports-related concussion. *Neurology*, 83, 352–358.

- Koscs, M., Kaminski, T. W., Swanik, C. B., & Edwards, D. G. (2009). Effects of exertional exercise on the standardized assessment of concussion (SAC) score. *Athletic Training & Sports Health Care*, 1(1), 24–30. Retrieved from http://doi.org/10.3928/19425864-20090101-01
- Ku, C. (2017). *Media portrayal of concussions in sporting matches: Influence on observers' perception, knowledge and attitude towards concussion.* University of Canterbury.
- Laerd Statistics. (2015a). Cronbach's alpha using SPSS statistics. Retrieved from https://statistics.laerd.com/
- Laerd Statistics. (2015b). Principle component analysis. Retrieved from https://statistics.laerd.com/
- Lemonda, B. C., Tam, D., Barr, W. B., & Rabin, L. A. (2017). Assessment trends among neuropsychologists conducting sport-related concussion evaluations. *Developmental Neuropsychology*, 42(2), 113–126. Retrieved from http://doi.org/10.1080/87565641.2016.1274315
- Manerikar, V., & Manerikar, S. (2015). Cronbach 's alpha. *Aweshkar Research Journal*, *19*(1), 117–119. Retrieved from http://doi.org/10.1007/bf02310555.Cronbach
- Matuszak, J. M., McVige, J., McPherson, J., Willer, B., & Leddy, J. (2016). A practical concussion physical examination toolbox: Evidence-based physical examination for concussion. *Sports Health*, 8(3), 260–269. Retrieved from http://doi.org/10.1177/1941738116641394
- McCrea, M. (2001a). Standardized mental status assessment of sports concussion. *Clinical Journal of Sport Medicine*, 11(3), 176–181. Retrieved from http://doi.org/http://dx.doi.org/10.1097/00042752-200107000-00008

- McCrea, M. (2001b). Standardized mental status testing on the sideline after sport-related concussion. *Journal of Athletic Training*, *36*(3), 274–279.
- McCrea, M., Barr, W., Guskiewicz, K., Randolph, C., Marshall, S., Cantu, R., ... Kelly, J. (2005). Standard regression-based methods for measuring recovery after sports-related concussion. *Journal of the International Neuropsychological Society*, *11*, 58–69. Retrieved from http://doi.org/10.1017/S1355617705050083
- McCrea, M., Kelly, J., & Randolph, C. (2007). *Standardized assessment of concussion: Manual for administration, scoring and interpretation* (3rd ed.). CSMi Medical Solutions.
- McCrea, M., Kelly, J., Randolph, C., Kluge, J., Bartolic, E., Finn, G., & Baxter, B. (1998). Standardized assessment of concussion (SAC): On-site mental status evaluation of an athlete. *Journal of Head Trauma Rehabilitation*, 13(2), 27–35.
- McCrory, P., Johnston, K., Meeuwisse, W., Aubry, M., Cantu, R., Dvorak, J., ... Schamasch, P. (2005). Summary and agreement statement of the second international conference on concussion in sport, Prague 2004. *The Physician and Sportsmedicine*, *33*(4), 29–44.
 Retrieved from http://doi.org/10.3810/psm.2005.04.76
- McCrory, P., Meeuwisse, W., Aubry, M., Cantu, B., Dvorak, J., Echemendia, R. J., ... Tator, C.
 H. (2013). Consensus statement on concussion in sport The 4th international conference on concussion in sport. *Clinical Journal of Sports Medicine*, 23(2), 89–117.
- McCrory, P., Meeuwisse, W., Aubry, M., Cantu, R., Dvoøák, J., & Echemendia, R. (2013). Sport concussion assessment tool 3rd edition. *British Journal of Sports Medicine*, 47(5), 259–262.

- McCrory, P., Meeuwisse, W., Dvorak, J., Aubry, M., Bailes, J., Broglio, S., ... Vos, P. E. (2017).
 Consensus statement on concussion in sport—the 5th international conference on concussion in sport held in Berlin, October 2016. *British Journal of Sports Medicine*, *51*, 838–847. Retrieved from http://doi.org/10.1136/bjsports-2017-097699
- McCrory, P., Meeuwisse, W., Johnston, K., Dvorak, J., Aubry, M., Molloy, M., & Cantu, R.
 (2009). Consensus statement on concussion in sport The third international conference on concussion in sport held in Zurich, November 2008. *The Physician and Sportsmedicine*, *37*(2), 141–159. Retrieved from http://doi.org/10.3810/psm.2009.06.1721
- McElhiney, D., Kang, M., Starkey, C., & Ragan, B. (2014). Improving the memory sections of the standardized assessment of concussion using item analysis. *Measurement in Physical Education and Exercise Science*, 18(2), 123–134. Retrieved from http://doi.org/10.1080/1091367X.2013.866558
- Mertler, C. A., & Vannatta, R. A. (2010). *Advanced and multivariate statistical methods: Practical application and interpretation* (4th ed.). Glendale: Pyrczak Publishing.
- Miller, J. R., Adamson, G. J., Pink, M. M., & Sweet, J. C. (2007). Comparison of preseason, midseason, and postseason neurocognitive scores in uninjured collegiate football players. *The American Journal of Sports Medicine*, *35*(8), 1284–1288. Retrieved from http://doi.org/10.1177/0363546507300261

Osborne, J. W. (2014). Best practices in exploratory factor analysis (Kindle).

Pachman, S., & Lamba, A. (2017). Legal aspects of concussion: The ever-evolving standard of care. *Journal of Athletic Training*, 52(3), 186–194. Retrieved from http://doi.org/10.4085/1062-6050-52.1.03

- Patricios, J., Fuller, G. W., Ellenbogen, R., Herring, S., Kutcher, J. S., Loosemore, M., ...
 Schneider, K. J. (2017). What are the critical elements of sideline screening that can be used to establish the diagnosis of concussion? A systematic review. *British Journal of Sports Medicine*, *51*(11), 888–894. Retrieved from http://doi.org/10.1136/bjsports-2016-097441
- Ragan, B. G., Herrmann, S. D., Kang, M., & Mack, M. G. (2009). Psychometric evaluation of the standardized assessment of concussion. *Athletic Training & Sports Health Care*, 1(4), 180–187. http://doi.org/10.3928/19425864-20090625-07
- Ragan, B. G., & Kang, M. (2007). Measurement issues in concussion testing. *Athletic Therapy Today*, *12*(5), 2–6.
- Rushton, J. P., & Ankney, C. D. (1996). Brain size and cognitive ability: Correlations with age, sex, social class, and race. *Psychonomic Bulletin & Review*, 3(1), 21–36. Retrieved from http://doi.org/10.3758/BF03210739
- Shehata, N., Wiley, J. P., Richea, S., Benson, B. W., Duits, L., & Meeuwisse, W. H. (2009). Sport concussion assessment tool: Baseline values for varsity collision sport athletes. *British Journal of Sports Medicine*, 43(10), 730–734. Retrieved from http://doi.org/10.1136/bjsm.2009.059832
- Short-term memory. (n.d.). Retrieved November 1, 2017, from /www.merriamwebster.com/dictionary/short-term memory

Sufrinko, A. M., Marchetti, G. F., Cohen, P. E., Elbin, R. J., Re, V., & Kontos, A. P. (2017).
Using acute performance on a comprehensive neurocognitive, vestibular, and ocular motor assessment battery to predict recovery duration after sport-related concussions. *The American Journal of Sports Medicine*, 45(5), 1187–1194. Retrieved from http://doi.org/10.1177/0363546516685061
- Yengo-Kahn, A. M., Hale, A. T., Zalneraitis, B. H., Zuckerman, S. L., Sills, A. K., & Solomon,
 G. S. (2016). The sport concussion assessment tool: A systematic review. *Neurosurgical Focus*, 40(4), 1–14. Retrieved from http://doi.org/10.3171/2016.1.FOCUS15611
- Zimmer, A., Marcinak, J., Hibyan, S., & Webbe, F. (2015). Normative values of major SCAT2 and SCAT3 components for a college athlete population. *Applied Neuropsychology:Adult*, 22(2), 132–140. Retrieved from http://doi.org/10.1080/23279095.2013.867265
- Zimmerman, M. E. (2011). Normative data. In J. S. Kreutzer, J. DeLuca, & B. Caplan (Eds.), *Encyclopedia of Clinical Neuropsychology*. New York, NY: Springer.
- Zuckerman, S. L., Kerr, Z. Y., Yengo-Kahn, A., Wasserman, E., Covassin, T., & Solomon, G. S. (2015). Epidemiology of sports-related concussion in NCAA athletes from 2009-2010 to 2013-2014. *American Journal of Sports Medicine*, 43(11), 2654–2662. Retrieved from http://doi.org/10.1177/0363546515599634

APPENDICES

Appendix A

Sport Concussion Assessment Tool 3 (SCAT3)



BACKGROUND

3

20

M	F
net	ar.
Y	N
Y	N
Y	N
Y	N
۷	N
Y	N
	Y

SCAT3 to be done in resting state. Best done 10 or more minutes post excerdee SYMPTOM EVALUATION

"You should alone yourse	if on the followi	ng sympt	torniç De	and on it	ow you	feelno	87.
	none		- 14	mod	eute -		**
Headache	0	1	2	3	4	5	6
"Pressure in head"	0	1	- 2	- 2	- 4	5	6
Neck Pain	0	1	2	2	4	5	6
Nausea or vomiting	0	1	2	2	4	5	6
Dizziness	0	1	2	2	4	5	6
Blurred vision	0	1	2	2	- 4	5	6
Balance problems	0	1	2	2	- 4	5	6
Sensitivity to light	0	1	2	2	- 4	5	6
Sensitivity to noise	0	1	- 2	- 2	4	5	6
Feeling slowed down	0	1	- 2	3	- 4	5	6
Feeling like "In a fog"	0	1	- 2	- 2	- 4	5	6
"Don't feel right"	0	1	- 2	- 2	- 4	5	6
Difficulty concentrating	0	1	- 2	- 2	- 4	5	6
Difficulty remembering	0	1	2	2	4	5	6
Fatigue or low energy	0	1	2	2	4	5	6
Confusion	0	1	2	2	4	5	6
Drowsiness	0	1	2	2	4	5	6
Touble failing asleep	0	1	2	2	4	5	6
More emotional	0	1	2	2	- 4	5	6
intability	0	1	2	2	- 4	5	6
Sadness	0	1	2	2	- 4	5	6
Nervous or Anxious	0	1	2	- 2	4	5	6
Total number of eye	ptons (Nain	in posit	ie 33)				
Symptom severity so	ore (Maximum)	oolbie 1	440				
Do the symptoms get	worse with phy	nical ac	tivity?			Y	
Do the symptoms get	wone with me	ntal act	vity?			Y	
self rated		self ra	ed and	dinick	n mon	bored	
clinidan interview		self ra	ed with	h paren	t input		
Overall rating: If you the athlete acting com Rease cicle are required.	know the athi pared to his/h	ete wei er usual	prior t self?	o the in	ijury, h	ow dff	erent
no different	very different		unsun	•		NIA	

return to con

over time, it i

COGNITIVE & PHYSICAL EVALUATION

4	Cognit Standardb	ive ed/	assi	ess ner	imen t of Ca	nt incus	ion	(SAC)*		
	Orientatio		coint for	each	COLOR 1					
	What mont	h is it	0						0	1
	What is the	date	todayi	2					0	1
	What is the	day	of the v	***	0				0	1
	What year i	8 H2	-						0	1
	what time i		gni no			Server (
	Orientatio	n sco	ore.							ors
	innediate	ne	mory							
	Lie	1	1011		1912	12	1 1	Abenative wi	and ling	
	ebow	0	1	0	1	0	1	candle	baby	finger
	apple		-1	0	1		4	baber	monkey	penny
	carpet	-	-	0			÷	sugar randaich	perturne	lamon
	bubble	÷	- 1	0	- 1	÷.	÷	Wagon	iron	inect
	Total									
	inmediate	me	mory s	con	total					of 15
	-								_	
	Concentral	tion:	Digiti		cine ar c	1				
	102	-	100		AD403	0.00		* 1 4		
	3444			-	3-3-3	9		1.7.6.5	4-1-0	
	63.634	-		÷	1.5.3			3.6.5.3.7	614	4.3
	7-1-8-4-6-2		ŏ	÷	5-3-9	1-4-8		8-3-1-9-6-4	7-2-4	4-5-6
	Total of 4									
					_	_				
	Concentral	tion	Mont	th in	Rever	se On	Ser (t pt. for entire o	ACURE OF COLOR	κα) 4
	Line-Non-Us		phone	-	un-90	il-ver	-0.8	-140-180		
	Concentral	tion	score							of S
				_						
5	No. 6						_			_
5	Neck E	xar	mina	ntio	on:			and lower line		land
5	Neck E Range of m	xar	mina	atio Tena	on: Iemes		iper	and lower lin	to sensation	iåstrength
5	Neck E Range of m Rindings:	xar	mina	atio Tena	on: terres	u	per	and lower lin	to sensation	nästrength
5	Nock E Range of m Rindings	xar	mina	Ten	on:	uş	.per	and lower lin	to sensation	nåstrength
5	Neck E Sange of m Rindings:	xar ctior	nina	atio Terra	on: inter	, uş	.per	and lower lin	tb sensation	nåstrength
5	Neck E Range of m Rindings: Balance bose of bal	xar otion	mina xam	in a	on: ation	: u;	per	and lower lin	tà sensatio	i åstrength
6	Nock E Range of m Pindings: Balance Do ser or bal Foctwear (b	e e	mina xam e talou	in a bot,	ation	: U) 1	oper etc.)	and lower lin	to sensation	nåstrength
6	Nock E Range of m Pindings: Balance Do ser or bal Foctwear (b Modified B	ction ction ction ction ction ction ction ction	mina xam states baret	in a bot, or S	ation ex. bace, coring	tape,	sper etc.)	and lower lin	to sensation	nåstrength
6	Neck E Range of m Rindings: Dose or bot Footwar (0 Modified B Which foot	xar ation e e: hofth hors, ialan wat t	mina xam e falos , barefi testedi	in a bot, or S	ation ensures braces, coring etich ico	tape,	etc.) m (B	and lower lin (SS) texting (Net for)	th semation	hästrength t Right
6	Neck E Range of m Rindings: Dose or bot Footwar (0 Modified B Which foot Thering surf Condition	xar ation e e: hofth hoes lalan wast hoes()	xam e falos barel ce Em tested hard fic	in a bot, or S (Le v oor,	ation denses braces, braces, deld lot field, et	tape, Syste te auto tc.)	sper etc.) m (8	and lower In (SS) tecting (LSS) tecting	to sensation	t Rgt
6	Neck E Jange of m Findings: Balance Do see of bel Footwaar (s Modified B Which foot Teeting surf Condition Double leg	xar ation both hat has lalan wat has () tance ()	xam states baret tested hard fo	in a lin a lin a lin a lin a a	ation ation to: braces, oring this iss field, et	tapa, Syste tc.)	etc.) m (B	and lower In (55) tecting event for)	nb semation	t Byte
6	Neck E Range of m Rindings: Do see of both Footware (st Modified B Which foot Teeting surf Condition Double leg is Single leg of	xar ation e e: hate hate hate hate	xam e fallow barefi terted hard fic x	in a bot, or S (A v oor)	ation etc. braces, coring etich icc field, et	: U) tape, Syste ic.)	etic.) m (8	and lower In (SS) tecting event foot)	nb semation	t Byte
6	Neck E Range of m Redings: Do see of both Footware (M Modified B Which foot Teeting suff Condition Double legs Single legs Tandem sta	C C C C C C C C C C C C C C C C C C C	mina xam ; barel ; barel icceEm tested ; hand fo ; ; ; ; ; ; ; ; ; ;	in a tic in a top 5 bot, or 5 (1.4 v bot, in a top 5 bot, in a top 5 top 5 t	ation encoding braces,	: U; tape, Syste Syste te con Stati;	etic) m (B	and lower In (55) tecting want for)	t unatio	h & strength h & strength E & Right E mars E mars
6	Neck E Range of m Redings: Douer of both Footware (b Modified B Which foot Teeting suff Condition Double lag Single lag that Tandem ta And/Or	ction ction bolt bolt bolt bolt bolt bolt bolt bolt	mina xam stated bard tetted hard fo x (ton-do	in a tio in a high bot, or 5 bot, or 5 bot, or 5 bot, in a bot, or 5 bot, or 5 bot, or 5 bot, or 5 bot, or 5 bot, bot, bot, bot, bot, bot, bot, bot,	ation etc. bracer, bracer, field ist field, et foot at	: U) , tape, ; Syste te soo tc.) ; tadi;	etc.) m (0	and lower lin (SS) tecting event for)	t unatio	t Byte Byte Errors Errors
6	Neck E Range of m Redings: Do sere to ball Societor ball Modified B Which foot Testing auf Condition Double leg to Single leg at Tandem dat And/Or	xar otion e e: hoth hoes talan vac t hoes hoes tance tance tance	mina xam e faloa barel hard fo kar-dar	in a bot, or S (A v oor,)	on: ation etc. braces, coring etcl ist feid, et feid, et	: U; tapa, Syste te non :: tadi;	etic) m (B	and lower in (SS) tecting event for)	to sensation	t Ryte Emars Emars Emars
6	Neck E Range of m Rindings: Do see of both Foctoware (or Modified B Which foot Testing suff Conditions Double leg of Tandem sa And/or Tandem ga	Car otion e e: hofth hoes tance tance tance tance (tance tance tance tance tance	mina xam states tote tote tote tote tote tote tote t	in a in a in a bot, or 5 bot, or 5 bot, or 5 bot, or 5 bot, or 5 bot, or 5 bot, or 5 bot, or 5 bot,	ation etc. braces, coring etich int field, et field, et	: Up tape, Syste te con tc.) back):	etc.) m (0 -dom	and lower in (55) tecting element foot)	to sensation	t Right Emors Emors
6	Neck E Range of m Rindings: Do see of both Foctware (st Modified B Which foot Teeting surf Condition Double lag) Single lag of Tandem sta And/ Or Tandem ga	C C C C C C C C C C C C C C C C C C C	mina xam e falos , baref ice Em tected i hard fo se: (con-do tected)	in a tic bot, or S (Le v bot, ison	ation etc. braces, coring which is t field, et art foot;	: Up tape, tape, te son te son te son te son te son te son	etc.) m (B -don	and lower in (1955) terting (next for)	to sensation	t Ryte Errors Errors
5	Neck E Range of m Rindings: Deare a bat Footwar (t Modified B Which foot Death leg (Single leg of Death leg (Single leg of Time (sec of Coording) Time (sec of Coording)	ction ction holth hoes tance tance tance tance tance tance tance tance tance tance tance	mina xam s folios sceEm tected hard fo sceEm tected hard fo sceEm tected to ceEm tected to ceEm tected to ceEm tected	extice extine	ation etc. braces, braces, feid, et feid, et four et amin	: U) , tape, te non te non te non te non te non te non te non te non	etc.) m (8 -dom	and lower lin	2 unatio	t Right Errors Errors
5	Neck E Range of m Redinge: Balance Dosens bei Foctware (M Modified B Which foot Technik (M Single Rig (Tandem ga Time (tech of Time (tech of Time (tech of Time (tech of the Time (tech of the Time (tech of the Time (tech of the Time (tech of the Time (tech of the Time (tech of the Time (tech of the Time (tech of the Time (tech of tech of the Time (tech of tech of	xar ation batto tance tance tance tance tance tance tance tance tance tance tance tance tance tance tance	mina xam e talou , baret toe Em terned hand fo se (por-do too too too too too too too	extin extin	ation etc. braces, braces, feid, et feid, et four et amin	: U; tape, Syste te non tc) _ : taci; nati	sper etc.) m (B da	and lower lin (SS) texting (Name for)	2 unutio	h & down gin t Right Errors Errors
5	Nock E Range of m Rodings: Doare to the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of the Content of	xar otion botto tance () ttance () t	mina xam e talou , baret tered hand fo sector in tered to continue tered to continue	ex tion	ation etc. braces, coring etch ict feid, et foot at amin	: U; tape, Syste te non tc); bad); nati	etc.) m (8 da	and lower In	d sensio	t Ryte Bross Bross Bross Bross Bross
5	Neck E Sange of m Redings: Do see or bail Societ on the f Noothed E Which foot Texting user Single lag th Tendem ga Time (best of Coordial Upper line) Which arm Coordial	C C C C C C C C C C C C C C C C C C C	mina xam xam xae fin sce Em teced hand fin sc (con-do teced an-do teced teced teced teced teced teced teced	ex	ation etc. braces, coring elicitics feid, et foot at amin	: U) tapa, System te nan te nan te nan te nan te nan te nan	etc.) m (8 -dom	and lower in	th sensition	tå strength t Right Emark Emark Emark Emark Emark
5	Nock E Range of m Rudings: Doares to the Foctware (tr Modifield Which foot Double leg) Single leg of Tandem ta Tandem ta Time (leg of Upper limit Which foot	xar ation ation hoes talan was t ace () tance ance () (4 stal b cost was t ison s	xam xam stalow tote Em tote Em tote Em tote Em set (non-do non-don tote for tote for	ex	ation etc. braces, braces, braces, braces, etc. field, et arr foot) fiber, et	: U) tapa, syste te non te non te non te non te non	etc) n (8 -dot	and lower In (55) testing isant for)	the semantice (t Right Right Right of 1
5	Neck E Range of m Rudings: Balance Doares to the Foctwar (0 Modifiel i Which foot Dadie ing Single ing of Single ing of Single ing of Time (act of Upper line) Which arm Coordition Upper line) Which arm Coordition	carrier e e: hofts hoes hoes hoes hoes hoes hoes hoes hoe	xam e foliow , barel hard fo hard for more for erafinat cone tion erafinat cone	ex ex ex	ation etc. braces, bra	: Uj 1 tapa, Syste te non te: tadi: tadi: tadi: tadi:	:per etc:) = (8 -dos	and lower In (SS) tecting (super for)	e senation	t Right Draws Draws Draws Draws Draws Draws

BCATE SPORT CONCUSSION ASSESSMENT TOOL 8 | PAGE 2

C 2013 Concession in Sport Group

Appendix B

Sport Concussion Assessment Tool 5 (SCAT5)

Downloaded from http://bjam.bmj.com/ on February 9, 2018 - Published by group bmj.com

SCAT5.	SPOR DEVEL	RT CONCL OPED BY TH SE BY MEDIC	JSSION A E CONCUSS CAL PROFES	SSES ION IN S	SMENT TOOL - 5TH EDITION SPORT GROUP S ONLY
			surrouted by		
	2	FIFA'		Ľ	FEI
Patient details					
Name:					
D08:					
Address:					
ID number:					
Examiner:					

WHAT IS THE SCAT5?

The SCAT5 is a standardized tool for evaluating concussions designed for use by physicians and licensed healthcare professionais'. The SCAT5 cannot be performed correctly in less than 10 minutes.

If you are not a physician or licensed healthcare professional, please use the Concussion Recognition Tool 5 (CRT5). The SCAT5 is to be used for evaluating athletes aged 13 years and older. For children aged 12 years or younger, please use the Child SCAT5.

Preseason SCAT5 baseline testing can be useful for interpreting post-injury test scores, but is not required for that purpose. Detailed instructions for use of the SCAT5 are provided on page 7. Please read through these instructions carefully before testing the athlets. Brief verbal instructions for each test are given in italics. The only equipment required for the tester is a watch or timer.

This tool may be freely copied in its current form for distribution to individuals, teams, groups and organizations. It should not be altered in any way, re-branded or solid for commercial gain. Any revision, translation or reproduction in a digital form requires specific approval by the Concussion in Sport Group.

Recognise and Remove

A head impact by either a direct blow or indirect transmission of force can be associated with a serious and potentially fatal brain injury. If there are significant concerns, including any of the red flags listed in Box 1, then activation of emergency procedures and urgent transport to the nearest hospital should be arranged.

- Key points
- Any athlete with suspected concussion should be REMOVED FROM PLAY, medically assessed and monitored for deterioration. No athlete diagnosed with concussion should be returned to play on the day of injury.
- If an athlete is suspected of having a concussion and medical personnel are not immediately available, the athlete should be referred to a medical facility for urgent assessment.
- Athletes with suspected concussion should not drink alcohol, use recreational drugs and should not drive a motor vehicle until cleared to do so by a medical professional.
- Concussion signs and symptoms evolve over time and it is important to consider repeat evaluation in the assessment of concussion.
- The diagnosis of a concussion is a clinical judgment, made by a medical professional. The SCATS should NOT be used by itself to make, or exclude, the diagnosis of concussion. An athlete may have a concussion even if their SCATS is "norma".

Remember:

- The basic principles of first aid (danger, response, airway, breathing, circulation) should be followed.
- Do not attempt to move the athlete (other than that required for airway management) unless trained to do so.
- Assessment for a spinal cord injury is a critical part of the initial on-field assessment.
- Do not remove a heimet or any other equipment unless trained to do so safely.

© Concussion in Sport Group 2017

Echemendia RJ er al. Br. / Spore. Med 2017;51:851–858. doi:10.1136b3ports-2017-0975065CAT5

851

Downloaded from http://bjsm.bmj.com/ on February 9, 2018 - Published by group.bmj.com

1							
			Name:				
IMMEDIATE OR ON-FIELD ASSESSM	IENT	r i	008:				
The following elements should be assessed for all	athlet	tes who	Address:				
are suspected of having a concussion prior to pro-	eedin	g to the	ID number: Examiner:				
the first first aid / emergency care priorities are comp	e on-hi leted.	eid after					
If any of the "Dad Eleve" or chearvable since are note	d after	a direct	Date:				
or indirect blow to the head, the athlete should be im asfely removed from participation and evaluated by licensed healthcare professional.	media a phys	tely and sician or					
Consideration of transportation to a medical facility the discretion of the physician or licensed healthcare	y shou profes	id be st sional.	STEP 4: EXAMINATION GLASGOW COMA SCALE (GCS) ^a			
The GCS is important as a standard measure for all pa	tienta	and can	Time of exceptions				
state. The Maddocks questions and cervical spine ex	am are	critical	Date of assessment				
steps of the immediate assessment; however, these be done serially.	do not	need to					
			Beeck age responses (8)	_			
STEP 1: RED FLAGS			No eye opening	1	1	1	
			tye opening in response to pain	2	2	2	
RED FLAGS:			tys specificate speech	3	3	3	
		_	Eyes opening spontaneously	4	4	4	
Neck pain or Setzure or con	WUINK	261	Real verbal response (V)				
Loss of conso	louse		No verbal response	1	1	1	
Deteriorating			Incomprehenable aconde	2	2	2	
 Weakness or lingling/ conscious sta burning in arms or legs 			Inappropriate words	3	з	3	
 Versiting 			Carifued	4	4		
headache increasingly n	es Ces	<u>.</u>	Orlanted	8	5	3	
egitanic of co			Real motor response (M)				
		_	No multior response	1	1	1	
STEP 2: OBSERVABLE SIGNS			Extension to path	2	2	2	
Witnessed D Observed on Video D			Acres nel fector to peri				
Lying motionizes on the playing surface	Y	н	Freedow in the second second second				
Relation / cell difficulties / motor incocedination: stambility, slow /			Character of part				
laboured movements		R	Classes Conservers (8 + V + M)				
Disortentation or confusion, or an inability to respond appropriately to questions		н					
Red or record lock		н					
			CERVICAL SPINE ASSESS	MENT			
Facial tripsy after head trauma	۲	н	Does the athlete report that their nexk is pain free at re	e17	٧	н	
STEP 3: MEMORY ASSESSMENT MADDOCKS QUESTIONS ²			If there is NO much pain at real, does the athlate have a range of ACTIVE pain free movement?	en la	۲	н	
7 are poly to and proce fee questions, please laten samfully and give year best effort. Piret, bill me solar happened?"			Is the Trob strength and sensation normal?		٧	н	
Mark T for correct assesse / N for incorrect							
What return are not all today?	۲	н					
Which half is it new?	×.	н	In a patient who is not is	icid or	fully		
Who assered last in this match?		н	conscious, a cervical spine	other	should		
What learn did you play last week / game?	¥	н	be assumed unce prover	outer			
DN your team win the last game?	¥	н					
Note: Appropriate aport-aportific guestione may be substituted.							
Echemendia RJ et el. Br. / Sports	Med 20	ussion in 017; 51 :8	spon croup 2017 \$1-858. doi:10.1136/bjsports-2017-0975065CATS				

Downloaded from http://bjsm.brnj.com/ on February 9, 2018 - Published by group.brnj.com

OFFICE OR OFF-FIELD ASSESSMENT			6								_
				Name:							
Please note that the neurocognitive assessment and distraction-free environment with the athlete in a rea	ting state	ione in a	· .	008:							
				Address:							
STEP 1: ATHLETE BACKGROUND				ID number:							
Sport/team/school				Examiner:							
Date / time of injury:				Date:							
Years of education completed:											
Age:				2							
Gender: M / F / Other				STEP 2: SYMPTO	м	EV/	LU		ON		
Dominant hand: left / neither / right				The other should be should be							
How many diagnosed concussions has the				paragraph out load than complete	the sp	et en	and a	for the			
athlete had in the past?:			.	the postinglary assessment the add		uhi san	their e			pole	in them.
When was the most recent concussion?:			.	Please Check: Basel	ine i	P	oet-b	ų ury			
How long was the recovery (time to being cleared to pl from the most recent concussion?:	ay)	(dava)		Please hand	the f	lorm t	o the	athle	te .		
		(seated)	' I		none		м	mod	e ele i		
Has the athlete over been:				Headache	0	1	2	3	4	5	
				"Pressure in head"	0	1	2	3	4	5	
Hospitalized for a need lighty?	194	~		Nexi: Pals	0	1	2	3	4	5	
Discovered (treated for handache discolar or minutesa)	Nes			Navana or vomiling	0	1	2	3	4	5	
	· · ·	~		Dischess	0	1	2	3	4	3	
Disongeed with a learning disability / dyales/a?	Yes	No		Blurred vision	0	1	2	3	4	3	
				Ralance problems	0	1	2	3	4		
Diagnosed with ADD / ADHD?	Yes	No		Security to type	•	-	-				
				Senaltivity to noise	0	-	2				
Diagnosed with depression, analety or other percellation discover?	Yes	No		Feeling Rocks and a		÷	2		-		
				"Don't feel right"	0	1	2	3	4	3	
Current medications? If yes, please list:				Difficulty concentrating	0	1	2	з	4	3	
				Difficulty remembering	0	1	2	3	4	8	
				Fallple or low energy	0	1	2	3	4	8	
				Confusion	0	1	2	3	4	3	
				Drovalness	0	1	2	3	4	3	
			·	More emotional	0	1	2	3	4	3	
				billability	0	1	2	3	4		
				Technese Indone	0		-				
			·	Trouble failing ealersp 17 exclosion	0	1	2	3	÷	5	
			·	Total number of symptoms:							622
				Symptom severity score							182
				Do your symptoms get worse with	physic	el eciti	aya.		1		
				Do your symptoms get worse with	mente	and the	97		1		
				If 100% is feeling perfectly normal percent of normal do you feel?	what						
				B'nut 100%, why?							
				Please hard	form	baci	to e	amir	er		

© Concussion in Sport Group 2017 Echemendia RJ +r +L &r / Sports Med 2017;51:851–858. doi:10.1136/bjports-2017-0975065CAT5

853

Downloaded from http://bjsm.bmj.com/ on February 9, 2018 - Published by group.bmj.com

3

STEP 3: COGNITIVE SCREENING Standardised Assessment of Concussion (SAC)⁴

ORIENTATION

What month is 17		1
What is the date to day?		1
What is the day of the uses!?	•	1
What year is 87		1
What time is it right new? (within 1 hour)		

IMMEDIATE MEMORY

The Immediate Memory component can be completed using the traditional 5-word per trial list or optionally using 10-words per trial to minimias any osling effect. All trials must be administered imspective of the number correct on the first trial. Administer at the rate of one word per second.

Please do see 517498 the 3 or 10 word het groups and sinds the specific word hat chosen for this basi.

I am going to test your memory. I will need you a third of words and when I am door, respect basis as many words as you can remember, it may order. For Titals 2.2.2.1 am going to repeat the same for again. Repeat basis as many words as you can remember in any other, even if you wait the word balance.

1.041	Alternate 3 wood Tate						one jos	20
-						101	Titel 2	The
A	Finger	Penny	Sanial	Lemon	Inerit			
8	Candle	Paper	Sugar	Sandwish	Wagon			
c	Baily	Mariley	Perfune	Surael	ben			
0	Show	Appie	Carpel	Sable	Bubble			
	Jacket	Aron	Papper	Collan	Movie			
۴	Dollar	Honey	Minur	Sable	Anahor			
			in the	mediate Marr	ary Boarn			ef 1
			Time that is	and to be a set of	empirised			
						-		

						3	ore (of	×)
			10.000			1041	1612	Titel 3
	Plager	Penny	Nation	Lemon	line(1			
۰.	Candle	Paper	Sugar	Sandwish	Wagon			
	Bally	Murikey	Perfune	Surant	ben			
	Ellow	Apple	Carpel	Saide	Bubble			
	Jacket	Aron	Pagest	Collian	Monte			
	Dollar	Honey	Minur	Saide	Anahor			
			in the	mediate Man	ory Boorn			u130
			Time that is	and table in some	empired			

Name:	
008:	
Address:	
ID number:	
Examiner:	
Date:	

CONCENTRATION

DIGITS BACKWARDS

Please circle the Digit list chosen (A, D, C, D, E, F). Administer at the rate of one digit per second reading DOWN the selected column.

Lem going to need a string of numbers and when Lem done, you repeat them back to me to revenue order of how tread them to you. For example, if Leay 3-14, you would say 9-17.

Concentra	dan Number La	da (stole one)			
UMA	LM B	LHIC			
493	526	142		н	
629	415	6-5-8		н	1
3-81-6	1795	6-80-1		н	
3274	49.98	2481		н	1
62.671	48-927	49103	τ.	н	٥
15286	618-63	68251		н	1
718462	801964	376-514		н	0
\$29148	724856	926514		н	1
LMD	CHER	List P			
282	1418 2402	100 P	×	н	0
282 926	582 518	221 429	r r	н н	0 1
282 926 6182	240 240 2743	201 201 409 1643	T T	*	0 1 0
240 742 924 4183 9723	2468 2492 2793 2793	201 201 409 1448 3404	т т т	* * *	0 1 0 1
282 926 4193 9733 17936	Lark 382 518 2793 2199 61899	Las P 271 479 1683 3924 24758	7 7 7 7	* * * *	0 1 0 1
240 742 924 4143 9723 17424 41762	Left 382 518 2793 2199 41899 94175	LasP 221 429 1683 2924 24758 89964	* * * * *	* * * *	0 1 0 1
1040 242 4192 9723 17934 41762 2944917	Last 3402 304 2743 2144 40845 94075 9407582	LarP 221 429 1683 3924 24736 83964 589249	* * * * *	* * * * *	0 1 0 1 0 1 0
Lato 782 924 4183 9723 17924 41752 244817 841925	Last 382 318 2743 2144 41844 94173 647342 427438	LatP 201 409 1683 2924 24058 83964 586249 210826	* * * * * *	* * * * *	0 1 0 1 0 1 0 1

MONTHS IN REVERSE ORDER

Non-tell one the most he of the year is revenue order. Shot with the last month an So you'll my December, Nonember, Goudread,	fpo becilverd
Des - Nov - Ool - Sept - Aug - Jul - Jun - May - Apr - Mar - Feb - Jun	0.1
Months Source	of1
Concentration Total Score (Digits + Months)	of 5

854

Concussion in Sport Group 2017 Echemendia RJ, et al. dr. / Sports. Med. 2017;51:851–858. doi:10.1136/bjports-2017-0975065CAT5 Downloaded from http://bjsm.bmj.com/ on February 9, 2018 - Published by group.bmj.com

4 Name: STEP 4: NEUROLOGICAL SCREEN 008:____ Address: See the instruction sheet (page 7) for details of test administration and scoring of the tests. ID number: Can the patient read alood (e.g. symptom sheek-het) and follow instructions without difficulty? N Examiner: Date: Does the patient have a full range of patients free PASSIVE cervical aplice movement? Y 8 Without moving their head or nexis, can the patient lask Y N Adde to adde and up and down without double relator? Canthe patient perform the finger nose coordination test normalis? Y 8 5 Canthe patient perform landern galt normally? Y N STEP 5: DELAYED RECALL: The delayed recall should be performed after 5 minutes have elapsed since the end of the immediate Recall section. Score 1 pt. for each correct response. BALANCE EXAMINATION Modified Balance Error Scoring System (mBESS) testing* Which fact was leaded (Le. which is the new dominant fact) in Left Dillight Do you remember that hat of words I read a few times as first Tell me as many words from the let as you can remember in any order. Testing surface (hard floor, field, etc.)_____ Footneer (shoes, barefloot, braces, tape, etc.)____ The Darted _ Place record each word correctly recalled. Total score equals runder of words recalled. Condition Broom Double leg statue urte. Engle ing status (non-dominant fool) ulte Tendem stance (non-dominant foot at the be w110 Total number of words recalled accurately. of 5 or of 10 Total Errors of 30 6 STEP 6: DECISION Date and time of lighty -*6 If the adhiete is known to you prior to their topicy, are they different from their usual self? The I No I Desare I No Applicable (f different, desarbe why in the obtical noise section) Symptom number (of 22) Concueston Diagnosed? © Yes © No © Unears © Not Applicable Symptom severity score (of 132) If re-testing, has the athlete improved? The C No C Unears C Not Applicable Orientation (of 2) af 15 af 30 of 15 of 20 of 15 of 30 Immediate memory I am a physician or licensed healthcare professional and I have pe administered or supervised the administration of this SCATS. cally Concentration (of 2) Signature:_____ Neuro exam Name Normal Normal Normal Abnormal Abnormal Abnormal Title:___ Balance errors (of 30) Registration number (if applicable): ofS of10 afs afs afte afte Delayed Recall Date: SCORING ON THE SCAT5 SHOULD NOT BE USED AS A STAND-ALONE METHOD TO DIAGNOSE CONCUSSION, MEASURE RECOVERY OR MAKE DECISIONS ABOUT AN ATHLETE'S READINESS TO RETURN TO COMPETITION AFTER CONCUSSION. © Concussion in Sport Group 2017 Echemendia Rij er el. dr. / Sports. Med. 2017;51:851–858. doi:10.1136/bjports-2017-09/5065CAT5

85

Appendix C

Data Use Approval

GEORGE FOX

Office of Intercollegiate Athletics 414 N. Meridian St. #6149, Newberg, OR 97132 503.554.2910 | athletics.georgefes.edu

February 15, 2018

Dear Committee Chair,

I give permission to Jill Sikkema to conduct research using data in the George Fox University Sports Medicine medical database for her dissertation. The purpose of the study is to examine the psychometric properties of the Sport Concussion Assessment Tool (SCAT) versions 3 and 5 along with examining their ceiling effects.

Mrs. Sikkema will have access to the secure medical database report platform for the 2016 and 2017 baseline SCAT testing. She has permission to analyze, evaluate, and report on this data to answer the following research questions:

 Are there differences in the factor structure and internal reliability between the SCAT3 and SCAT5 subsections of immediate memory and delayed recall assessments in an analysis of baseline data?

Are there differences in the proportions of floor/ceiling effects for the SCAT3 and SCAT5 subsections of immediate memory and delayed recall assessments in an analysis of baseline data?

NITE OF 2011 POT CERTIFIE ELEVELED FOR S

Please contact me if you have any questions.

Sincerely

Dale Isaak Director of Sports Medicine George Fox University <u>disaak@georgefox.edu</u> 503-554-2916.

Appendix D

Institutional Review Board Approval

GEORGE FOX UNIVERSITY HSRC INITIAL REVIEW QUESTIONNAIRE

Page 6

Title: EXAMINATION OF THE PSYCHOMETRIC PROPERTIES OF THE SCAT5 BASELINE TESTING DATA ON NCAA DIVISION III COLLEGE ATHLETES

Principal Researcher(s): Jill Sikkema

Date application completed: 2/27/18

(The researcher needs to complete the above information on this page)

COMMITTEE FINDING:

For Committee Use Only

(1) The proposed research makes adequate provision for safeguarding the health and dignity of the subjects and is therefore approved.

(2) Due to the assessment of risk being questionable or being subject to change, the research must be periodically reviewed by the HSRC on a basis throughout the course of the research or until otherwise notified. This requires resubmission of this form, with updated information, for each periodic review.

(3) The proposed research evidences some unnecessary risk to participants and therefore must be revised to remedy the following specific area(s) on non-compliance:

(4) The proposed research contains serious and potentially damaging risks to subjects and is therefore not approved.

Winster Legater.

03/06/2018

Chair or designated member

Date

Appendix E

Items	O_Date	O_Day	O_ Month	O_Time	O_Year	C1	C2	C3
O_Date		-0.036	-0.016	-0.056	-0.016	-0.016	0.007	0.072
O_Day			-0.011	0.031	-0.011	-0.011	0.063	0.074
O_Month				-0.017	-0.005	-0.005	-0.021	0.090
O_Time					-0.017	-0.017	0.000	0.009
O_Year						-0.005	0.102	-0.054
C1							-0.021	0.090
C2								0.346
C3								
C4								
C_Month								
IM1_1								
IM1_2								
IM1_3								
IM1_4								
IM1_5								
IM2_1								
IM 2_2								
IM 2_3								
IM 2_4								
IM 2_5								
IM 3_1								
IM 3_2								
IM3_3								
IM3_4								
IM3_5								
DR1								
DR2								
DR3								
DR4								
DR5								

Correlation Matrix for SCAT3

Item	C4	C_Month	IM1_1	IM1_2	IM1_3	IM1_4	IM1_5
O_Date	0.023	0.031	-0.020	-0.040	0.011	-0.024	0.027
O_Day	-0.023	0.046	-0.013	0.067	0.050	0.195	-0.066
O_Month	0.049	0.146	-0.006	-0.012	-0.014	-0.020	0.068
O_Time	0.059	0.047	0.104	-0.042	-0.048	0.011	-0.042
O_Year	-0.099	-0.033	-0.006	-0.012	-0.014	-0.020	0.068
C1	0.049	0.056	-0.006	-0.012	-0.014	-0.020	-0.029
C2	0.181	0.117	-0.026	-0.002	-0.017	0.075	0.015
C3	0.469	0.145	-0.007	0.045	0.027	0.027	0.068
C4		0.047	0.060	0.121	0.035	0.064	0.123
C_Month			0.106	-0.008	-0.031	0.006	0.009
IM1_1				-0.015	-0.017	0.084	-0.036
IM1_2					0.264	0.115	0.130
IM1_3						0.086	0.127
IM1_4							0.061
IM1_5							
IM2_1							
IM 2_2							
IM 2_3							
IM 2_4							
IM 2_5							
IM 3_1							
IM 3_2							
IM3_3							
IM3_4							
IM3_5							
DR1							
DR2							
DR3							
DR4							
DR5							

Item	IM2_1	IM2_2	IM2_3	IM2_4	IM2_5	IM3_1	IM3_2	IM3_3	IM3_4
O_Date	-0.011	-0.038	-0.025	0.117	0.119	-0.016	0.090	-0.016	0.075
O_Day	-0.008	-0.026	-0.017	0.085	0.047	-0.011	-0.015	-0.011	-0.017
O_Month	-0.003	-0.011	-0.008	-0.010	-0.014	-0.005	-0.007	-0.005	-0.008
O_Time	-0.012	0.026	-0.027	-0.036	0.056	-0.017	-0.024	-0.017	-0.027
O_Year	-0.003	-0.011	-0.008	-0.010	-0.014	-0.005	-0.007	-0.005	-0.008
C1	-0.003	0.205	-0.008	0.229	0.161	-0.005	0.349	-0.005	0.311
C2	-0.015	0.056	-0.034	0.013	-0.020	-0.021	0.057	-0.021	0.123
C3	0.064	0.059	0.006	0.090	-0.008	0.018	0.026	0.018	0.052
C4	0.035	0.021	0.078	0.035	0.094	0.049	0.069	0.049	0.031
C_Month	-0.023	-0.001	0.004	0.142	-0.005	0.056	-0.047	-0.033	0.004
IM1_1	-0.004	-0.014	-0.009	-0.013	0.126	-0.006	-0.008	-0.006	-0.009
IM1_2	-0.008	0.061	0.245	0.172	0.037	-0.012	-0.017	-0.012	-0.019
IM1_3	-0.010	0.123	0.207	0.142	0.085	-0.014	-0.020	-0.014	-0.022
IM1_4	-0.014	0.010	0.305	0.209	0.080	-0.020	-0.028	-0.020	-0.031
IM1_5	-0.021	-0.027	0.016	0.031	0.323	0.068	-0.041	-0.029	0.078
IM2_1		-0.008	-0.005	-0.007	-0.010	-0.003	-0.005	-0.003	-0.005
IM 2_2			0.119	0.181	0.042	-0.011	0.291	0.205	0.119
IM 2_3				0.135	-0.023	-0.008	0.215	-0.008	-0.012
IM 2_4					0.220	-0.010	0.155	-0.010	0.135
IM 2_5						-0.014	0.104	-0.014	0.200
IM 3_1							-0.007	-0.005	-0.008
IM 3_2								-0.007	0.215
IM3_3									-0.008
IM3_4									
IM3_5									
DR1									
DR2									
DR3									
DR4									
DR5									

Item	IM3_5	DR1	DR2	DR3	DR4	DR5
O_Date	-0.025	0.016	0.019	0.067	0.012	0.050
O_Day	-0.017	0.016	0.037	-0.092	0.012	-0.010
O_Month	0.311	0.039	0.137	0.039	0.061	0.081
O_Time	-0.027	0.124	0.034	-0.045	0.114	0.040
O_Year	-0.008	-0.041	-0.035	-0.041	-0.032	-0.026
C1	0.311	0.039	-0.035	-0.041	0.061	-0.026
C2	-0.034	-0.043	0.014	-0.023	0.065	-0.035
C3	0.052	0.051	-0.033	-0.028	0.007	-0.010
C4	0.078	0.084	0.028	0.049	0.035	0.056
C_Month	0.061	0.020	0.081	0.062	0.063	-0.005
IM1_1	-0.009	0.015	-0.043	-0.050	-0.039	-0.032
IM1_2	-0.019	-0.002	-0.087	-0.068	0.036	-0.064
IM1_3	-0.022	0.084	-0.008	-0.088	0.109	-0.035
IM1_4	0.053	0.023	0.015	-0.061	-0.056	0.008
IM1_5	0.078	0.019	-0.078	0.081	0.061	-0.051
IM2_1	-0.005	-0.029	-0.025	-0.029	-0.022	-0.018
IM 2_2	0.257	0.076	-0.046	-0.028	0.124	0.077
IM 2_3	-0.012	0.087	-0.001	-0.014	0.009	0.095
IM 2_4	0.135	-0.011	-0.034	0.027	0.064	-0.004
IM 2_5	0.312	0.019	-0.044	0.046	0.100	-0.002
IM 3_1	-0.008	0.039	-0.035	-0.041	-0.032	-0.026
IM 3_2	0.215	0.055	0.011	-0.001	0.086	0.191
IM3_3	-0.008	-0.041	-0.035	0.119	-0.032	-0.026
IM3_4	0.190	0.037	-0.056	-0.064	0.184	-0.041
IM3_5		-0.014	-0.001	-0.064	0.067	0.163
DR1			0.251	0.177	0.248	0.208
DR2				0.128	0.198	0.161
DR3					0.101	0.157
DR4						0.304
DR5						

Appendix F

Correlation Matrix for SCAT5

Items	0_D	ay O_N	Month	O_Time	O_Year	C1		C2	C3
O_Date	0.22		295	0.144	0.248	0.15	55 0	.048	0.117
O_Day		0.	608	0.361	0.524	0.34	4 0	.089	0.057
O_Month				0.456	0.865	0.57	' 4 0	.188	0.066
O_Time					0.391	0.25	64 0	.194	0.105
O_Year						0.49	05 0	.155	0.040
C1							0	.155	0.094
C2									0.277
C3									
Items	C4	C_Mo	IM	IM	IM	IM	IM	IM	IM
		nth	1_1	1_2	1_3	1_4	1_5	1_6	1_7
O_Date	0.082	0.059	0.080	0.012	0.040	0.018	-0.106	0.031	0.008
O_Day	0.049	0.076	0.114	0.030	-0.017	0.099	0.049	0.071	0.114
O_Month	0.074	0.113	0.222	0.115	0.041	0.082	0.074	0.043	0.070
O_Time	0.079	0.092	0.123	0.021	-0.002	0.098	0.024	0.061	0.040
O_Year	0.034	0.087	0.292	0.088	0.014	0.095	0.086	0.050	0.080
C1	0.086	0.087	0.186	0.088	0.065	0.044	0.086	0.050	0.029
C2	0.157	0.108	0.041	0.061	0.078	0.037	0.045	0.054	0.008
C3	0.278	0.117	0.146	0.093	0.087	0.071	0.013	0.023	-0.007
C4		0.146	0.151	0.169	0.150	0.069	-0.051	-0.008	0.043
C_Month			0.050	0.020	-0.012	0.044	0.038	0.010	0.078
IM1_1				0.333	0.141	0.090	-0.042	-0.007	0.006
IM1_2					0.278	-0.002	0.033	-0.043	-0.174
IM1_3						0.103	0.004	0.061	-0.144
IM1_4							0.090	0.108	0.066
IM1_5								0.018	-0.104
IM1_6									0.036
IM1_7									
IM1_8									
IM1_9									
IM1_10									
IM2_1									
IM2_2									
IM2_3									
IM2_4									
IM2_5									

Itarea	IM								
Items	1_8	1_9	1_10	2_1	2_2	2_3	2_4	2_5	2_6
O_Date	0.017	0.076	0.025	0.078	-0.011	0.070	0.043	-0.050	0.018
O_Day	-0.087	0.055	-0.033	0.216	0.079	0.128	0.059	0.025	-0.009
O_Month	0.001	0.041	0.031	0.184	0.116	0.086	0.068	0.071	0.024
O_Time	-0.007	-0.003	-0.023	0.091	-0.014	0.019	0.050	0.026	0.015
O_Year	0.019	0.013	0.053	0.152	0.089	0.061	0.042	0.045	0.044
C1	0.072	0.064	0.103	0.152	0.089	0.120	0.097	0.045	0.095
C2	0.056	0.076	0.081	0.044	0.039	0.026	0.043	0.092	0.074
C3	-0.007	0.039	0.043	0.050	0.113	0.021	0.064	0.012	0.060
C4	0.033	0.071	0.058	0.126	0.096	0.001	0.166	0.190	0.110
C_Month	0.090	0.134	0.097	-0.018	0.005	-0.043	0.053	0.023	0.084
IM1_1	-0.019	-0.074	-0.121	0.115	0.196	0.128	0.243	0.021	0.048
IM1_2	-0.112	-0.089	-0.164	0.080	0.258	0.132	0.116	0.071	0.025
IM1_3	-0.108	-0.118	-0.231	0.087	0.082	0.220	0.256	0.172	0.165
IM1_4	-0.094	-0.240	-0.189	0.082	0.046	0.094	0.221	0.142	0.156
IM1_5	-0.076	-0.115	0.037	-0.060	0.041	-0.035	-0.023	0.145	0.141
IM1_6	-0.046	-0.116	-0.113	0.081	0.038	-0.001	0.015	0.071	0.197
IM1_7	0.067	-0.014	0.060	0.015	-0.125	0.039	0.070	0.026	0.014
IM1_8		0.004	0.144	0.023	-0.047	-0.043	0.028	-0.039	0.056
IM1_9			0.286	-0.025	0.011	-0.093	-0.012	0.068	0.026
IM1_10				-0.021	0.004	-0.093	-0.138	0.031	0.014
IM2_1					0.278	0.213	0.135	0.085	0.008
IM2_2						0.184	0.077	0.165	0.032
IM2_3							0.107	0.007	0.035
IM2_4								0.082	0.045
IM2_5									0.131
IM2_6									

Itarea	IM								
nems	2_7	2_8	2_9	2_10	3_1	3_2	3_3	3_4	3_5
O_Date	0.099	0.051	0.092	0.075	0.033	0.001	0.052	0.008	0.076
O_Day	0.048	-0.057	0.104	0.017	0.153	0.055	0.072	0.059	0.086
O_Month	0.095	0.039	0.071	0.128	0.278	0.144	0.110	0.099	0.123
O_Time	0.071	0.049	-0.004	0.074	0.100	0.052	0.087	0.003	-0.007
O_Year	0.059	0.062	0.045	0.094	0.236	0.115	0.083	0.073	0.095
C1	0.059	0.113	0.101	0.094	0.236	0.192	0.083	0.073	0.095
C2	0.062	-0.061	0.031	0.075	0.080	0.085	0.052	0.031	0.025
C3	0.019	-0.019	0.107	0.037	0.053	0.157	0.074	0.081	0.066
C4	0.037	0.109	0.089	0.095	0.018	0.054	0.094	0.092	0.111
C_Month	0.112	0.057	0.067	0.086	0.075	0.009	0.007	0.077	0.040
IM1_1	0.043	0.007	-0.003	-0.038	0.163	0.097	0.237	0.183	0.036
IM1_2	0.025	-0.031	-0.002	-0.053	0.110	0.174	0.202	0.048	0.098
IM1_3	0.027	0.008	-0.142	-0.116	0.157	0.059	0.104	0.145	0.101
IM1_4	0.079	0.096	-0.080	-0.163	0.064	0.030	0.036	0.155	0.042
IM1_5	-0.045	0.026	-0.069	-0.004	-0.034	0.070	-0.065	0.079	0.027
IM1_6	0.035	0.074	-0.026	0.012	0.070	0.052	0.004	-0.028	-0.031
IM1_7	0.251	0.033	0.061	-0.024	0.032	-0.032	0.052	0.048	0.057
IM1_8	0.026	0.287	0.172	0.135	-0.016	0.010	0.061	0.037	0.015
IM1_9	-0.039	0.035	0.213	0.188	0.001	0.041	-0.044	0.029	0.056
IM1_10	0.067	0.153	0.220	0.374	0.030	0.001	-0.042	0.024	0.093
IM2_1	-0.039	0.061	-0.016	-0.089	0.263	0.025	0.331	0.116	0.121
IM2_2	-0.145	0.002	-0.012	-0.107	0.078	0.301	0.241	0.151	0.194
IM2_3	-0.166	-0.069	0.020	-0.097	0.003	0.055	0.268	0.147	0.129
IM2_4	-0.010	-0.026	-0.110	-0.152	0.113	0.147	0.153	0.279	0.118
IM2_5	0.060	0.010	-0.129	0.008	0.035	0.126	0.024	0.133	0.330
IM2_6	0.028	0.096	-0.002	0.000	0.064	-0.032	-0.004	0.068	0.042
IM2_7		0.044	0.038	0.003	0.069	0.024	0.013	0.080	0.106
IM2_8			0.188	0.184	0.023	0.033	0.039	0.056	0.033
IM2_9				0.246	0.035	0.008	-0.005	-0.100	-0.036
IM2_10					-0.003	-0.010	-0.056	0.011	0.048
IM3_1						0.117	0.103	0.053	0.017
IM3_2							0.121	0.016	0.154
IM3_3								0.130	0.121
IM3_4									0.174
IM3_5									
IM3_6									

	IM	IM	IM	IM	IM			
Items	3_6	3_7	3_8	3_9	3_10	DR1	DR2	DR3
O_Date	0.059	0.046	-0.006	0.113	0.054	0.093	-0.005	0.096
O_Day	0.065	0.100	-0.022	0.044	0.015	0.082	0.008	0.106
O_Month	0.047	0.069	0.065	0.086	0.099	0.118	0.092	0.072
O_Time	0.010	0.052	0.044	0.019	0.071	-0.011	0.060	0.058
O_Year	0.020	0.043	0.039	0.061	0.135	0.092	0.066	0.047
C1	0.020	0.098	0.147	0.179	0.135	0.092	0.127	0.102
C2	0.002	0.086	0.014	0.113	0.100	0.068	-0.005	0.034
C3	0.052	0.048	0.021	0.059	-0.013	0.124	0.102	0.067
C4	0.109	0.113	0.067	0.025	0.079	0.113	0.162	0.057
C_Month	0.071	0.102	0.026	0.097	0.060	0.066	0.071	0.028
IM1_1	-0.014	0.062	0.097	-0.021	-0.026	0.059	0.091	0.094
IM1_2	0.104	0.005	0.016	-0.086	-0.085	0.087	0.091	0.018
IM1_3	0.189	0.045	0.131	0.017	-0.170	0.095	0.055	-0.020
IM1_4	0.139	0.113	0.070	0.011	-0.070	0.088	0.081	0.110
IM1_5	0.025	0.013	0.143	-0.035	-0.022	-0.053	0.027	0.046
IM1_6	0.161	0.106	0.056	0.014	-0.044	0.049	0.100	0.080
IM1_7	0.091	0.198	0.047	0.051	0.074	0.045	0.018	0.041
IM1_8	0.010	-0.050	0.047	0.132	0.129	0.102	-0.010	-0.059
IM1_9	0.008	0.019	0.008	0.098	0.205	-0.032	0.052	-0.012
IM1_10	0.014	0.015	0.043	0.072	0.349	0.031	-0.018	-0.002
IM2_1	0.107	0.099	0.066	0.000	0.071	0.113	0.078	0.189
IM2_2	0.165	-0.020	-0.008	-0.004	-0.032	0.054	0.144	0.096
IM2_3	0.209	0.062	0.065	-0.018	-0.043	0.003	0.076	0.183
IM2_4	0.144	0.130	0.056	-0.008	-0.058	0.176	0.164	0.078
IM2_5	0.180	0.210	0.064	0.020	-0.018	0.086	-0.005	0.024
IM2_6	0.274	0.113	0.201	0.106	0.055	0.074	0.069	0.010
IM2_7	0.105	0.332	-0.002	0.154	0.106	0.058	-0.034	0.026
IM2_8	0.043	0.017	0.332	0.145	0.194	0.152	0.114	0.098
IM2_9	0.033	0.078	0.218	0.369	0.283	0.011	0.075	-0.001
IM2_10	-0.008	0.020	0.134	0.131	0.359	0.077	0.060	-0.020
IM3_1	0.041	0.032	0.024	0.033	0.053	0.359	0.074	0.066
IM3_2	0.145	-0.031	-0.012	0.020	-0.098	0.102	0.349	0.046
IM3_3	0.040	0.045	-0.030	-0.023	-0.048	-0.033	0.028	0.448
IM3_4	0.089	0.137	-0.005	-0.115	-0.047	0.110	0.023	0.070
IM3_5	0.157	0.078	-0.059	0.015	0.054	0.090	-0.033	0.076
IM3_6		0.131	0.021	0.040	-0.038	0.151	0.175	0.110
IM3_7			0.158	0.101	0.083	0.167	0.052	0.134
IM3_8				0.141	0.169	0.045	0.135	0.036
IM3_9					0.263	0.019	0.048	0.092
IM3_10						0.059	0.038	0.029

Items	DR4	DR5	DR6	DR7	DR8	DR9	DR10
O_Date	0.060	0.004	0.014	0.082	-0.001	0.061	0.058
O_Day	0.039	0.112	0.079	0.094	0.008	0.083	0.111
O_Month	0.082	0.076	0.033	0.043	0.034	0.036	0.113
O_Time	0.013	-0.026	-0.018	0.057	0.039	-0.012	0.021
O_Year	0.057	0.051	0.056	0.015	0.006	0.008	0.131
C1	0.115	0.107	0.056	0.066	0.107	0.109	0.078
C2	0.039	0.024	0.070	0.139	0.036	0.098	0.154
C3	0.138	0.067	-0.038	0.053	0.023	0.070	0.081
C4	0.132	0.144	0.057	0.116	0.038	0.031	0.094
C_Month	0.051	0.104	0.042	0.009	0.114	0.123	0.126
IM1_1	0.191	0.151	0.033	0.017	0.015	0.041	0.000
IM1_2	0.056	0.049	0.103	0.055	0.000	-0.018	-0.007
IM1_3	0.191	0.124	0.129	0.158	0.087	0.033	-0.064
IM1_4	0.097	0.103	0.163	0.099	0.082	-0.008	0.024
IM1_5	0.038	-0.017	0.077	0.012	0.192	-0.031	0.072
IM1_6	-0.001	0.007	0.154	0.131	0.073	0.050	0.056
IM1_7	0.013	0.061	0.092	0.200	0.042	-0.026	-0.007
IM1_8	-0.021	0.024	0.040	0.022	0.044	0.115	0.080
IM1_9	0.070	0.040	0.012	-0.043	0.051	0.100	0.016
IM1_10	0.042	0.110	0.042	-0.052	0.082	0.001	0.248
IM2_1	0.159	0.121	0.024	0.147	0.008	-0.077	-0.007
IM2_2	0.137	0.113	0.110	0.007	0.114	0.056	0.026
IM2_3	0.128	0.086	0.181	0.062	0.057	-0.003	-0.012
IM2_4	0.324	0.127	0.109	0.112	0.028	0.097	0.071
IM2_5	0.115	0.269	0.226	0.126	0.099	-0.032	-0.021
IM2_6	0.097	0.001	0.244	0.120	0.123	0.013	0.119
IM2_7	-0.038	0.103	-0.012	0.195	-0.013	-0.014	0.075
IM2_8	0.000	0.073	0.084	0.032	0.318	0.071	0.148
IM2_9	-0.024	0.034	0.037	0.059	0.154	0.279	0.139
IM2_10	0.022	0.100	-0.011	-0.039	0.105	0.097	0.278
IM3_1	0.027	-0.012	-0.042	0.135	0.038	0.042	0.053
IM3_2	-0.009	0.111	0.118	-0.013	0.045	0.037	0.010
IM3_3	0.129	0.075	0.010	0.032	-0.064	-0.029	0.027
IM3_4	0.544	0.145	0.025	0.103	0.067	-0.047	0.081
IM3_5	0.146	0.465	0.077	0.051	0.025	-0.064	0.097
IM3_6	0.111	0.105	0.582	0.108	0.115	0.028	0.015
IM3_7	0.130	0.072	0.142	0.456	0.160	0.107	0.137
IM3_8	0.069	0.036	0.093	0.120	0.480	0.038	0.072
IM3_9	-0.021	0.099	0.098	0.145	0.128	0.376	0.147
IM3_10	0.058	0.047	-0.038	0.014	0.155	0.178	0.519

Items	DR2	DR3	DR4	DR5	DR6	DR7	DR8	DR9	DR10
DR1	0.284	0.121	0.053	0.169	0.124	0.143	0.115	0.124	0.108
DR2		0.081	0.073	0.047	0.153	0.039	0.123	0.134	-0.017
DR3			0.161	0.091	0.148	0.046	0.032	0.000	0.057
DR4				0.119	0.069	0.083	0.063	-0.041	0.122
DR5					0.160	0.042	0.087	-0.014	0.037
DR6						0.144	0.106	0.004	-0.036
DR7							0.204	0.037	0.053
DR8								0.095	0.160
DR9									0.187
DR10									

Appendix G

SCAT3 Total Variance Explained

				Extra	ction Sums	of Squared	Rotation Sums of Squared Loadings		
		Initial Eigen	values		Loading	gs		Loading	gs I vi
Com.	Total	% Variance	Cumulative %	Total	% Variance	Cumulative %	Total	% Variance	Cumulative %
1	2.575	8.584	8.584	2.575	8.584	8.584	2.003	6.678	6.678
2	1.882	6.274	14.857	1.882	6.274	14.857	1.832	6.105	12.783
3	1.767	5.891	20.748	1.767	5.891	20.748	1.717	5.722	18.505
4	1.656	5.521	26.269	1.656	5.521	26.269	1.535	5.116	23.621
5	1.439	4.797	31.066	1.439	4.797	31.066	1.509	5.029	28.650
6	1.325	4.418	35.484	1.325	4.418	35.484	1.434	4.781	33.431
7	1.254	4.179	39.663	1.254	4.179	39.663	1.388	4.627	38.057
8	1.200	3.999	43.662	1.200	3.999	43.662	1.247	4.157	42.215
9	1.179	3.929	47.591	1.179	3.929	47.591	1.235	4.116	46.330
10	1.113	3.710	51.301	1.113	3.710	51.301	1.232	4.107	50.437
11	1.080	3.599	54.899	1.080	3.599	54.899	1.222	4.073	54.510
12	1.054	3.513	58.412	1.054	3.513	58.412	1.115	3.717	58.227
13	1.007	3.356	61.768	1.007	3.356	61.768	1.062	3.541	61.768
14	0.976	3.253	65.021						
15	0.930	3.100	68.120						
16	0.880	2.932	71.052						
17	0.857	2.858	73.910						
18	0.828	2.761	76.671						
19	0.778	2.592	79.263						
20	0.734	2.446	81.708						
21	0.726	2.420	84.128						
22	0.671	2.237	86.365						
23	0.645	2.149	88.515						
24	0.596	1.986	90.501						

			Extraction Sums of Squared			Rotation Sums of Squared			
	Initial Eigenvalues			Loadings			Loadings		
		%	Cumulative		%	Cumulative		%	Cumulative
Com.	Total	Variance	%	Total	Variance	%	Total	Variance	%
25	0.565	1.884	92.385						
26	0.537	1.791	94.176						
27	0.515	1.716	95.892						
28	0.463	1.544	97.436						
29	0.392	1.306	98.742						
30	0.377	1.258	100.000						

Appendix H

SCAT5 Total Variance Explained

	Initial Eigenvalues		Extraction Sums of Squared Loadings			Rotation Sums of Squared Loadings			
Com.	Total	% Variance	Cumulative %	Total	% Variance	Cumulative %	Total	% Variance	Cumulative %
1	4.895	9.789	9.789	4.895	9.789	9.789	3.306	6.611	6.611
2	3.327	6.654	16.444	3.327	6.654	16.444	2.189	4.377	10.988
3	2.716	5.431	21.875	2.716	5.431	21.875	2.041	4.081	15.07
4	2.055	4.11	25.985	2.055	4.11	25.985	2.038	4.076	19.146
5	1.898	3.796	29.782	1.898	3.796	29.782	2.002	4.004	23.15
6	1.736	3.471	33.253	1.736	3.471	33.253	1.928	3.856	27.006
7	1.656	3.311	36.564	1.656	3.311	36.564	1.879	3.757	30.763
8	1.63	3.261	39.825	1.63	3.261	39.825	1.735	3.47	34.233
9	1.421	2.841	42.666	1.421	2.841	42.666	1.717	3.434	37.668
10	1.357	2.714	45.38	1.357	2.714	45.38	1.689	3.378	41.046
11	1.308	2.616	47.996	1.308	2.616	47.996	1.617	3.234	44.28
12	1.256	2.512	50.508	1.256	2.512	50.508	1.616	3.232	47.512
13	1.229	2.458	52.966	1.229	2.458	52.966	1.564	3.128	50.64
14	1.223	2.447	55.413	1.223	2.447	55.413	1.542	3.084	53.724
15	1.117	2.233	57.646	1.117	2.233	57.646	1.398	2.797	56.521
16	1.078	2.156	59.802	1.078	2.156	59.802	1.346	2.692	59.213
17	1.01	2.019	61.822	1.01	2.019	61.822	1.305	2.609	61.822
18	0.98	1.96	63.781						
19	0.968	1.936	65.717						
20	0.939	1.879	67.596						
21	0.906	1.812	69.408						
22	0.886	1.772	71.18						
23	0.862	1.725	72.904						
24	0.793	1.586	74.49						
25	0.783	1.565	76.055						
26	0.72	1.44	77.495						
27	0.705	1.411	78.906						
28	0.682	1.364	80.27						
29	0.665	1.33	81.6						
30	0.656	1.312	82.912						
31	0.633	1.266	84.178						
32	0.611	1.222	85.4						
33	0.587	1.174	86.574						

	Initial Eigenvalues			Extraction Sums of Squared Loadings			Rotation Sums of Squared Loadings		
Com.	Total	% Variance	Cumulative %	Total	% Variance	Cumulative %	Total	% Variance	Cumulative %
34	0.559	1.119	87.693						
35	0.531	1.062	88.755						
36	0.52	1.039	89.794						
37	0.5	1	90.795						
38	0.477	0.954	91.749						
39	0.472	0.944	92.692						
40	0.445	0.889	93.582						
41	0.413	0.826	94.408						
42	0.399	0.797	95.205						
43	0.377	0.753	95.958						
44	0.354	0.707	96.666						
45	0.34	0.679	97.345						
46	0.324	0.648	97.993						
47	0.323	0.646	98.639						
48	0.305	0.61	99.249						
49	0.272	0.544	99.794						
50	0.103	0.206	100						

Appendix I

Item		Fac	etors	
(N = 30)	F1	F2	F3	F4
C1	0.688			
IM3_5	0.643			
IM3_2	0.603			
IM3_4	0.572			
IM2_2	0.495			
IM2_5	0.461			
IM2_4	0.402			
O_Date				
IM3_3				
DR1		0.628		
DR5		0.611		
DR2		0.594		
DR4		0.575		
DR3		0.429		
O_Month		0.259		
O_Time		0.208		
O_Year				
IM2_3			0.637	
IM1_2			0.628	
IM1_3			0.595	
IM1_4			0.561	
O_Day			0.237	
IM1_5			0.212	
C3				0.785
C4				0.643
C2				0.585
C_Months				0.353
IM3_1				
IM1_1				
IM2_1				
Extraction Method: I	Principal Component A	nalysis.		

Pattern Matrix for SCAT3

Appendix J

Item	Factors							
(N = 30)	F1	F2	F3	F4				
IM2_2	0.552							
DR4	0.513							
IM3_4	0.481							
IM3_3	0.474							
IM2_4	0.467							
IM3_5	0.464							
DR5	0.432							
IM1_2	0.432							
IM1_1	0.429							
IM2_1	0.414							
IM1_3	0.401							
IM2_3	0.394							
C4	0.380							
IM3_2	0.375							
IM2_5	0.352							
DR3	0.323							
C3	0.296							
DR2	0.286							
IM3_10		0.679						
IM1_10		0.610						
IM2_10		0.607						
IM2_9		0.566						
DR10		0.514						
IM3_9		0.451						
IM2_8		0.427						
IM1_9		0.425						
IM1_8		0.353						
DR9		0.349						
C_Months		0.251						
Extraction Method: Principal	l Component Analysis.							

Pattern Matrix for SCAT5

Rotation Method: Oblimin with Kaiser Normalization.

Factor loadings < 0.2 were suppressed.

Item		Fac	ctors	
(N = 30)	F1	F2	F3	F4
O_Month			-0.902	
O_Year			-0.849	
O_Day			-0.714	
C1			-0.641	
O_Time			-0.593	
O_Date			-0.376	
IM3_1			-0.373	
C2			-0.262	
DR7				-0.564
IM3_7				-0.562
DR6				-0.499
IM3_6				-0.478
IM2_6				-0.460
DR8				-0.438
IM1_4				-0.427
IM3_8				-0.413
IM1_6				-0.383
IM2_7				-0.336
IM1_7				-0.311
DR1				-0.294
IM1_5				
Extraction Method: P	rincipal Component Analys	sis.		

Rotation Method: Oblimin with Kaiser Normalization. Factor loadings < 0.2 were suppressed.